State of Rhode Island

Department of Human Services

Center for Child and Family Health

Revision of

Certification Standards

Providers of Home Based Therapeutic Services

June 2006

TABLE OF CONTENTS

Secti	on				Page
1.0	SER	VICE 1	NFORMATIO	N AND BACKGROUND	9
	1.1	Intro	duction		9
		1.1.1	Home-Based	Therapeutic Services	10
			1.1.1.1	Clinical Supervision	
			1.1.1.2	Home-Based Therapy Direct Services	
			1.2.1.		
			1.2.1.		
			1.2.1.		
			1.1.1.3	Child Specific Orientation for Newly Assigned HBTS Worker	13
			1.1.1.4	Group Intervention	
			1.1.1.5	HBTS Treatment Consultation	
			1.1.1.5		
			1.1.1.5		
			1.1.1.5		
			1.1.1.6	Treatment Coordination	
	1.2	Inten	ded Outcomes o	of Certification Standards and Services	14
	1.3	HBT	S as a CEDARR	R Direct Service	15
		1.3.1	Prior Approva	al Process, Period of Authorization, and	
			PA and Non-l	PA Services	16
			1.3.1.1	Period of Authorized Services	16
			1.3.1.2	Prior Authorization – PA Services	
			1.3.1.3	Non – PA Services	
		.3.2		of Care and HBTS	
		.3.3		rization Process (CEDARR and DHS)	
		.3.4		with CEDARR Family Center for PA of HBTS	
		.3.5		with DHS for Prior Authorization of HBTS	
		.3.6		to Family Centered Care	
	1.4	CED	ARR Family Ce	enters	20
2.0	CER	TIFIC	ATION PROCI	ESS	21
	2.1	Subn	nission of Certif	ication Application Required	21
	2.2			fication to Applicants	
	2.3			ested Parties	
	2.4		fication		
		2.4.1		come of Certification Review Process	
			2.4.1.1	Certification Offer	
			2.4.1.2	Period of Certification	
			2.4.1.3	Recertification	24

		2.4.2 Certification Status and Reimbursement Schedules	24
	2.5	Continued Compliance with Certification Standards	25
		2.5.1 Provisional Certification	
	2.6	Licensure Requirements for Certified HBTS Provider-Agencies	
		2.6.1 Variance from Professional Licensure	
		2.6.1.1 Conditions for Variance	
		2.6.1.1.1 Holder of Variance	
		2.6.1.1.2 Duration of Variance	
		2.6.1.1.3 Changes in Status	
		2.6.1.1.4 Portability	
		2.6.1.2 Exception from Professional Licensure	
		2.6.1.2.1 Temporary Exception Status License Eligible He	
		Care Professionals	
		2.6.1.2.2 Temporary Exception for Other Credentialed	20
		Professionals	29
		2.6.1.3 Exemption from Professional Licensure	
	2.7	Clinical Oversight and Monitoring.	
	2.7	2.7.1 Licensed clinician's Professional Oversight Responsibility	
		2.7.1.1 Expertise and Competence	
	2.8	Home-Based Workers	
	2.0	2.8.1 February 2003 Certification Standards	
		2.8.2 New Accommodations	
	2.9	DHS Responsibilities.	
	2.7	2.9.1 Oversight and Authorization.	
		2.7.1 Oversight and rathorization	52
3.0	BAC	CKGROUND DEVELOPMENT OF HOME BASED THERAPEUTIC	
•••		VICES	33
	SER		
	3.1	History	33
	3.2	Additional Supports for Children with Special Health Care Needs	
	3.2	reductional Supports for Children with Special Fleaten Care recess	55
4.0	TAR	RGET POPULATION AND LOCATION OF SERVICE WITHIN A	
•••		NTINUUM OF CARE	35
	001		
	4.1	Eligibility	35
	4.2	Home-Based Therapeutic Services within the Continuum of Care;	
		Appropriateness of this Level of Care	35
		4.2.1 Clinical Appropriateness Criteria for Initiation of Services	
		4.2.2 Clinical Appropriateness Criteria for Continuing Care	
		4.2.3 Discontinuing Services	
		4.2.4 Discharge Criteria	
	4.3	Potential of Service and Limitations of Service	
	т.Э	4.3.1 Potential of Service	
		4.3.2 Limitations of Service	
		1.5.2 Difficultions of Get vice	,

5.0	SERV	VICE DI	ESCRIPT	ION – REQUIRED SCOPE OF SERVICES	40
	5.1	Service	e Name an	nd Definition	40
	5.2	Service	e Compone	ents	40
	5.3	Units a	and Rate of	f HBTS Services	40
		5.3.1	Home-Bas	sed Therapeutic Services Units	41
	5.4	Descri	ption of Se	ervice Components	42
			Table 4:	HBTS Reimbursable Services	42
			Table 5:	Pre-Treatment Consultation – Reimbursable Services	47
			Table 6:	Post-Treatment Consultation - Reimbursable Services	48
		5.4.1	Prior App	proval for HBTS	50
			5.4.1.1	Prior Approval and Authorization Process	
			5.4.1.2	Modifications to Prior Authorization Requests	
			5.4.1.3	Alignment of CEDARR and HBTS Treatment Plans	51
		5.4.2	Clinical S	Supervision	52
			5.4.2.1	Clinical Supervision of HBTS Worker	52
			5.4.2.2	Clinical Supervision of Treatment Support Worker	53
		5.4.3	Home-Ba	ased Therapy Direct Services	53
			5.4.3.1	HBTS – Specialized Treatment	54
				5.4.3.1.1 Specialty Consultations	54
				5.4.3.1.1.1 Occupational Therapy	54
				5.4.3.1.1.2 Physical Therapy	
				5.4.3.1.1.3 Speech and Language Therapy.	55
			5.4.3.2	HBTS – Treatment Support	55
		5.4.4	Group Int	terventions	56
			5.4.4.1	Group Intervention Service Supports	56
		5.4.5	Child Spe	ecific Orientation	57
		5.4.6	Treatmen	t Consultation	58
			5.4.6.1	Pre-Treatment Consultation	58
			5.4.6.2	Treatment Consultation	58
			5.4.6.3	Specialty Consultation	59
			5.4.6.4	Post-Treatment Consultation	59
		5.4.7	Treatmen	t Coordination	59
	5.5			sity of HBTS and Therapeutic Approaches	
	5.6	Duration		ntinuation of Service	
		5.6.1	Categorie	es of Treatment Requests	61
			5.6.1.1	New HBTS Treatment Plans	
			5.6.1.2	Reauthorization of Treatment Plan (Renewals)	
	5.7	Family	Involvem	nent and Responsibility	62
	5.8	Transp		nd Travel Time Reimbursement	
		5.8.1	-	tation and 2:1 Coverage	
		5.8.2		me Reimbursement	
	5.9			Current HBTS Referral Lists	
	5.10			Other DHS Procedural Directives for HBTS	
	5.11	HBTS	Performar	nce Standards	67

		5.11.1	Timelines	ss of Service Provision	
			Parent Sat		
		5.11.3	Provision	of Authorized Services	67
5.0	CER	TIFICA	ATION STA	ANDARDS	69
	6.1	Requi	rements for	Organization of Delivery of Service	69
	6.2	Agree		cept Appropriate Referrals	
		6.2.1		of Authorized Services	
	6.3			ness, Client Rights, and Ethical Standards Practice	
		6.3.1		enteredness	
		6.3.2		ghts and Family Service	
			6.3.2.1	J	
			6.3.2.2	3	
			6.3.2.3		
		6.3.3	Ethical St		
				onflicts of Interest	
		6.3.4		Reporting of Child Abuse and Neglect	
	<i>c</i> 1	6.3.5		sysical Restraint Interventions	
	6.4			Communication with CEDARR Family Center	
		6.4.1		ferral to a CEDARR Family Center	/ /
		6.4.2		R Family Center Initial Family Assessment (IFA)	77
		(12	and Basic		
		6.4.3		Family Center Care Plan	
		6.4.4 6.4.5		R Family Center Certified HBTS Treatment Plan	
		0.4.3	6.4.5.1	R – HBTS Provider-Agency Dispute Resolution Process HBTS Provider-Agency and CEDARR Family Center	19
			0.4.3.1	Disagreement Process	70
			6.4.5.2	DHS Fair Hearing Process	
	6.5	Strene		ram Approach: Process of Care	
	0.5			of Clinical Services	80
			Process of		
		0.5.1	6.5.1.1	Treatment Approach and Clinical Guidelines	80
			6.5.1.2	Screening and Intake for HBTS	
			6.5.1.3	Assessment and Treatment Planning	
			0.0.11.0	6.5.1.3.1 Diagnosis and Treatment History	
				6.5.1.3.2 Treatment Plan Development	
			6.5.1.4	Treatment Plan Implementation	
			6.5.1.5	Treatment Plan Modification and Renewals	
		6.5.2		nent of Clinical Services	
			6.5.2.1	Clinical Roles and Scope of Practice	
				6.5.2.1.1 Agency Orientation and Training	
				6.5.2.1.2 Preparation of Home-Based Staff	
			6.5.2.2	Clinical Supervision	
			6.5.2.3	Applied Behavior Analytic Services	

		6.5.2.3.1 Lead Therapy	
		6.5.2.4 Staffing and Staff Qualifications	88
		6.5.2.4.1 Background Screening for New Staff	89
		6.5.2.4.1.1 Home-Based Specialized	
		Treatment Worker	89
		6.5.2.4.1.2 Home-Based Treatment	
		Support Worker	90
		6.5.2.4.1.3 Clinical Supervisor	90
		6.5.2.4.1.4 Treatment Consultant	91
		6.5.2.4.1.5 Treatment Coordinator	
		6.5.2.3.1.6 Lead Therapist	
	6.6	Timeliness of Service, Other Access Standards	92
		6.6.1 Timeliness Standards for NEW Referrals	
		6.6.1.1 Intake Appointment	
		6.6.1.2 Treatment Plan Submissions	
		6.6.13 Initiation of Direct Services	
		6.6.2 Timeliness for RENEWING Cases	
		6.6.3 Timeliness Standards for Treatment Plan Clinical Review Process	
		6.6.4 Hours of Service	
		6.6.4.1 Continuity of Care	
		6.6.5 Measures of Parent Satisfaction	
	6.7	Service Monitoring and Reporting	
		6.7.1 Quarterly Reports	
		6.7.2 Annual Reports	
		6.7.3 Additional Service Monitoring and Reporting	
	6.8	Record Keeping Requirements	
	6.9	Emergency Coverage	95
7.0	QUA	LIFIED ENTITY	96
	- 1		0.4
	7.1	Incorporation and Accountable Entity	
	7.0	7.1.1 Partnership or Collaboration	
	7.2	Governance and Mission	97
	7.3	Well Integrated and Organized Management and Operating Structure	
	7.4	7.3.2 Financial Systems Human Resources, Staffing	
	7.4	Quality Assurance/Performance Improvement	
	7.6	Information Management, Record Keeping	
	7.7	Health and Safety, Risk Management	
	7.7	Transportation	
APPI	ENDIC	ES	
Appe	ndix 1:	Definition of Medical Necessity	105

Appendix 2:	CEDARR	Authorization Process for HBTS	106
Appendix 3:	DHS Autl	norization Process for HBTS	108
Appendix 4:	Licensure	and Practice Standard	110
	1.0	Core Requirements for HBTS Certification	110
	1.1	Licensure	110
	1.2	Competency for Licensed Health Care Professionals	110
	1.2.1	Training	111
	1.2.2	Education	
	1.2.3	Continuing Education	111
	2.0	Special Provisions – Exception from Professional Licensure	
	2.0.1	Transition Clinician Group	
	2.0.1.1	Demonstrated Competency Required for Exception Status	
	2.0.2	Temporary Exception Status from Professional Licensure	
	2.0.2.1	License Eligible Healthcare Professionals	
	2.0.2.1.1	Licensed Clinical Social Worker	
	2.0.2.2	Other Credentialed Professionals	114
	2.0.2.2.1	Professional Teacher Certification – School Social Worker.	
	2.0.2.2.2	School Psychologist	
	2.0.2.2.3	Teacher with Masters Degree in Special Education	
	2.0.2.2.4	Rhode Island Department of MHRH	
	2.1	Reciprocity and Temporary Exception Status	
	2.1.1.	Individuals with Professional Licenses from Other States	
	2.1.2	Certification from National Association of Social Workers	
	2.1.3	Continuing Education Requirement	
	2.1.4	Clinical Oversight of Non-Licensed Individuals	
	2.1.5	Exemption from Professional Licensure	
	2.1.5.1	Board Certified Behavior Analyst or Associate Certified	
	2.1.3.1	Behavior Analyst	117
		Table 9: Summary of Licensure and Practice Standards	
		- 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1	
Appendix 5:	Description	on of Conditions Associated with Target Populations	118
Appendix 6:	Provider-A	Agency Responsibility for Monitoring Medicaid Eligibility	120
Appendix 7:	Treatment	t Support Domains	121
Appendix 8:	Treatment	t Intensity for HBTS Specialized Treatment	123
Appendix 9:	Parent Par	rticipation in HBTS	125
Appendix 10:	Provider-A	Agency Responsibilities for Discontinuation of HBTS	126
Appendix 11:	Appeal Ri	ights-Rhode Island Department of Human Services	128
Appendix 12:		of Global Assessment Scales	
PPendin 12.	2/14/11/2100	or orona rabbooman boured minimum.	101

	Global Assessment of Functioning Scale	
Appendix 13:	Descriptions of Accreditation Abbreviations	133
Appendix 14:	Guidelines for Consumers of Applied Behavior Analysis Services to Individuals with Autism	134
Appendix 15:	Documentation Guidelines for HBTS	137
Appendix 16:	CEDARR Family Centers	141
Appendix 17:	Service Monitoring and Reporting Requirements	142
Appendix 18:	Recertification Requirements and Process	144
Appendix 19:	Documentation Guidelines and Requirements for Specialty Consultations – OT, PT and SLP	
Appendix 20:	List of HBTS Provider – Agencies and Programs	147
Appendix 21:	Prior Approval and Prior Authorization	148
Appendix 22:	Travel Reimbursement	149
Appendix 23:	Frequently Asked Questions Licensure Related Questions Billing and Reimbursement Questions Treatment Plan Questions Training Questions Prior Approval Questions Travel Reimbursement Questions	152 153 153 154
Attachments:		
Attachment A:	Application Guide	157

State of Rhode Island

Department of Human Services

Center for Child and Family Health

Certification Standards

Providers of Home Based Therapeutic Services

June 2006

1.0 SERVICE INFORMATION AND BACKGROUND

1.1 Introduction

The Rhode Island Department of Human Services (DHS) issued Certification Standards for Home Based Therapeutic Services (HBTS) in February of 2003. HBTS is for children who are Medicaid eligible with significant impairments in functioning. Children with special health care needs are those who have or are at risk for chronic physical, developmental, behavioral or emotional conditions and who also require health and related services of a type or amount beyond that required by children generally.

Since HBTS Certification Standards were issued, DHS has continued to elicit comments from provider-agencies, parents, and advocacy groups regarding the administration and delivery of HBTS. This process allowed DHS to modify certain aspects of the February 2003 HBTS Certification Standards. DHS informed interested parties of these changes, enhancements or further clarifications through meetings and written correspondence. This edition of HBTS Certification Standards supercedes all previous guidelines, verbal and written, issued by DHS.

Home Based Therapeutic Services require prior authorization. HBTS Certification Standards provide the basis for determination of provider-agencies eligible to offer and receive payment for provision of HBTS. These standards describe the basis and mechanisms of prior authorization, as well for payment for services. Parties interested in becoming a provider-agency for HBTS may apply for certification at any time.

These Certification Standards serve to provide families, potential applicants, service providers and other interested parties with a full description of Home Based Therapeutic Services, including guidance as to certification requirements and methods for application. Sections 1 through 5 contain service description and background as follows:

Section 1: Service Information and Background

Section 2: Certification Process

Section 3: Home Based Therapeutic Services

Section 4: Target Population and Location of Service Within the Continuum of Care

Section 5: Service Description – Required Scope of Services

Section 1 provides an introduction to the service. Section 2 describes the process for certification. Section 3 contains a statement of the need for the service and the processes leading to development of these standards. Section 4 identifies the group of children that this service is expected to benefit and delineates how this service relates to the overall continuum of care. Section 5, Service Description, contains a detailed description of the service and identifies core requirements for the service.

The Certification Standards include two additional sections as follows:

Section 6: Requirements for Organization of Service Delivery – Performance

Standards

Section 7: Qualified Entity Requirements

Sections 6 and Section 7 specifically describe the requirements for certification. Satisfactory compliance with these requirements must be demonstrated for certification; continuing compliance is required in order to maintain full certification status.

Certification applications will be primarily focused on Section 6. Although certified entities must comply with the requirements set forth in Section 7, the requirement to demonstrate such compliance in the application itself is more limited.

1.1.1 Home Based Therapeutic Services

HBTS are provided for children living at home (may also include children living with a foster family) who have been diagnosed with moderate to severe physical, developmental, behavioral or emotional conditions. These children have chronic health care needs that require health and related services beyond those required by children generally. Section 4.0 provides greater discussion about target populations and clinical appropriateness criteria.

HBTS can only be provided when there is medical necessity, as documented by a physician's prescription (See - Appendix 1: Definition of Medical Necessity), and documented evidence that HBTS can meet the needs of the child with special health care needs. These services require authorization from a CEDARR Family Center (CFC) or DHS. HBTS are more intensive than outpatient treatment but less restrictive than inpatient hospitalization or residential care.

Home Based Therapeutic Services are specialized health services delivered in a child's home and in community settings. They represent an integrated set of service components involving the provision of specialized treatment provided in accordance with an approved individualized Treatment Plan with measurable goals and objectives. The Treatment Plan must be developed in

consultation with a licensed health care professional¹ with expertise to oversee the entire plan of care. This requires that the provider-agency have a licensed clinician identified on the Treatment Plan. This requirement also applies to situations whereby clinical staff has met variance from professional licensure.

Once a Treatment Plan is approved by a CEDARR Family Center or DHS, the Treatment Plan is implemented by the HBTS provider-agency. It is the responsibility of the provider-agency to show that it is working in collaboration and communication with all relevant parties (e.g., CEDARR Family Centers, parents, school personnel, medical home, and mental health providers) in the development and implementation of services. Collaboration is an ongoing activity. Evidence of this collaboration must be referenced in the Treatment Plan.

An integrated HBTS Treatment Plan can include, in specified amounts as approved, the following reimbursable service components:

1.1.1.1 Clinical Supervision

Clinical Supervision of the home-based worker is provided by a competent and licensed health care professional², unless otherwise approved by DHS, to ensure effective implementation and oversight of the Treatment Plan. Overall, Clinical Supervision includes, but is not limited to, review and consultation on therapeutic methods used by the home-based worker, analysis of the child's response and progress, and adjustments to the therapeutic regimen as appropriate. Unless otherwise indicated by the provider-agency to DHS, the Clinical Supervisor has complete responsibility for the content and management of a child's HBTS Treatment Plan.

1.1.1.2 Home-Based Therapy Direct Services

There are two components of HBTS direct services, namely, Specialized Treatment and Treatment Support. Specialized Treatment and Treatment Support are provided to a child by a home-based worker in accordance with the approved Treatment Plan, and under the supervision of the licensed Clinical Supervisor.

1.1.1.2.1 HBTS Specialized Treatment: Represents one-on-one therapeutic services given to a child by a home-based worker (paraprofessional) in accordance with the approved Treatment Plan under the supervision of the licensed Clinical Supervisor^{1,2}. Specialized Treatment may address the development of behavioral, communication, social, and functional skills, as well as reinforce skills included in a child's Individual

Department of Health licensed health care professionals eligible to serve in this capacity for HBTS include the following categories: licensed social worker, licensed independent clinical social worker, marriage and family therapist, mental health counselor, psychologist, and/or registered nurse with a Masters degree.

² Licensed health care professionals eligible to provide Clinical Supervision include the following categories: licensed clinical social worker, licensed independent clinical social worker, marriage and family therapist, mental health counselor, psychologist, registered nurse with a Masters degree and/or psychiatrist. Individual clinicians with Exemption or Exception status may also provide Clinical Supervision.

Educational Plan (IEP) or Individualized Family Service Plan (IFSP). This is accomplished by incorporating HBTS Specialty Consultations.

- 1.1.1.2.2 HBTS Specialty Consultations: For some children, it may be appropriate to integrate the recommendations of licensed Occupational Therapists (OT), Physical Therapists (PT), and Speech and Language Pathologists (SLP) into an HBTS Treatment Plan to provide additional practice on targeted skills. Specialty Consultations (OT, PT, or SLP) can only occur for children receiving these therapies as part of an IEP or IFSP. Specialty Consultations are intended to support a child's Treatment Plan by providing specific direction to the HBTS Treatment Worker to facilitate the child's mastery of treatment objectives. HBTS Specialty Consultations and related goals are not a substitution for actual therapy provided by an OT, PT or SLP therapist(s). There is no limit to the number or type Specialty Consultants permitted on a Treatment Plan, as long as services are medically necessary and consistent with prerequisite criteria.
- **1.1.1.2.3 HBTS Treatment Support:** DHS recognizes that for some children and adolescents with moderate to severe mental retardation, other developmental disorders, or neuromedical conditions the ability to fully participate in goal-directed HBTS is severely limited by their level of functioning and coexisting condition(s). The frequency and intensity of engaging in sustained Specialized Treatment can become too taxing and may be of limited effectiveness. The latter often involves intensive periods of practice or application of reinforcement procedures in efforts to improve skill levels and/or changing inappropriate behaviors.

The intent of Treatment Support is to facilitate the child's ability to remain at home, maintain activities of daily living, and transition to adulthood. Therefore, the inclusion of Treatment Support in a child's HBTS plan is to augment care based on the needs of a child. It does not represent a minimization of therapeutic effort and is not equivalent to respite care.

Treatment Support allows for a portion of approved HBTS hours to be used for the purposes of providing structure, supervision, guidance, and redirection when the child is not directly engaged in active goal-directed treatment. Treatment Support can be provided by the home-based worker engaged in offering Specialized Treatment or by another individual (Treatment Support Worker). The Clinical Supervisor must direct the activities of the home-based worker providing Treatment Support unless otherwise approved by DHS allowing for the use of non-licensed clinicians.

Any new provider-agency must demonstrate the ability to include Treatment Support as a component of care.

1.1.1.3 Child Specific Orientation for Newly Assigned Home-Based Worker

Child specific orientation provides the newly assigned home-based worker with detailed knowledge about a child's condition, treatment goals and objectives, methods of intervention, and other related aspects of care such as observing the child and/or other staff working with the child. It is provided by the Treatment Consultant or Clinical Supervisor and with an experienced home-based worker, when appropriate, to prepare new staff to work with a child and family already receiving HBTS.

Initially, DHS approved up to 4 units (2 hours) occurring twice (maximum) per treatment plan for Child Specific Orientation. DHS has now increased Child Specific Orientation to be a maximum of 20 units (10 hours) for Child Specific Orientation per 6-month period of HBTS intervention.

1.1.1.4 Group Intervention

Group Intervention is intended to address the development of social skills as well as communication, anger management, and problem solving capabilities in children or adolescents. It can be provided prior to the establishment of intensive home-based services as well as during a course of HBTS care or following the completion of HBTS interventions. Group interventions can be included and reimbursed as an additional component of treatment when provided by a licensed health care professional unless otherwise approved by DHS.³

Group intervention is of maximum benefit when children are able to carryover skills into naturally occurring community settings. It is critical that children involved in social skill groups, for example, also receive coaching and the support of HBTS staff to use and practice new skills. Learning is enhanced when children learn social skills in real life settings from and with typically developing age peers.

Any new provider-agency must demonstrate the ability to include Group Intervention as a component of care. For existing certified provider-agencies that have not developed Group Intervention as part of their set of services, DHS must be informed of a plan to include Group Intervention at the time of a provider-agency's recertification.

1.1.1.5 HBTS Treatment Consultation

HBTS Treatment Consultation shall be provided to the treatment team (Clinical Supervisor and home-based worker) by a licensed health care professional unless otherwise approved by DHS⁴

³ Licensed health care professionals eligible to provide Group Intervention include: licensed clinical social worker, licensed independent clinical social worker, marriage and family therapist, mental health counselor, and psychologist. Individual clinicians with Exemption or Exception status may also provide group intervention.

⁴ Licensed health care professionals eligible to provide Treatment Consultation include: psychiatrist, psychologist, licensed independent clinical social worker, marriage and family therapist, mental health counselor, occupational

with recognized expertise and competence in the specific area of the child's needs. It is intended to bring specific expertise and direction to the therapeutic regimen employed in the Treatment Plan. There are three categories of HBTS Treatment Consultation:

1.1.1.5.1 HBTS Treatment Consultant

The Treatment Consultant addresses a child's developmental, emotional, and behavioral needs by providing collaboration to the Clinical Supervisor and family. It is provided to the child's parents or guardian. Treatment Consultation takes place during the course of intensive HBTS Specialized Treatment. There are three types of HBTS Treatment Consultation, namely:

1.1.1.5.2 HBTS Pre and Post Treatment Consultation

Pre-Treatment Consultation involves consultation to the family regarding interventions to assist them in managing their child's behaviors.

Post-Treatment Consultation provides short-term support to the family upon the conclusion of a period of HBTS Specialized Treatment.

1.1.1.5.3 HBTS Specialty Treatment Consultation

An OT, PT, or SLP therapist provides this service. The Specialty Consultant addresses specific HBTS goals and objectives within one's professional discipline. It can also take place as part of Pre or Post Treatment.

1.1.1.6 Treatment Coordination

Treatment Coordination represents activities by a team member on behalf of a specific child receiving HBTS services to ensure coordination and collaboration with parents, providers, the medical home, and other agencies (e.g., school, Early Intervention, DCYF or CASSP) including the CEDARR Family Center. Collaboration and communication is ongoing throughout a child's course of HBTS.

1.2 Intended Outcomes of Certification Standards and Services

The development of Certification Standards and the provision of certified Home Based Therapeutic Services are intended to:

1. Improve the functioning of children with special health care needs as set forth in approved HBTS Treatment Plans. This includes maximizing their ability to live at home and actively participate as valued members of their families and communities to the best of their abilities, and to support the transition to adulthood

therapist, physical therapist, speck and language pathologist, and registered nurse with a Masters degree. Individual clinicians with Exception status may also provide Treatment Consultation.

- 2. Improve the access to provision of services as authorized by DHS or CEDARR Family Centers. Certified HBTS direct service providers are expected to render efficient, cost-effective treatment aimed at addressing clearly defined treatment goals and objectives
- 3. Improve monitoring, oversight and quality assurance of HBTS provider-agencies by DHS. Regular performance reports will be provided to DHS
- 4. Improve program reliability, consistency, and quality of service across provider-agencies by adhering to policies and DHS requirements governing the operation of HBTS. Existing HBTS provider-agencies must fully meet these Certification Standards. All new provider-agencies are to fully meet Certification Standards in order to provide HBTS
- 5. Increase provider capacity by adding new provider-agencies
- 6. Assure the adherence to family centered principles in the service relationship between provider-agency and family

1.3 HBTS as a CEDARR Direct Service

Certification of provider-agencies for the provision of Home Based Therapeutic Services is intended to further the "Statewide Vision for Children and Families with Special Health Care Needs." The Director of DHS convened a Leadership Roundtable on Children with Special Health Care Needs with a representative group of family members, providers, public and private administrators, and advocates. Their effort produced the following objective:

Statewide Vision

"All Rhode Island children and their families have an evolving, family centered, strength based system of care, dedicated to excellence, so they can reach their full potential and thrive in their own communities."

Leadership Roundtable on Children and Their Families with Special Health Care Needs, April 15, 1999

CEDARR stands for <u>C</u>omprehensive <u>E</u>valuation, <u>D</u>iagnosis, <u>A</u>ssessment, <u>R</u>eferral and <u>R</u>eevaluation services and supports. The CEDARR Program Initiative includes two broad delivery system components:

- CEDARR Family Centers, and
- CEDARR Certified Direct Services

The CEDARR Family Center Certification Standards more fully describe the role of the CEDARR Family Centers and the related Certification Standards. They are available on line on the DHS website reached at www.dhs.state.ri.us

CEDARR Direct Services are specific services developed pursuant to the CEDARR Initiative and available to Medicaid beneficiaries when included as part of an approved CEDARR Family Center Family Care Plan. Development of CEDARR Direct Services is based on two principles:

- 1. Identify current service needs and gaps in health care services for children and families with special health care needs; and
- 2. Establish and operate an accountable system for the purchase of appropriate, high quality services to meet those needs

These CEDARR Direct Service Certification Standards for HBTS specify the requirements that must be met to be a certified provider and provide guidance to interested parties who may choose to apply for certification. DHS reserves the right to amend these standards periodically, with reasonable notice to participants.

1.3.1 Prior Approval Process, Period of Authorization, and PA and Non-PA Services

Since issuing HBTS Certification Standards in 2003, DHS made procedural changes to the prior approval process in order to facilitate the management of HBTS. This was achieved by removing some services from the Rhode Island Medical Assistance Prior Authorization Request Form (PA) while also requiring provider-agencies to list non-PA services when seeking approval for an HBTS Treatment Plan.

Provider-agencies must still submit all necessary information for approval from DHS or a CEDARR Family Center in order to be reimbursed for all services. With the above changes, Electronic Data Systems⁵ will notify provider-agencies that PA-covered services have been formally entered into the claims system.

Either DHS or a CFC will address prior approval for non-PA services at the time of clinical review. The clinical review process shall remain the basis for notification to provider-agencies regarding approval for all treatment services and hours requested per week.

Provider-agencies who do not submit required information regarding authorization of services shall not be reimbursed. Claims must be submitted within 365 days of service being provided.

1.3.1.1 Period of Authorized Services

The period of authorized service is stipulated in the prior authorization (PA) notice sent out by EDS. The maximum period for authorization of an HBTS Treatment Plan is six months unless otherwise directed by DHS and stipulated by EDS documentation. There is no limit to the number of HBTS treatment cycles that can be authorized by DHS or a CEDARR Family Center. A child can receive HBTS Specialized Treatment from only one provider-agency during a given

⁵ Electronic Data Systems (EDS) is the fiscal agent for DHS. EDS adjudicates all claims in accordance with DHS and Federal Medicaid policy and program rules.

authorization period. However, Pre-treatment Consultation and Group Intervention can involve more than one provider – agency at the same time.

1.3.1.2 Prior Authorization – PA Services

The PA stipulates the number of approved units for services by type. Reimbursement will be based on units of service provided as part of an authorized Treatment Plan. HBTS covered services on the Rhode Island Medical Assistances Prior Authorization Request Form are as follows:

- ☐ Home-Based Therapy Specialized Treatment
- □ Home-Based Therapy Treatment Support

The provider-agency has the responsibility to verify Medicaid coverage throughout a course of HBTS. Services delivered during a period of loss of Medicaid eligibility will not be reimbursed. Refer to Appendix 6 for additional information.

1.3.1.3 Non-PA Services

At the time of requesting HBTS, provider-agencies must complete the <u>HBTS Request Form for Non-PA-Services</u>. Refer to Appendix 21 for further explanation and a copy of this form.

1.3.2 Coordination of Care and HBTS

DHS recognizes that a child receiving HBTS from one provider-agency may for a variety of reasons end care and transfer services to another HBTS provider-agency. This most often takes place at the initiation of a child's parents or because the child's current provider-agency is recommending a different service. It is the responsibility of the child's current provider-agency to assist in an orderly transfer by collaborating and coordinating care with the new HBTS provider-agency. Coordination of Care can be provided through Treatment Plan Development. Written documentation must be maintained in support of ongoing coordination of care for HBTS.

1.3.3 HBTS Authorization Process (CEDARR and DHS)

Currently there are two paths to obtaining HBTS. One is through a Treatment Plan approved by a CEDARR Family Center and the other is through a Treatment Plan approved directly by DHS.

As of January 1, 2003, CEDARR Family Centers were given the responsibility to initiate <u>all new</u> HBTS requests to provider-agencies, provide clinical reviews for these referrals, and render authorizations for HBTS Treatment Plans. DHS has maintained responsibility for cases that it currently oversees unless a child's family subsequently sought participation from a CEDARR Family Center.

1.3.4 Coordination with CEDARR Family Center for Prior Authorization of HBTS

A family may contact a CEDARR Family Center for a variety of reasons, including recommendation by an HBTS provider-agency. Based on the child and family's interest, the CEDARR Family Center will conduct an Initial Family Assessment (IFA), working with the family to understand their special needs and circumstances, and review available options. As appropriate, the CEDARR Family Center develops a Family Care Plan (FCP) that may identify a range of specialized service options and providers, including recommendations for CEDARR Direct Services (e.g., HBTS, TCYC, and PASS).

If HBTS is identified as a direct service, the CEDARR Family Center will inform the family of available choices of certified provider-agencies for HBTS. The family, with guidance from the CEDARR Family Center, if desired, will choose the provider-agency from which they want to receive services. The CEDARR Family Center will provide information from its own assessment to the HBTS provider-agency to avoid duplication of effort and unnecessary repetitions by the family. This also minimizes the work otherwise required of the HBTS provider-agency to develop a proposed Treatment Plan.

In developing the proposed Treatment Plan, the HBTS provider-agency will conduct a more focused assessment directed toward determining the specifics of the proposed treatment plan. The CEDARR Family Center will review the proposed Treatment Plan and based on the clinical review and concurrence, the prior authorization is made. Appendix 2 provides a further outline of the CEDARR authorization process.

1.3.5 Coordination with DHS for Prior Authorization of HBTS

Until December 31, 2002, families could choose to directly obtain HBTS from a provider-agency with authorization from DHS. DHS, however, was not responsible for initiating referrals, determining the scope of care necessary for a child, or arranging for HBTS. DHS was solely responsible for providing clinical reviews relative to medical necessity and to process prior authorizations for HBTS. DHS also attempted to assist families by providing them with updated lists of available HBTS resources but was not capable of managing referrals.

After January 1, 2003, DHS required CEDARR Family Centers to address HBTS screening, referral and authorization on a voluntary basis for families currently receiving HBTS and for all families seeking HBTS for the first time. With the inception of CEDARR Family Centers, families became able to obtain a comprehensive set of services and resources in one place.

For those children whose HBTS remains overseen by DHS, the HBTS provider-agency in conjunction with the family prepares a proposed Treatment Plan that defines treatment goals and objectives, treatment intensity, hours of scheduled service, approach in working with the family, and other relevant considerations. The proposed Treatment Plan is then submitted to DHS for clinical review and authorization. On the basis of this clinical review and final action, a prior authorization is entered into the EDS system.

For the reauthorization of HBTS services for children who have had no contact with a CEDARR Family Center, DHS will continue to provide clinical review and approval for treatment for services until otherwise notified. Appendix 3 provides an outline of the DHS authorization process.

1.3.6 Commitment to Family Centered Care

The CEDARR Initiative seeks to incorporate the key elements of family centered, community-based care into practice. Participating providers of HBTS are required to develop practices and services consistent with the principles of family centered care. Core practices of family centered care include:

- 1. Incorporating into policy and practice the recognition that the family is the constant in a child's life, while the service system and support personnel within those systems fluctuate
- 2. Providing individualized services in accordance with the unique needs and potential of each child and guided by the child and family specific care plan that recognizes health, emotional, social, and educational strengths, as well as needs
- 3. Facilitating family/professional collaboration at all levels of hospital, home and community care
- 4. Exchanging complete and unbiased information between families and professionals in a supportive manner at all times
- 5. Incorporating into policy and practice the recognition and honoring of cultural diversity, strengths and individuality within and across all families, including ethnic, racial, spiritual, social, economic, educational and geographic diversity
- 6. Encouraging and facilitating family-to-family support and networking
- 7. Appreciating families as families and children as children, recognizing that they possess a wide range of strengths, concerns, emotions, and aspirations beyond their need for specialized health and developmental services and support
- 8. Ensuring services that enable smooth transitions among service systems and natural supports, which are appropriate to developmental stages of the child and family

Full disclosure to families of any anticipated delays in start of services, changes in personnel, and provider-agency policies and procedures in the provision of home-based services.

1.4 CEDARR Family Centers

CEDARR Family Centers have statewide capabilities with development of additional CEDARR Family Centers open to all interested parties. Appendix 16 lists all current CEDARR Family Centers and their respective satellite offices.

For additional information about CEDARR Family Centers and related activities for children with special health care needs and their families, please refer to the DHS website at www.dhs.ri.gov.

2.0 CERTIFICATION PROCESS

2.1 Submission of Certification Application for New Provider-Agencies

There is no limit to the number of entities that may become certified as provider-agencies of HBTS. Any interested parties may submit applications for certification at any time. All HBTS applicants will be evaluated on the basis of written materials submitted to DHS addressing these Certification Standards. DHS reserves the right to conduct on-site reviews and to seek additional clarifications prior to final scoring. Provider-agencies will be notified of their certification status when the review is complete. Applicants should anticipate a minimum of two months for the review process.

During the time that a provider-agency is preparing its application, DHS is available to provide technical assistance. Such guidance may also take place following a preliminary review of the applicant's application by DHS but not during the final scoring process.

If an applicant is expected to receive an unfavorable decision from DHS, DHS will notify the applicant and allow its application to be removed from consideration before a final written decision is made. If the applicant elects to remove its application, DHS is available to provide further comment and direction with copies of model applications available for onsite viewing.

There is no limit to the number of times that a potential applicant can seek to become a Certified CEDARR Direct Service Provider for HBTS.

2.2. Instructions and Notifications to Applicants

This document sets forth the Certification Standards for direct service providers of HBTS. In accepting certification from DHS, Certified CEDARR Direct Service Providers agree to comply with these certification standards as presently issued and as amended from time to time by DHS, with reasonable notice to providers.

These Certification Standards also provide the application guide for applicants. Sections 6 and 7 of this document identify the core standards against which applicants will be evaluated.

Within Sections 6 and 7, specific standards and expectations are identified. Applications will be scored on the basis of responses to each of these specific standards and expectations. Applications are to address each of these areas in the sequence presented. Applicants are to use the numbering system in these standards to identify the sections being addressed in the application.

An <u>Application Guide</u> is presented in Attachment A to guide the organization of application materials. Prior to technical review, submitted applications will be reviewed for completeness and for compliance with core expectations. Incomplete applications will be returned without further review.

Applicants are advised that all materials submitted to the State for consideration in response to these Certification Standards will be considered to be Public Records as defined in Title 38 Chapter 2 of the Rhode Island General Laws, without exception.

Interested parties are encouraged to contact the Center for Child and Family Health (CCFH) for further information and clarification. Letters of Interest are strongly encouraged to ensure that DHS is able to keep interested parties up to date regarding scheduled meetings or program clarifications that may be needed. Inquiries and completed applications should be directed to:

Anne M. Roach, RN, M.Ed. Consultant Public Health Nurse Center for Child and Family Health Department of Human Services 600 New London Avenue Cranston, Rhode Island 02920 Phone: (401) 462-6370

Once a provider is certified as eligible to provide HBTS, the provider shall be enrolled with EDS as a provider of these services. If you have any questions about the enrollment form or enrollment process, please call EDS at 1-800-964-6211.

2.3 Information for Interested Parties

Upon initial release of these CEDARR HBTS Direct Service Provider Certification Standards, DHS staff will be available to provide information for those pursuing certification applications. If appropriate, DHS will provide written addenda to these standards to further clarify certification requirements.

2.4 Certification

As set forth in these standards, certification as a HBTS provider is required in order for DHS to reimburse a provider agency for provision of HBTS services. Certification requires that provider-agencies adhere to these standards and performance expectations, as well as provide periodic reports to DHS. These Certification Standards include certain performance standards.

Following certification DHS will monitor the performance of certified HBTS provider agencies and their continued compliance with certification requirements. Certified agencies are required to notify DHS of any material changes in their organization's circumstances or in program operations. On the basis of ongoing monitoring, including review of required reports submitted by certified provider-agencies, DHS may identify deficiencies in performance and/or compliance with certification requirements. Based on such review and related communications, certification status may be modified to Provisional Certification. Fully certified and Provisionally Certified agencies will be reimbursed using different rate schedules (see Table 2 in Section 2.4.2 for Rate Schedules; see Section 2.5, "Continued Compliance with Certification Standards" for a fuller discussion of Provisional Certification).

2.4.1 Possible Outcomes of Certification Review Process

Certification applications will be reviewed and scored based on the degree to which an applicant demonstrates a program that complies with the requirements set forth in these HBTS Certification Standards.

Three basic outcomes are possible as a result of the application review process. These are:

- 1. Certification with no conditions
- 2. Certification with conditions
- 3. Not certified

As a result of the review, provider-agencies may be deemed in compliance with all requirements and be offered "Certification with no conditions." Alternatively, an applicant may describe a program that meets most of the Certification Standards, but for one reason or another does not fully comply with the certification requirements at the time of application submission. In such a case, the applicant may be offered "Certification with conditions" with application deficiencies identified by the State. The applicant must then submit to the State, a plan of corrective action plan with specific dates for addressing deficient areas of compliance. This plan must be accepted and approved by DHS.

In other cases the review team may determine that an application does not meet the requirements for certification and certification will not be offered to that agency. Deficiencies in the application will be identified. This will be done without prejudice and interested applicants will be encouraged to address deficiencies and submit an amended application. Certification is not a competitive process limited to a fixed number of providers. Rather, all applicants who demonstrate preparedness to comply with the standards will be certified.

2.4.1.1 Certification Offer

DHS will convene an application review committee to evaluate applications and submit recommendations for certification to the Associate Director, Division of Health Care Quality, Financing and Purchasing, Department of Human Services. Based on a positive action, a letter will be sent to the applicant provider-agency offering it certification and identifying any conditions to the certification. A signed acceptance of certification is required and must be returned to DHS in order for certification to commence

2.4.1.2 Period of Certification

The first period of certification began after February 1, 2003. Fourteen provider-agencies received Certification on a rolling admission basis. Certification letters were issued to each provider-agency after DHS received signed letters of agreement stipulating conditions and requirements necessary for Certification as a CEDARR Direct Service Provider. Certification was granted for up to a three-year period. Appendix 20 lists all HBTS certified provider-agencies and programs to date.

2.4.1.3 Recertification of HBTS Provider-Agencies

Extension of certification beyond the initial three-year period will be granted following a new certification application for an additional period. DHS also reserves the right to extend the period of certification beyond the three-years without application for one or more certified provider-agencies.

Appendix 18 stipulates recertification requirements and provides an abbreviated application guide.

2.4.2 Certification Status and Reimbursement Schedules

Reimbursement for services varies based on certification status. Table 1 lists the possible outcomes of the certification review process and related reimbursement rate schedule.

Table 1: Certification Status and Applicable Reimbursement Schedule				
Certification Status	Reimbursement Rate Schedule effective January <u>15</u> , 2003	Reimbursement Rate Schedule effective April 1, 2003		
Certified – with no conditions	Schedule A	Schedule A		
Certification – with conditions	Schedule A	Schedule A		
Existing provider agency, not certified	Schedule B	Not applicable		
Provisional Certification (applies only where a certified agency is deemed to be out of compliance with standards; provisional certification status cannot last longer than six months; see Section 2.5)	Schedule B	Schedule B		

Table 2 lists each of the HBTS services and the related schedules of reimbursement.

Table 2: Service Description and Schedules of Reimbursement

Treatment Related	Rate Schedule A	Rate Schedule B
Service Descriptions	(1 Unit = 30 minutes)	(1 Unit = 30 minutes)
HBTS- Specialized Treatment Worker	\$13.23	\$11.50
HBTS – Treatment Support Worker	\$10.50	\$9.50
HBTS Group Intervention (per child)	\$11.00	\$11.00

Table 2 Continued: Service Description and Schedules of Reimbursement

Clinical Supports	Rate Schedule A (1 Unit = 30 minutes)	Rate Schedule B (1 Unit = 30 minutes)
Clinical Supervision by Doctoral Level Staff	\$35.00	\$30.00
Clinical Supervision by Masters Level Staff	\$30.00	\$25.00
Clinical Supervision by Bachelors Level Staff	\$16.50	\$14.00
Treatment Consultation including Pre and Post treatment consultation by Doctoral Level Staff	\$35.00	\$30.00
Treatment Consultation including Pre and Post treatment consultation by Masters Level Staff	\$30.00	\$25.00
Specialty Consultation including Pre and Post treatment consultation by Masters Level Clinician		
in OT, PT, and SLP	\$30.00	\$25.00
Treatment Coordination (15 minute unit)	\$8.25	\$8.25
Lead Therapy	\$14.00	\$14.00

Table 2 Continued: Service Description and Schedules of Reimbursement

Administrative Supports	Rate Schedule A	Rate Schedule B
Child Specific Orientation – Newly Assigned		
HBTS Treatment Worker (max. 20 units)	\$40.00 per unit	\$40.00 per unit
Treatment Plan Development	\$280.00 per plan	\$280.00 per plan

2.5 Continued Compliance with Certification Standards

Certified HBTS provider-agencies shall comply with these HBTS Certification Standards throughout the awarded period of certification. Failure of DHS to insist on strict compliance with all certification standards and performance standards shall not constitute a waiver of any of the provisions of these certification standards and shall not limit DHS' right to insist on such compliance. DHS reserves the right to monitor and evaluate provider-agencies of HBTS for compliance with Medicaid and State laws, as well as these Standards and DHS regulations and policies pursuant to the management of HBTS. HBTS providers are required to provide periodic reports to DHS as identified in Section 6.6, "Service Monitoring and Reporting." For purposes of review, certified and provisionally certified providers will provide access to DHS and/or its agents at reasonable times to appropriate personnel and written records.

DHS reserves the right to apply a range of sanctions to provider-agencies that are out of compliance. These may include, but are not limited to:

1. Suspending new referrals:

a) Should DHS apply this sanction, the provider-agency must address all deficiencies in a written <u>Plan of Action</u> to DHS <u>within 10 business days</u> unless otherwise ordered by DHS. Failure to do so may result in immediate change in

certification status and/or other sanctions deemed to be appropriate by DHS. The following steps must then be followed:

- i. A Plan of Corrective Action must set forth clearly defined actions to remedy all identified deficiencies by stipulating objectives, personnel responsible for managing and achieving success, methods of remediation, and deadlines for the complete and successful resolution of all deficiencies.
- ii. DHS shall review the Plan of Action. DHS may then issue its acceptance or require additional clarifications or action steps from the provideragency. A modified Plan of Action must then be received by DHS within 5 business days. DHS will then review the written material and issue its findings.

DHS shall continuously monitor the provider-agency for compliance and at its own discretion determine whether it is appropriate to reinstate referrals. Under no circumstances shall suspension of referrals last longer than 6 months. Failure of a provider – agency to correct all deficiencies in a rapid and reliable manner within a 6-month period or sooner shall result in a change in certification status or issuance of other sanctions.

- 2. Change of certification status to Provisional Certification
- 3. Recoupment of funds when violations of Medicaid regulations, State law, or DHS policies, including these Certification Standards have taken place
- 4. Suspension of certification for severe violation of HBTS Certification Standards
- 5. Referral to appropriate legal authorities

2.5.1 Provisional Certification

As a result of its review activities, DHS may identify deficiencies wherein a provider-agency is not in satisfactory compliance with certification and/or performance standards. In such instance, DHS will notify the provider-agency in writing of such deficiencies, and will set forth a period of time whereby the provider-agency must come into compliance.

Provisional Certification shall commence on the one hundred twenty-first (121st) day following formal notice by DHS to the provider-agency identifying elements of non-compliance if the provider has not: a) cured the identified deficiencies, or b) submitted an appeal to DHS disputing the non-compliance, and setting forth the factual basis on which the notice of non-compliance is disputed, and having its appeal sustained by DHS. Such an appeal must be submitted no later than thirty (30) days following notice of non-compliance by DHS.

In the event that a provider-agency's appeal is not successful, the provider-agency may seek resolution through the Administrative Procedures Act (APA appeal) to Superior Court. In the event a provider-agency takes this action, imposition of Provisional Certification will be stayed pending the outcome of the appeal.

2.6 Licensure Requirements for Certified HBTS Provider-Agencies

A requirement for certification is that all clinical staff engaging in providing Clinical Supervision or Treatment Consultation, unless otherwise approved by DHS, must be health care professionals licensed by the Department of Health (DOH) in Rhode Island. In addition, licensed clinical staff must be able to demonstrate clinical competency⁶ to render Treatment Consultation or Clinical Supervision to home-based staff.

At the time of application, DHS must be provided with a complete list of clinical staff by discipline with license numbers.

All new applicants must meet licensure and competency requirements. No exceptions are permitted. Please refer to Appendix 4 for additional information.

2.6.1 Variance from Professional Licensure

In order to comply with Medicaid and State law, DHS formulated criteria necessary to accommodate conditions that would constitute a Variance from Professional Licensure for clinical staff with some provider-agencies that received certification in February of 2003 and thereafter.

2.6.1.1 Conditions for Variance

Exception and Exemption from Professional Licensure is the only means by which a Variance from Professional Licensure can be obtained. Appendix 4 provides a summary table identifying licensure and practice standards in addition to detailed written information.

The non-licensed clinician has the responsibility to fully comply with any stipulations associated with having been granted a Variance from Professional Licensure. Such stipulations are offered in writing to the provider-agency at the time of certification approval by DHS.

2.6.1.1.1 Holder of Variance

While the non-licensed clinician may become eligible to engage in the provision of Clinical Supervision or Treatment Consultation when approved, Variance from Professional Licensure is specifically given to the provider-agency. The provider-agency has the responsibility to ensure that clinical staff granted Exception or Exemption from Professional Licensure meet all requirements associated with each of these specific conditions.

2.6.1.1.2 Duration of Variance

Variance is awarded for a <u>three-year period of time</u>, based on the non-licensed clinician's continued employment with the requesting provider-agency. It can be renewed thereafter with no limitation on the number of times allowed.

There is no permanent exclusion or "life long" privilege that applies to variance.

2.6.1.1.3 Change in Status

Once a Variance from Professional Licensure has been issued by DHS, the individual clinician and requesting provider-agency must immediately notify DHS and address any changes that materially alter the basis for having been granted a variance. Most importantly, the individual clinician has to maintain and be able to demonstrate completion of required continued education hours (Appendix 4).

2.6.1.1.4 Portability of Variance from Professional Licensure

In May 2004, Exception and Exemption from Professional Licensure was made portable. Any individual for whom Exception or Exemption status was granted can have the variance apply if this individual is working at more than one HBTS provider-agency, or if the individual obtains new employment with another HBTS provider-agency.

DHS has continued to require each clinician's employer (i.e., provider-agency) to apply for Exception or Exemption status, and to maintain full compliance with DHS requirements (Appendix 4).

2.6.1.2 Exception from Professional Licensure

Exception from Professional Licensure is the category applied to specific non-licensed individuals at the time of HBTS Certification in February 2003. They were referred to as the Transition Clinician Group. To be eligible for Exception status and the ability to offer Clinical Supervision, an individual was required to have a minimum of a Bachelor's degree in a related field⁶ and meet all other requirements listed in Appendix 4. DHS then reviewed provideragency requests for non-licensed individuals to receive Exception status on a case-by-case basis. This was a time-limited allowance effecting specific individuals at the time of initial certification. Exception status does not permit a Bachelor's level clinician to provide Treatment Consultation.

2.6.1.2.1 Temporary Exception Status for License Eligible Health Care Professionals

Effective May 1 of 2004, DHS further clarified conditions under which non-licensed but license eligible candidates might be able to provide Clinical Supervision. An individual is licensed

⁶ Minimum of a Bachelor's degree in special education, child development, psychology, counseling, social work or nursing.

eligible if qualified to take a professional licensing examination offered by the Rhode Island Department of Health within 6 months of hire. <u>Exception status does not permit a license eligible Masters level clinician to provide Treatment Consultation.</u>

Provider-agencies may request consideration for Temporary Exception when an individual's license eligible status conforms to one of the conditions listed below, and fully meets all requirements pertaining to Temporary Exception Status listed in Appendix 4:

- 1. Temporary Exception Status for License Eligible Health Care Professionals permits provider-agencies to hire an individual who is *license eligible* to become licensed as: a Licensed Independent Clinical Social Worker (LICSW), Mental Health Clinician (LMHC), Marriage and Family Therapist (LMFT), or Psychologist
- 2. Temporary Exception Status for License Eligible Health Care Professionals also allows provider-agencies to hire an individual with the credential of a Licensed Clinical Social Worker (LCSW) from Rhode Island Department of Health
- 3. Under no conditions will DHS grant Temporary Exception Status for License Eligible Health Care Professionals if the license eligible candidate is unable to sit for licensing examination within 6 months of hire

2.6.1.2.2 Temporary Exception for Other Credentialed Professionals

The following categories apply on a case-by-case review for consideration of Temporary Exception for Other Credentialed Professionals:

- 1. An individual with a Masters Degree in Social Work and certification as a School Social Worker from Rhode Island Department of Education
- 2. An individual with certification from Rhode Island Department of Education as a School Psychologist
- 3. An individual with a Masters Degree in Special Education and certification from Rhode Island Department of Health in the following areas:
 - a) Early Childhood Special Educator (Birth through Kindergarten)
 - b) Special Educator for Elementary and Middle School Level
 - c) Special Educator for Middle and Secondary Level
 - d) Special Educator for Deaf/Hard-of-Hearing
 - e) Special Educator for Blind/Partially Sighted
 - f) Special Educator for Severe/Profound Disabilities
- 4. An individual with certification as a Counselor or Principal Counselor from Rhode Island Department of Mental Health, Retardation and Hospitals (MHRH):

MHRH certifications may be appropriate for consideration to receive Temporary Exemption Status for Other Credentialed Professionals. An individual must satisfactorily meet requirements stipulated by MHRH, which require at a minimum:

- a) Masters degree from an accredited program in counseling and
- b) One year full-time supervised clinical experience in a mental health setting or equivalent experience (required for Counselor) or
- c) Two years full-time supervised clinical experience in a mental health setting or equivalent experience (required for Principal Counselor)

2.6.1.3 Exemption from Professional Licensure

A clinician that meets qualifications for Exemption may also provide Clinical Supervision or Treatment Consultation. For the condition of Exemption from Professional Licensure to apply, DHS has continued to accept individuals with the following professional credentials:

- 1. Board Certified Behavior Analyst (BCBA)
- 2. Board Certified Associate Behavior Analyst (BCABA)

2.7 Clinical Oversight and Monitoring

As of January 1, 2003, DHS required that for each recipient of HBTS, a licensed and competent health care professional oversees all non-licensed staff providing Clinical Supervision and Treatment Consultation. This has not been changed.

Clinical oversight applies to individuals covered by Variance from Professional Licensure.

2.7.1 Licensed Clinician's Professional Oversight Responsibility

It is the responsibility of this licensed clinician to ensure the integrity of the Treatment Plan with respect to: identified goals and treatment objectives, intensity of treatment hours, type of interventions, and delivery of care during a period of HBTS intervention. Overall, this clinician assumes full responsibility for clinical services provided to a child and family. It is intended that clinician with oversight responsibility ensure that the non-licensed Clinical Supervisor is providing appropriate guidance and direction to the home-based workers in a competent and professional manner. The licensed health care professional must:

- 1. Be knowledgeable about the Treatment Plan's goals and its objectives
- 2. Be knowledgeable about methods of intervention and child's progress
- 3. Observe a supervision/consultation session twice within an authorization period, and document and describe this observation in the clinical record
- 4. Review written documentation of supervisory/consultation sessions on a monthly basis

- 5. Meet with the non-licensed staff that is providing clinical supervision or treatment consultation to address treatment concerns (frequency determined by the licensed health care professional)
- 6. The licensed clinician's efforts must be clearly and consistently documented
- 7. Review and sign all Treatment Plans

2.7.1.1 Expertise and Competence

The licensed clinician must have the expertise and competence to render professional oversight and must practice within the limits of his or her professional licensure.

It is insufficient and unacceptable to use individually licensed clinicians providing Specialty Consultations (i.e., OT, PT, and SLP) for the purposes of rendering clinical oversight for the entire HBTS Treatment Plan.

DHS has the sole discretion to invoke any and all remedies upon identification of performance deficiencies as well as unethical or illegal conduct.

Refer to Appendix 4 for further information regarding licensure and practice standard.

2.8 Home-Based Workers

2.8.1 February 2003 Certification Standards

Home-based workers hired after January 1, 2003 have been required to meet Certification Standards regarding staffing qualifications. While there have been no further modifications of standards regarding this particular group of home-based workers, DHS has maintained a practice of responding to provider-agency requests on a case-by-case basis.

Home-based workers employed by provider-agencies at the time of initial certification who did not meet certification requirements were allowed to continue in their current positions. This exception remained in effect only as long as the individual worker was employed by the provider-agency in that position. It could not be transferred to another agency or shared if the individual was working at more than one provider-agency. Each provider-agency had to request consideration for an individual working in two or more provider-agencies as a home-based worker.

2.8.2 New Accommodations

DHS informed provider-agencies, effective May 1 of 2004, that they could employ as home-based workers for both Specialized Treatment and Treatment Support individuals with at least 3 years of experience with adults experiencing developmental disabilities. The provider-agency

must offer the prospective HBTS worker training in Child Specific Orientation and adhere to training requirements as outlined in Section 6.5.2.1.2.

2.9 DHS Responsibilities

DHS has the responsibility to inform appropriate State agencies of any instances of fraud, suspected fraud, misuse of Medicaid funds, or professional misconduct.

As a Medicaid provider, the provider-agency is obligated to comply with all applicable State and Federal rules and regulations. Certified provider-agencies agree to comply with DHS program requirements. DHS reserves the right to amend program requirements from time to time, with reasonable notice to participating provider-agencies.

2.9.1 Oversight and Authorization

DHS or the CEDARR Family Center, in accordance with Medicaid regulations may, place limits on services (e.g., establish amount, duration, and scope of services) and exclude any item or service that it determines is not medically necessary, is unsafe, experimental, or is not generally recognized as an accepted method of medical practice or treatment.

3.0 BACKGROUND TO DEVELOPMENT OF HOME BASED THERAPEUTIC SERVICES

3.1 History

Since the mid-1990's, intensive HBTS have been provided to certain children under the provisions of Early Periodic Screening, Diagnosis and Treatment (EPSDT). HBTS represents an array of therapeutic services designed to reduce or ameliorate deficits in cognitive, communication, psychosocial, and physical functioning in children with special health care needs and is expected to maintain, stabilize and/or improve adaptive functioning. HBTS provides intensive treatment through the application of professional and research based interventions.⁷

DHS has consistently stated that "best practices" are required of HBTS providers. The November 1999 Guidelines stated that:

- 1. Provider-agencies of HBTS shall adhere to evidence based treatment approaches
- 2. Treatment interventions that are lacking scientific research and support will not be approved
- 3. Interventions that are of an "experimental" nature may be considered but the provider-agency will need to present scientific information to support proposed clinical interventions

HBTS are intended for a child living at home (or foster home) and participating in the community. HBTS vary with respect to scope of treatment services, treatment objectives, intensity of treatment hours provided, and duration of care. These factors are modified, as a child's needs change.

Since its inception, utilization of this service has increased substantially, involving a range of provider-agencies. Over this time, DHS has implemented various requirements to provide additional structure, quality assurance and accountability for services. These Certification Standards serve to both continue this process and to assemble all program requirements within a single document.

3.2 Additional Supports for Children with Special Health Care Needs

In July 2003, DHS convened a work group made up of parents, advocacy groups, HBTS provider-agencies, CEDARR Family Centers, and staff for the Center for Child and Family Health to address access to HBTS care. The project was referred to as the "HBTS Challenge."

Page 33

⁷ Refers to evidenced based treatment and consensus-based treatment protocols cited in professionally respected journals. While research articles are not immune to errors, built-in checks and balances include emphasis on objective data, independent replication, and critical peer review or research reports. Experimental or unsubstantiated treatment is not acceptable.

Its objectives were to identify new ways to provide services and supports to families with children with special health care needs in a more timely fashion. As a result of the HBTS Challenge, two major developments took place and are now in effect.

The first area of concentration related to the scope of services that HBTS provider-agencies could offer. Service descriptions were broadened to enable greater flexibility in responding to parents and children with special health care needs as follows:

- 1. Pre-Treatment Consultation became the mechanism to achieve initial support from a HBTS provider-agency prior to the development and delivery of an intensive treatment plan
- 2. Post-Treatment Consultation was designed to better help HBTS be more supportive of families whose course of services was concluding. Post-Treatment was designed to help families as they approach discharge from HBTS or transition to other services
- 3. Access to Group Intervention was also modified so that it could take place, when appropriate, prior to, or after, the conclusion of intensive treatment

The second area of concentration related solely to increasing the range of supports to families offered by the CEDARR Family Centers. CEDARR Family Center's Enhanced Basic Services and Supports include: Health Needs Coordination Counseling, Therapeutic Intervention Consultation, and Health Promotion and Maintenance Group Intervention. They are provided in the child's home, CEDARR Family Center, or community-based setting.

DHS also issued Certification Standards for two new services: Kids Connect (formerly known as Therapeutic Services in Child and Youth Care) and Personal Assistance Services and Supports (PASS). Kids Connect has certified provider-agencies engaged in working with children with special needs in a day care setting. PASS is a consumer-directed program where families recruit, train and manage their direct service worker. Referral to both services can be accessed through the CEDARR Family Centers.

Please refer to the DHS website at <u>www.dhs.ri.gov</u> for further information about programs or initiatives.

TARGET POPULATION AND LOCATION OF SERVICE WITHIN A 4.0 **CONTINUUM OF CARE**

4.1 **Eligibility**

The population eligible to be served by Home Based Therapeutic Services must meet all of the following criteria:

- Children who are Medicaid eligible from birth to their 21st birthday
 Children who are eligible for Medical Assistance through <u>Supplemental Security Income</u> (SSI), Katie Beckett (through age 18), Adoption Assistance, RIte Care, and RIte Share
- 3. Children who have potentially chronic (twelve months or longer in experienced or expected duration) and moderate to severe cognitive, developmental, medical and/or psychiatric conditions whose level of functioning is significantly compromised

4.2 Home Based Therapeutic Services Within the Continuum of Care; Appropriateness of this Level of Care

HBTS is an intensive outpatient service option within the continuum of care. It is not as restrictive as a day treatment program, inpatient hospitalization or residential level of care. HBTS is more intensive and of greater frequency than typical outpatient services. It may be provided in conjunction with other outpatient services as part of a coordinated program of care and can include other CEDARR Direct Services.

Appendix 5 includes a description of conditions typically associated with the target population receiving HBTS. Decisions regarding the appropriateness of HBTS need to take into consideration the criteria for HBTS set forth below.

4.2.1 **Clinical Appropriateness Criteria for Initiation of Service**

These criteria pertain to the initial determination of appropriateness. Treatment Plan approval requires all of the following criteria to be met and documented:

- 1. A formal diagnosis made within the last two years is required by an appropriately licensed health care professional with competence in child psychology, child psychiatry, or child development
- 2. The child demonstrates symptomatology consistent with a DSM-IV or ICD-9 diagnosis that requires, and on the basis of best available clinical and evidence based practice standards can be expected to respond to, HBTS intervention
- 3. The child presents with medical and/or physical condition(s) that require intensive therapeutic intervention

- 4. Outpatient services provided at an intensified level have not been sufficient and/or are inappropriate due to the child's special healthcare needs. However, this does not necessarily preclude from consideration the role of family therapy or other supports for a family that may be seeking HBTS
 - CEDARR Family Centers are best able to address the overall concerns of families and children when it comes to comprehensive treatment planning
- 5. There is evidence that the child requires a comprehensive and integrated program of medical and psychosocial services to support improved functioning at the least restrictive level of care
- 6. The child and family require support in order to remain stable outside of an inpatient environment, or to transition to independent living from a more restrictive setting
- 7. The child, the parents, or the individual's legal guardian, if appropriate, are willing to accept and cooperate with HBTS, including the degree of parental participation outlined in the HBTS Treatment Plan

In some instances, the following criteria may also apply:

8. The child may be at risk for hospitalization(s) or out-of-home placement without eventual use of HBTS. HBTS, however, is not intended to serve as emergency care

4.2.2 Clinical Appropriateness Criteria for Continuing Care

Reasons for a Treatment Plan at this level of care to be continued and/or reauthorized involve <u>all</u> of the following criteria:

- 1. Severity of condition(s) and resulting impairment continue to require this level of treatment
- 2. Treatment Planning is individualized and appropriate to the individual's changing condition with realistic and specific goals and objectives stated. The mode, intensity and frequency of treatment are appropriate and consistent with best known clinical and/or evidence based practice
- 3. Active treatment is occurring and continued progress toward goals is expected. Progress in relation to goals is clearly evident, measurable and described in observable terms
- 4. Treatment objectives have not yet been achieved; documentation supports continued interventions
- 5. The family is participating to the extent that it is medically and psychosocially capable or able

4.2.3 Discontinuing Services

Reasons for a Treatment Plan to be terminated can involve <u>any</u> of the following criteria:

- 1. Loss of Medicaid eligibility (Appendix 6: Provider-Agency Responsibility for Monitoring of Medicaid Eligibility)
- 2. Issues that may necessitate termination or temporary suspension of care during a period of authorized treatment include:
 - a) The child is at risk of harm to self or others, or sufficient impairment exists requiring a more intensive level of service beyond community-based intervention
 - b) The child's home environment presents safety risks to the staff making home visits. These include, but are not limited to: sexual harassment, threats of violence or assault, alcohol or illegal drug use, and health risks. The provider-agency has the legal responsibility to report suspected child abuse or neglect
 - The child, family, or guardian is not successfully following the provider-agency's program rules and regulations, despite multiple, documented attempts to address noncompliance issues
 - d) The provider-agency is unable to maintain staffing for an authorized Treatment Plan and, therefore, treatment goals cannot be adequately addressed during a period of care

All instances and circumstances that effect temporary suspension of services or termination are serious. Provider-agencies have the obligation to effect a smooth transition, whenever possible, and are required to conform to the rules and requirements stipulated by DHS. ⁸ A Termination Form must be completed and sent to DHS or to the referring CEDARR Family Center.

4.2.4 Discharge Criteria

Reasons to end HBTS can include any of the following criteria to end this level of care:

- 1. The child's documented Treatment Plan goals and objectives have been successfully met
- 2. The child no longer meets service initiation or continuing care criteria, or meets criteria for a less/more intensive level of care
- 3. The recipient at age 18, or his/her parents or legal guardian, withdraw consent for treatment

⁸ Provider-agency responsibility to discontinue treatment in a responsible manner is of paramount importance to DHS and families. Appendix 10 addresses situations and required processes for discontinuation or temporary suspension of HBTS.

4. Loss of Medicaid eligibility (Appendix 6: Provider Responsibility for Monitoring of Medicaid Eligibility)

4.3 Potential of Service and Limitations of Service

4.3.1 Potential of Service

HBTS is unique because services are delivered in a child's home and other age-appropriate community settings with direct and ongoing parent participation to maximize therapeutic effectiveness. As children learn to apply these skills within the family and age-appropriate community settings, HBTS enhances their abilities to actively participate as valued family and community members. Treatment consistency across settings and improvement in communication, behavior, psychosocial skills, and developmental functioning of a child are the goals of HBTS. This is achieved by maintaining a strong relationship between the CEDARR Direct Service Provider-agency and the CEDARR Family Center. As a result, a strong and mutually reinforcing integration of services for the child and family can be achieved.

4.3.2 Limitations of Service

DHS or the CEDARR Family Center reserves the right to determine that HBTS are being used appropriately to reach target populations. The degree of appropriateness will depend on the target population served and the individual needs of the child. The following guidelines shall be followed:

- 1. HBTS will not be used for respite or childcare
- 2. For many children it is expected that a course of outpatient treatment by a licensed mental health professional has been attempted prior to seeking HBTS
- 3. For children with psychiatric/behavioral conditions, a previous evaluation by an appropriately licensed mental health professional must have taken place within two years prior to the start of an initial HBTS Treatment Plan. During the course of HBTS care, an additional evaluation must take place within two years following the beginning of home-based treatment as needed
- 4. HBTS will not exist in isolation when other supports are indicated (e.g., family or individual psychotherapy, medical treatment, or school services and Early Intervention). HBTS are expected, when appropriate, to complement other services already in place for the individual. HBTS is not a substitute for mental health services provided by licensed professional clinicians
- 5. When children do not meet admission criteria for HBTS, other services should be investigated

- 6. HBTS will not take the place of services provided by Private Duty Nursing, Pediatric Home Care or the roles and responsibilities assigned to Certified Nursing Assistants (e.g., personal care, basic nursing skills, rehabilitation skills, care of patient environment, and recognition/reporting of symptoms). However, it is recognized that some children may require both HBTS and nursing care, including CNA services
- 7. Medicaid does not reimburse experimental and investigational treatments

5.0 SERVICE DESCRIPTION - REQUIRED SCOPE OF SERVICES

5.1 Service Name and Definition

Home Based Therapeutic Services are specialized health services delivered in a child's home and in community settings. HBTS represents an integrated set of service components involving the provision of Specialized Treatment provided in accord with an approved individualized Treatment Plan with measurable goals and objectives. All Treatment Plans must be approved by a designated licensed health care professional within the HBTS provider-agency. Once approved by a clinical reviewer at a CEDARR Family Center or DHS, the HBTS provider-agency can then begin to implement a 6 month Treatment Plan.

5.2 Service Components

An integrated HBTS Treatment Plan can include, in specified amounts as set forth in an approved HBTS Treatment Plan, the following reimbursable service components:

- 1. Treatment Consultation which may include the following:
 - a) Pre-Treatment and Post Treatment Consultation are separate from HBTS direct services
 - b) Pre-Treatment and Post Treatment Specialty Consultations (OT, PT, and SLP) are separate from HBTS direct services
 - c) Treatment Consultation with HBTS Specialized Treatment that can include Specialty Consultations
- 2. Clinical Supervision
- 3. Treatment Coordination
- 4. Home Based Therapy Direct Services:
 - a) HBTS Specialized Treatment
 - b) HBTS Treatment Support
- 5. Group Interventions
- 6. Child Specific Orientation for Newly Assigned Home-Based Worker
- 7. Treatment Plan Development

5.3 Units and Rates for HBTS Services

Rates of reimbursement are based on the certification status of the provider-agency. The provider-agency may be fully certified or provisionally certified based on compliance with Certification Standards, including reporting requirements and levels of performance, as stipulated in these standards.

Unless otherwise specified by DHS, provider-agencies will be reimbursed only for the unit of service actually delivered each month at allowable rates. A unit is one half hour of service (30 minutes) except for Treatment Coordination where one unit is 15 minutes of service.

5.3.1 Home-Based Therapeutic Service Units

HBTS provider-agencies are to comply with all established DHS billing reimbursement practices, and all future modifications as directed.

Table 3: Rate Schedule for HBTS

Table 3. Rate Sche		
Description	Rate Schedule A (1 unit = 30 minutes)	Rate Schedule B (1 unit = 30 minutes)
Treatment Consultation by Doctoral Level Staff		
Treatment consultation by Doctoral Level Stair	\$35.00	\$30.00
Treatment Consultation by Masters Level Staff	700100	70000
	\$30.00	\$25.00
		·
Treatment Specialty Consultation: OT, PT, & SLP	\$30.00	\$25.00
Clinical Supervision of Home-Based Worker by		
Doctoral Level Staff	\$35.00	\$30.00
Clinical Supervision of Home-Based Worker by		
Masters Level Staff	\$30.00	\$25.00
Clinical Cunowisian of Home Deced Worker, her		
Clinical Supervision of Home-Based Worker by Bachelors Level Staff	\$16.50	\$14.00
Dacheiors Level Staff	\$10.50	φ14.00
Clinical Supervision of Treatment Support Worker by		
Doctoral Level Staff	\$35.00	\$30.00
Clinical Supervision of Treatment Support Worker by	¢20.00	Φ 25 00
Masters Level Staff	\$30.00	\$25.00
Clinical Supervision of Treatment Support Worker by Bachelors Level Staff	\$16.50	¢14.00
Bachelors Level Staff	\$16.50	\$14.00
Home Deced Thomasy, Casciolized Treatment	\$13.23	\$11.50
Home Based Therapy - Specialized Treatment	\$13.23	\$11.30
Home Bosed Thomasy Treatment Connect	\$10.50	\$9.50
Home Based Therapy - Treatment Support	\$10.50	\$9.50
Group Interventions	\$11.00	\$11.00
Group Thier vehicions	φ11.00	φ11.00
Treatment Coordination (15 minute unit)	\$8.25	\$8.25
Child Specific Orientation - Newly Assigned Worker	\$40.00 for each orientation	\$40.00 for each
(max 20 units)	session	orientation session
(max 20 umus)	30331011	011011441011 50551011
Treatment Plan Development	\$280.00	\$280.00
Treatment I fan Development	Ψ200.00	Ψ200.00
Lead Therapist	\$14.00	\$14.00
Aller American	Ψ11100	Ψ1 1.00

5.4 Description of Service Components – Reimbursable Services

Table 4 provides a description of service components, required personnel qualifications, rates, range of approved hours, and functions assigned to each service component. HBTS reimbursable services are as follows:

Table 4: HBTS Reimbursable Services

		Table 4: HB18 Kelmbursa		5
HBTS Service Name	Personnel	Qualifications	Approved Units per plan (1 unit=30 minutes)	Comments
Treatment Consultation	Treatment Consultant	 Licensed health-care professional with established competencies in working with children with special health care needs * Masters or Doctoral degree 	Minimum None Maximum 2 units per week	 Not the same person as clinical supervisor Provides specific expertise and direction to therapeutic regimen. Conducts functional behavior assessments Can be episodic or ongoing Provides direction for emergency situations Documents consultation Must be child-specific sustained activity greater than 15 minutes in duration
Treatment Consultation: Occupational Physical & Speech and Language Therapies	Occupational Physical or Speech Therapist	Licensed OT, PT or SLP with established competency in working with children with special health care needs*	Where applicable, Minimum = 2 units per month Maximum = 4 units per month	 Writes OT, PT, SLP goals and objectives for treatment plan, in coordination with child's IEP or IFSP Instruct home-based workers on proper implementation of treatment interventions Observe home-based workers treating the child on a monthly basis Documents using specialty consultation forms Must be child-specific sustained activity greater than 15 minutes in duration
Clinical Supervision for Home- Based Worker	Clinical Supervisor	 Licensed health-care professional with established competency in working with children with special health care needs * Masters or Doctoral degree Individuals operating under DHS-authorized exception 	Minimum = 2 units per week Maximum = 4 units per week	 Individual or group supervision if more than 2 home-based workers working with child Responsible for the development of Treatment Plan and writing of goals and objectives Instruct home-based workers on proper implementation of treatment interventions Observe home-based workers treating the child on a monthly basis Provides direction for emergency situations Documents supervision

Table 4 Continued: HBTS Reimbursable Services

	lan	le 4 Continuea: HB18 Rein		
HBTS Service Name	Personnel	Qualifications	Approved Units per plan (1 unit=30 minutes)	Comments
Clinical Supervision for Home- Based Worker	Clinical Supervisor			Must be child-specific sustained activity greater than 15 minutes
Lead Therapy	Lead Therapist	 Be at least 19 years of age; have a high-school degree or equivalent, and two years of supervised experience working with children with special health care needs, or Have an Associates degree in human services (i.e., psychology, counseling, child development, education or nursing, etc.), or Be currently enrolled in not less than six (6) semester hours of relevant undergraduate coursework at an accredited college or university, or Demonstrate competency to work with children with special health care needs as evidenced by active participation in agency – specific formal training with completion of objective testing within 12 months of hire, and 	Determined on a case by case basis	 Participate in the development of an HBTS plan Participate in the development of instructional strategies Prepare instructional materials Observe the treatment worker and provide guidance Work with families to apply instructional strategies Maintain the integrity of a Treatment plan, that is, look to observe on a daily basis how a child is succeeding. This involves data collection and data management Provide emergency coverage as an HBTS treatment worker, if indicated Provide direct instruction to a child when needed
Child Specific Orientation	Treatment Consultant Clinical Supervisor Experienced Home-Based Worker	 Licensed health- care professional with established competencies in working with children with special health care needs * Masters or Doctoral degree, or Home-based worker with a minimum of 2 years related experience 	Maximum = 20 units per child per Treatment Plan	Used to orient and train new worker to the child's Treatment Plan Can include 1:1 supervision and observing experienced home-based workers treating a child

DELETE

THIS

PAGE

HBTS Service Name	Personnel	Qualifications	Approved Units per plan (1 unit=30 minutes)	Comments
Home-Based Specialized Treatment	Home-Based Treatment Worker	 19 years old; highschool degree or equivalent, minimum 2 years of supervised experience working with children with special health care needs, or Associates degree in human services, or Currently enrolled in not less than six (6) semester hours of relevant undergraduate coursework at an accredited college or university, or Previous work related experience with adults with developmental disabilities – DHS approval required, and As outlined in 6.5.2.1.2, active participation in an agency-specific formal training program, approved by DHS, and successful completion of objective testing within twelve (12) months of hire 	Minimum = 10 units per week Maximum = 80 units per week	 Intensive treatment provided in the home and/or community setting Implement child's individualized Treatment Plan Home-based worker collects data on responses to interventions for each treatment goal and objective
Treatment Support	Home-Based Treatment Worker Home-Based Support Worker	 19 years old, high-school graduate or equivalent, minimum plus 1 year of supervised experience working with children, or Associates degree in human services 	Determined on a case by case basis	 Less intensive services, which provide structure, supervision, guidance and redirection to a child in the home or community Component of direct treatment service plan Staff trained on interventions and behavior program used with child Data kept during shift for targeted behaviors

Table 4 Continued: HBTS Reimbursable Services

HBTS Service Name	Personnel	Qualifications	Approved Units per plan (1 unit=30 minutes)	Comments
Clinical Supervision of Treatment Support Worker	Clinical Supervisor	Masters or Doctoral degree		 Documents supervision Must be child-specific sustained activity greater than 15 minutes in duration
Treatment Coordination	Treatment Coordinator	Bachelors degree	Maximum = 8 units per week	 Activities conducted on behalf of a specific child to ensure coordination with all relevant caregivers and others involved in child's plan of care Collects and manages data for summary Reports

• Licensure is rendered by the Department of Health (DOH) with relevance for HBTS in the areas of: psychology, social work (LCSW & LICSW), marriage and family therapist, mental health counselor, registered nurse (Masters degree required by DHS), occupational therapist, speech and language therapist, and physical therapist). Competency is established by formal education credits, internships, work history and supervised practice.

Group Intervention is an HBTS Specialized Treatment. Section 5.4.4.1 provides specific information that supports the delivery of this treatment. Group Intervention reimbursable services are listed below.

 Table 4 Continued:
 HBTS Reimbursable Services Summary for Group Intervention

HBTS Service	Personnel	Qualifications	Approved	Comments
			Units	
Group	Treatment	 Licensed health- care 	Maximum =	 Facilitates acquisition of socially
Intervention	Consultant	professional with	twice a week	appropriate behaviors in group setting
Therapist		established competency	for up to 3	
	or	in working with	units (1 unit =	Utilizes recognized curriculum and
		children with special	30 minutes)	instructional techniques
	Clinical	health care needs *	per session of	Maximum 8 children with 2 staff
	Supervisor	Masters or Doctoral	Group	
		degree	Intervention	
		8	per child	
Treatment	Treatment	Bachelors Degree	Maximum = 1	• Refer to Treatment Coordinator
Coordination	Coordinator		unit per child	responsibilities
			per week (unit	•
			= 15 minutes)	

Pre-Treatment Consultation Services can precede the delivery of home based hours. Table 5 describes Pre-Treatment Consultation Services below:

Table 5: Pre-Treatment Consultation – Reimbursable Services

TIDEG C	Table 5:	Fre-Treatment Consultat	1	
HBTS Service	Personnel	Qualifications	Approved	Comments
Name			Units per plan	Prior Authorization Required for Each Service
			(1 unit=30	6 – month authorization period
			minutes	0 – month authorization period
Pre-Treatment	Treatment	Licensed health-care	Maximum =	Provides direct support and
Consultation	Consultant or Clinical Supervisor	professional with established competency in working with children with special health care needs * Masters or Doctoral degree and Staff who have met DHS variance criteria	12 units per week for first month Maximum = 4 units per week for the remaining 5 months	 information to families of children on the HBTS referral list Teach parents specific therapeutic interventions to reduce a child's challenging behaviors and improve child's functional skills Conducts functional assessments Not crisis intervention Documents consultation Takes place in child's home or
				provider-agency
Clinical Supervision for non-licensed Treatment Consultant	Clinical Supervisor or Treatment Consultant	Licensed health-care professional with established competencies working with children with special health care needs*	Maximum = 2 units per week for non- licensed staff	 Responsible for assisting in the development of Pre-Treatment Consultation Plan and writing of goals and objectives Provides supervision to non-licensed treatment staff on issues arising in treatment Documents supervision
Treatment Coordination	Treatment Coordinator	Bachelors Degree	Maximum = 4 units per week for the first month Maximum = 2 units per week for remaining 5 months	 Activities conducted on behalf of a specific child and family to ensure coordination with all relevant caregivers and others involved in child's plan of care Collects and manages data for summary reports

^{*} Licensure is rendered by the Department of Health (DOH) with relevance for HBTS in the areas of: psychology, social work (LCSW & LICSW), marriage and family therapist, mental health counselor, registered nurse (Masters degree required by DHS), occupational therapist, speech and language therapist, and physical therapist). Competency is established by formal education credits, internships, work history and supervised practice.

Post-Treatment Consultation Services begin when home-based specialized treatment (including treatment support) hours are no longer being supplied to a family. Post-Treatment Consultation can be provided, when requested, by families to support gains and/or assist in transition to other services. Table 6 describes reimbursable services associated with Post Treatment Consultation:

Table 6: Post-Treatment Consultation- Reimbursable Services

HBTS Service Name	Personnel	Qualifications	Approved Units per plan (1 unit=30 minutes	Comments Prior Authorization Required for Each Service 6 – month authorization period
Post Treatment Consultation	Treatment Consultant Or Clinical Supervisor	 Licensed health-care professional with established competencies in working with children with special health care needs * Masters or Doctoral degree and Staff who have met DHS variance criteria 	Maximum 2 units per week	 Continues to provides direct support and information to families Continues to review and refine therapeutic interventions to maintain gains and progress made during HBTS Not crisis intervention Documents consultation Takes place in child's home or provider-agency
Post Treatment OT, PT, SLP Therapies	Occupational, Physical And/Or Speech Therapist	Licensed OT, PT, or SPL with established competency in working with children with special health care needs*	Maximum 1 unit per week	 Writes OT, PT, or SLP goals and objectives for Post-Treatment Consultation Plan, in coordination with child's IEP or IFSP Child must have received OT, PT, SLP consultation by HBTS provideragency Reviews with parents proper implementation of treatment interventions Documents consultation using specialty consultation form
Clinical Supervision for non-licensed Treatment Consultant	Clinical Supervisor Or Treatment Consultant	Licensed health-care professional with established competencies working with children with special health care needs*	Maximum 1 unit per week for non- licensed staff	 Responsible for assisting in the development of Post-Treatment Consultation Plan and writing of goals and objectives Provides supervision to non-licensed treatment consultant on issues arising in treatment Documents supervision
Treatment Coordination	Treatment Coordinator	Bachelors Degree	Maximum =1 unit per week	 Activities conducted on behalf of a specific child and family to ensure coordination with all relevant caregivers and others involved in child's plan of care Collects and manages data for summary reports

Table 6 Continued: Post-Treatment Consultation- Reimbursable Services

HBTS Service Name	Personnel	Qualifications	Approved Units per plan (1 unit=30 minutes	Comments Prior Authorization Required for Each Service 6 – month authorization period
Treatment Coordination	Treatment Coordinator	Bachelors Degree	Maximum =1 unit per week	 Activities conducted on behalf of a specific child and family to ensure coordination with all relevant caregivers and others involved in child's plan of care Collects and manages data for summary reports

[•] Licensure is rendered by the Department of Health (DOH) with relevance for HBTS in the areas of: psychology, social work (LCSW & LICSW), marriage and family therapist, mental health counselor, registered nurse (Masters degree required by DHS), occupational therapist, speech and language therapist, and physical therapist). Competency is established by formal education credits, internships, work history and supervised practice.

5.4.1 Prior Approval for HBTS

Prior Approval (PA) is required for all components of HBTS. When Certification Standards were first issued in 2003, DHS required all components of an HBTS Treatment Plan to be listed on a PA for authorization. In November 24, 2004 DHS made modifications to the prior authorization process by removing Treatment Coordination and Child Specific Orientation from the PA when submitting an HBTS Treatment Plan. DHS further modified the authorization process by removing additional services from PA in May of 2005. This was done to facilitate the management of claims and assist provider-agencies in reconciling accounts in a more timely and efficient manner.

Effective May 1 of 2005, the following services are covered by Prior Authorization and non-PA services, namely:

Prior Authorization Required Services

Service	Procedure Code
Home Based – Specialized Treatment	T1024 TG
Home Based – Treatment Support	T1024 TF

Non-PA Services

Service	Procedure Code
Child Specific Orientation	X0801
Clinical Supervision Master Level Clinician	X0800 HO
Clinical Supervision Doctoral Level Clinician	X0800 JF
Group Intervention	X0802
Lead Therapy	X0800
Pre-Treatment Consultation	S4539
Post-Treatment Consultation	S4539
Specialized Treatment Consultation – Occupational Therapist	S4539
Specialized Treatment Consultation – Physical Therapist	S4539
Specialized Treatment Consultation – Speech & Language Therapist	S4539
Treatment Consultation	S4539
Treatment Coordination	T1016
Travel Time (refer to Appendix 22 for additional information)	X0188

5.4.1.1 Prior Approval and Authorization Process

The following steps are required to facilitate the Prior Approval and Authorization process. Appendix 21 contains additional information and corresponding forms.

- 1. The provider-agency must submit a child's Treatment Plan to either DHS or a CFC. This is determined based on the child's status as either a CFC referral or a client for whom DHS has managed previous Treatment Plans
- 2. The provider-agency must also complete and attach to either DHS or a CFC the following two forms:
 - a) The <u>Rhode Island Medical Assistance Prior Authorization Request Form</u> (PA) for HBTS utilizing a PA.
 - b) The HBTS Request Form for Non-PA Services

5.4.1.2 Modifications to Prior Authorization Requests

In exceptional circumstances, DHS recognizes that at the beginning of a new plan or during a course of treatment extenuating circumstances may warrant a time-limited increase in Clinical Supervision and/or Treatment Consultation (e.g., discharge from hospital or residential placement, traumatic events). Additional modification to HBTS Specialized Treatment and Treatment Support hours will be considered upon request to DHS or a CEDARR Family Center.

The provider-agency has the responsibility to document and inform the CFC or DHS of these circumstances and request additional clinical supports. Please include the request for additional units for consideration by the clinical reviewer.

5.4.1.3 Alignment of CEDARR Family Care Plans and HBTS Treatment Plans

Effective August 1 of 2004, all HBTS Treatment Plan service dates had to be aligned to match the CEDARR Family Center's Family Care Plan. This was done in order to coordinate care for any child receiving CEDARR Direct Services. HBTS provider-agencies received an authorization letters from EDS with the aligned dates listed. The following situations may have applied:

- 1. If the authorization for services was less than <u>five</u> months, provider-agencies did not need to resubmit a Treatment Plan at the end of the authorization period. A staff member from the CEDARR Family Center will have called the contact person for the HBTS Treatment Plan to review treatment progress and obtain information for the next 6-month period of HBTS care. CEDARR Family Center staff may also request all pertinent information in writing. If care is to continue, the CEDARR Family Center will issue another Prior Authorization for 6-months.
- 2. If authorization letter from EDS was not received by the end of the first month of the treatment period, the HBTS provider–agency should contact EDS to check on the status of the Prior Authorization. If needed, a representative from the HBTS provider-agency may call the CEDARR Family Center to address PA questions with its representative to clarify and resolve matters.

5.4.2 Clinical Supervision

5.4.2.1 Clinical Supervision of Home-Based Treatment Workers

Clinical Supervision is a required component of HBTS. It serves to ensure effective implementation and oversight of the Treatment Plan. It is the responsibility of the provideragency to maintain clinical supervision through out a Treatment Plan.

Licensed health care professionals, who have competence and experience working with the population being served, can only provide clinical supervision unless otherwise approved by DHS. Competence is evidenced by clinical internship, post-doctoral or advanced degree training, and ongoing continuing education with concentration in areas of study appropriate to the target population.

The services of the Clinical Supervisor must be documented in writing with respect to date, duration of supervision, which home-based worker received supervision, and reflect sufficient content to substantiate the delivery of this service. Clinical Supervision allowed for Specialized Treatment is 1 hour (minimum) to 2 hours (maximum) per week per plan. Table 7 summarizes the scope and parameters of clinical supervision for the Home Based Treatment Worker with reference to required and non-reimbursable activities.

Table 7: Clinical Supervision for Specialized Treatment

Required Activities	Non-Reimbursable Activities
 Observe worker in the home with the child implementing the Treatment Plan on a monthly basis Model techniques for staff and/or work with the child Instruct workers on proper implementation of treatment interventions Assess efficacy of treatment and address clinical issues Assist in development/revisions of the Treatment Plan and writing of goals and objectives Communication and collaboration with others (e.g., school personnel, OT, PT, SLP consultants) regarding treatment Attend IEP or IFSP meetings when appropriate in order to maintain or modify Treatment Plan In person consultation to home-based worker and family Provide group supervision when there are two or more home-based workers treating a child. Group supervision is necessary to maintain optimal communication and ensure consistent implementation of treatment. 	 Agency administrative meetings Telephone supervision, except in emergency situations Telephone consultation In person or telephone discussions relating to administrative issues

DHS recognizes that an individual engaged in providing clinical supervision may also have a caseload of children to address. It is not the intention of DHS to propose a ratio to govern the number of cases a Clinical Supervisor may carry. However, DHS does require that the provideragency and the Clinical Supervisor show discretion and engage responsible ethical practice.

⁹ Exception and Exemption from professional licensure is a one-time occurrence per individual clinician and subject to the requirements listed in Appendix 4.

This means that the Clinical Supervisor will have adequate time to provide Clinical Supervision while remaining available and responsive to home-based staff and families. This also applies to any individual who may serve in the capacity of as Treatment Consultant for a provider-agency. Any pattern of conduct by a Clinical Supervisor or Treatment Consultant providing more than 30 hours of service per week, unless approved by DHS, shall be subject to review by DHS.

Additionally, an individual offering clinical supervision may not also provide treatment consultation. It is required that separate, qualified staff members assume the role of Treatment Consultant and Clinical Supervisor for each case.

It is the responsibility of the Clinical Supervisor to educate the home-based staff on issues of domestic violence, substance abuse and risk to child welfare, harassment of home-based staff or any other serious circumstances that may compromise or interfere with treatment.

5.4.2.2 Clinical Supervision of Treatment Support Worker

Clinical Supervision is required to ensure that treatment support provides structure, supervision, guidance, and redirection for children whose developmental condition and level of functioning prevents continuous participation in direct treatment. The amount of clinical supervision allowed for Treatment Support is 1 unit (minimum) to 2 units (maximum) every week. Table 8 lists the parameters for the supervision of the Home-Based Support Worker.

Table 8: Clinical Supervision for Home-Based Treatment Support

Required Activities	Non-Reimbursable Activities
Individual or group supervision	Agency administrative meeting
Review changes in Treatment Plan	• Telephone supervision, except in emergency situations
Address challenging behavior(s)	Telephone consultation
Review functional domains	 In person or telephone discussions relating to administrative issues

5.4.3 Home-Based Therapy Direct Services

Specialized Treatment and Treatment Support make up Home-Based Therapy Direct Services. Specialized Treatment and Treatment Support are provided to a child by a home-based worker (paraprofessional) in accordance with the approved Treatment Plan, and under the supervision of the licensed clinical supervisor.

5.4.3.1 HBTS – Specialized Treatment

Specialized Treatment addresses the development of behavioral, communication, social, and functional skills based upon the application of scientifically established procedures from special education, child psychiatry, clinical psychology including behavior analytic therapy, and child development. The focus of treatment can include: increasing language, improving attention to tasks, enhancing imitation, generalizing social behaviors, developing independence skills, decreasing aggression or other maladaptive behaviors, and learning problem solving skills (e.g., organization, conflict resolution, and relaxation training). Treatment interventions are provided on a continuous basis in the home and community for an approved number of hours per week.

The home-based worker providing Specialized Treatment is responsible for implementing the Treatment Plan in the child's home and/or community. This individual(s) collect data on responses to interventions for each treatment goal and objective and receive weekly supervision by the Clinical Supervisor, and treatment consultation monthly by the physical therapist, occupational therapist and/or speech and language pathologist.

5.4.3.1.1 Specialty Consultations:

For certain children, it may also be appropriate, when authorized, to include supports for occupational therapy (OT), physical therapy (PT), and/or speech and language therapy (SLP) to reinforce relevant skill development in each of these areas. These support activities can only be considered if the child has a current IEP or IFSP providing OT, PT or SLP.

The goals and objectives included in an HBTS Treatment Plan must be consistent with the goals of the IEP or IFSP. All goals and objectives included in an HBTS Treatment Plan must be based upon the written recommendations from the licensed Occupational Therapist, Physical Therapist or Speech Pathologist consulting to HBTS.

Specialty consultations are not substitutes for therapy or services provided by licensed therapists in public schools, Early Intervention programs or other settings. No more than 25% of weekly treatment hours can be spent on OT, PT and/or SLP goals.

5.4.3.1.1.1 Occupational Therapy Consultation

This service includes goals, objectives, and activities to address the functional needs of a child related to adaptive development, adaptive behavior and play as well as sensory, motor and postural development. These services are designed to improve the child's functional ability to perform tasks in home and community settings, (e.g., feeding and eating, toileting, assisting with dressing/undressing, assisting with grooming, oral hygiene, bathing, functional communication, play skills, and community mobility).

5.4.3.1.1.2 Physical Therapy Consultation

This service includes goals, objectives, and activities to promote sensory-motor function through enhancement of musculoskeletal status, neurobehavioral organization, perceptual and motor development, cardiopulmonary status, and effective environmental adaptation. These services are for the purposes of increasing functioning during the child's natural activities and routines.

5.4.3.1.1.3 Speech and Language Therapy Consultation

This service includes goals, objectives, and activities for the habilitation or rehabilitation of communicative or oropharyngeal disorders and delays in development of communication skills. These services are to increase the functional and meaningful communication of the child by engaging in learning opportunities occurring in the home and community settings.

5.4.3.2 HBTS – Treatment Support

Treatment Support is a less therapeutically intensive component of home-based direct treatment services. Treatment Support is provided as a complement to Specialized Treatment. The request to include Treatment Support as part of an HBTS Treatment Plan may be initiated by parents, the provider-agency or a CEDARR Family Center.

The goal of Treatment Support is to assist children with moderate to severe developmental and neuro-medical conditions whose level of functioning limits their participation and ability to engage in sustained Specialized Treatment. Treatment Support is intended to facilitate some children's transition into adulthood by supporting a child's ability to remain at home and to participate in the community. It encourages and facilitates activities of daily living by providing structure, supervision, guidance, and redirection while engaging in cognitive, physical and social activities that would be typical for a child their age. Accordingly, one or more of the following domains must be addressed (Appendix 7):

- 1. The child's ability to acquire and use information
- 2. The child's ability to attend and complete tasks
- 3. The child's ability to interact and relate with others
- 4. The child's ability to care for himself
- 5. The child's ability to maintain his health and physical well-being, which includes participation in community activities

Treatment Support can be provided by the same individual providing Specialized Treatment or by an individual who meets the qualifications for a home-based Treatment Support worker. The home-based Treatment Support Worker is trained on interventions and behavioral approaches used during Specialized Treatment. The occurrence of target behaviors will be monitored and recorded during shift hours.

The rationale for seeking Treatment Support must be clearly articulated and linked to its intended purpose and domains of focus. Treatment Support services is not for the convenience of others

or to be substituted when other more appropriate services are indicated (e.g., respite, certified nursing assistance, or child care).

As part of an approved HBTS Treatment Plan, Specialized Treatment can be provided to a child attending a summer program as long as there are identified treatment goals and objectives. Treatment Support in conjunction with HBTS Specialized Treatment must be part of the approved plan. Treatment Support cannot be the only service.

5.4.4 Group Interventions

Group Intervention is intended to address the development of communication, anger management, problem solving, and interpersonal or social skills. This therapeutic service is conducted in a group setting and uses recognized instructional techniques.

Prior to implementing a group program, the provider – agency must complete the HBTS Group Intervention Information Sheet and submit this to DHS. DHS will review and respond to requests to provide group intervention. This information is required prior to the implementation of new group intervention programs.

DHS also recognizes that if a child requires 1 to 1 staffing in order to effectively participate in the group, the provider–agency may submit an authorization request to include a home-based treatment worker for the group intervention time. Criteria for 1 to 1 staffing includes the following: the need for ongoing re-direction to attend to the group leader, and/or to participate in group activities relative to a child's developmental level or requires frequent explanations of group's requirements or staff prompts to use relaxation or other self-management techniques to avoid disrupting the group. Such staffing is limited to a child's participation in Group Intervention being offered by the provider-agency as part of an HBTS Treatment Plan.

Transportation to group may also be provided if medically necessary and consistent with HBTS standards.

A child may received Group Intervention from one HBTS provider-agency and also receive HBTS specialized services and/or pre or post treatment from a different HBTS provider-agency. Each service requires a separate authorization and collaboration between the provider-agencies is required.

5.4.4.1 Group Intervention Service Supports

Relevant service components that support the delivery of group intervention include:

1. Clinical supervision by a licensed health care professional is required for clinicians approved under the variance criteria. The provider–agency is responsible to ensure that these non-licensed clinical staff has the expertise and training to offer group interventions. Clinical supervision is allowed up to .5 units per week per child.

- 2. Treatment coordination is available up to 1 unit per week per child.
- 3. When appropriate, Specialty consultation by an Occupational, Physical or Speech Therapist may be included as a support to a maximum of 1 unit per week per child

DHS recognizes that a group constitutes a minimum of 2 participants with the provider-agency having the authority to determine appropriate size. Group intervention can occur 1-2 times per week for up to 90 minutes per session (3 units). Models that go beyond this standard require discussion and approval by DHS. Group can take place at the provider – agency or community setting.¹⁰

Group therapy can only be provided by a licensed health care professional unless otherwise approved by DHS in the case of a clinician who enjoys Variance from Professional Licensure.

The process of approval for a 6-month period of authorization, which may be renewed if appropriate, requires completion of the following information to be sent to either DHS or a CEDARR Family Center 30 days in advance depending on the child's referral list status:

- 1. HBTS Treatment Plan Data Sheet
- 2. A parent's signature indicating acceptance of treatment
- 3. A physician's referral for HBTS
- 4. Prior Authorization Request Form
- 5. Copy of the HBTS Group Intervention Treatment Plan
- 6. For a renewal of Group Intervention, a summary of progress is also required

The provider-agency has the responsibility to document and summarize the child's response to Group Intervention at the conclusion of care. It is expected that this will address transition planning or discharge needs.

5.4.5 Child Specific Orientation for Newly Assigned Home-Based Workers

The State recognizes that for various reasons, HBTS provider-agencies may experience staff turnover and need to recruit new home-based workers to implement Treatment Plans (i.e., Specialized Treatment and Treatment Support Workers). A newly assigned staff person is expected to be knowledgeable about a child's condition, treatment approaches and past treatment history. Child Specific Orientation assures that the home-based worker is prepared to carry forward the goals, objectives and techniques of the child's specific Treatment Plan.

DHS regards the effectiveness of treatment to be related to the availability and continuity of treatment staff. It is the responsibility of the provider-agency to ensure that newly assigned workers are ready to provide child specific treatment by arranging for a period of Child Specific Orientation. The Clinical Supervisor is responsible for overseeing Child Specific Orientation.

Reimbursement for Group Intervention must first be accessed through a child's primary health insurance, if available. If it is available but services are denied, please inform the CEDARR Family Center or DHS in order for Group Intervention to be authorized and reimbursed.

The Clinical Supervisor or Treatment Consultant, as identified in the Treatment Plan, provides the orientation with the assistance of an experienced home-based worker, if necessary.

The provider-agency may utilize this service if it becomes necessary. Clear documentation of the service must be maintained.

DHS previously allowed 4 units for Child Specific Orientation. Effective May 1 of 2005, DHS increased Child Specific Orientation for newly assigned staff to a maximum of 20 units within an authorized period of care. Each orientation session may include more than one staff member assigned to a specific child. This service does not require prior authorization by DHS.

5.4.6 Treatment Consultation

Treatment Consultation can take place prior to the implementation of Specialized Treatment (known as Pre-Treatment Consultation); it can occur during the course of HBTS intervention (known as Treatment Consultation); and it can follow a period of intensive HBTS as a child or adolescent is approaching discharge or transfer to another service (known as Post-Treatment Consultation). Clinicians eligible to provide Treatment Consultation are expected to be licensed health care professionals unless otherwise allowed by variance policy.

The intent of Treatment Consultation is to provide expert recommendations to the treatment team by a licensed health care professional with recognized expertise in the specific area of the child's needs.

Treatment Consultation has multiple purposes depending on the type of consultation being provided, namely:

5.4.6.1 Pre-Treatment Consultation

This service provides parents and families with specific therapeutic interventions to reduce a child's challenging behaviors and improves a child's functional skills. Activities can take place in the child's home or provider-agency. As a result of this support, families are expected to gain meaningful information, learn strategies for managing behavior, and - in some instances – may not require the implementation of direct intensive home-based treatment. For other families, it is expected that Pre-Treatment Consultation will also help in the development of a more comprehensive and informed HBTS Treatment Plan based upon having had Pre-Treatment Consultation. OT, PT, and SLP clinicians can also provide pre-Treatment Consultation.

5.4.6.2 Treatment Consultation

This service takes place during a period of intensive home-based treatment. It is designed to address Treatment Plan implementation, response of the child to the therapy, approaches to understanding and addressing emerging issues in the course of treatment, and adjustment to treatment mode. Treatment Consultation is intended to bring specific expertise and direction to the therapeutic regimen employed in the Treatment Plan. This service can be episodic to address

particular issues or concerns, or ongoing through a child's Treatment Plan. The Treatment Consultant may not simultaneously provide Clinical Supervision to a case. However, a Treatment Consultant may be used to provide supervision and oversight to non-licensed clinical staff.

5.4.6.3 Specialty Consultations

This aspect of Treatment Consultation allows for inclusion of expertise provided by OT, PT, and SLP licensed clinicians who can bring further guidance and direction to a child's HBTS plan. The mechanism for this is Occupational Therapy Support, Physical Therapy Support, and Speech and Language Therapy Support. Guidelines and documentation requirements for HBTS Specialty Consultations are contained in Appendix 19.

5.4.6.4 Post-Treatment Consultation

This is a time-limited service to allow for support as a child and family ends HBTS and/or is preparing to transition to other services. The treatment consultant reviews with parents specific therapeutic interventions to maintain gains and progress made during HBTS. Mental health and specialty consultants can provide post-treatment consultation. It is expected that post-treatment consultation be provided over a period of 6 months. If additional time is required, a request and justification should be made to DHS or the CEDARR Family Center.

5.4.7 Treatment Coordination

This service is provided by a team member on behalf of a specific child receiving HBTS services to ensure coordination with parents, other providers, or agencies directly involved with the provision of services to the child. Service activities can include gathering child specific information to facilitate HBTS, direct communication with other service providers regarding the child, and communications with parents, the "medical home", and/or primary care provider.

The concept of the "medical home," as put forth by the American Academy of Pediatrics (November, 1992) recognizes that the medical care of infants, children and adolescents should be directed and, when appropriate, delivered by well-trained physicians who are able to manage and facilitate all aspects of pediatric care. It is therefore imperative that an HBTS provider-agency facilitates and maintains timely communication with a child's pediatric primary health care provider during a course of HBTS intervention.

Treatment Coordination provides an array of administrative supports. Treatment Coordination is not considered to constitute a clinical service regardless of staff providing these functions.

5.5 Treatment Intensity of HBTS and Therapeutic Approaches

Treatment intensity refers to the number of direct service hours in an approved Treatment Plan. Treatment intensity requires continuous monitoring to ensure its appropriateness during an authorized period of care. If a child's referral for HBTS was the result of a CEDARR Center

referral, the provider-agency will have been given a recommendation regarding the number of hours considered to be medically necessary to address a plan of care. It is the provider-agency's responsibility to meet that level of treatment intensity necessary to promote the achievement of treatment objectives. The provider-agency also has the prerogative to seek a modification in treatment intensity from the referring CEDARR Family Center whenever circumstances warrant. For children and families whose HBTS Treatment Plan remains with DHS, the provider-agency is required to justify a level of treatment intensity that is medically necessary to address achievement of treatment objectives. In all cases, DHS expects recommendations for treatment intensity to be based on the individual needs of a child.

It is unacceptable to make treatment recommendations in order to facilitate staffing or to satisfy provider-agency practices. In relation to this, any pattern of a provider-agency seeking a certain number of hours across cases is unacceptable and inconsistent with Medicaid rules. Care is to be individualized and based on collaboration with the child's family and all relevant parties involved in developing a plan of care for the child. Ultimately, it is the provider-agency's responsibility to justify that its number of treatment hours are medically necessary.

Treatment intensity is individually determined and based on a comprehensive assessment of a child. Relevant background information shall include previous evaluations, IEP, IFSP, treatment summaries, recommendations from current providers of care, and collaboration with the child's family. The scope of interventions including treatment intensity should be appropriate to the child's needs and learning style. Many factors influence decision making when considering treatment intensity. Arriving at an appropriate level of treatment intensity must take into account the following factors:

- (a) The child's age
- (b) Ability to engage in sustained treatment (e.g., span of attention, stamina, developmental level, etc.) and expectations for progress
- (c) Type, nature, and course of presenting condition and diagnosis
- (d) Severity of presenting behaviors
- (e) Other treatment or educational services being received
- (f) Impact on family functioning
- (g) Presence of co-existing conditions
- (h) Presence of biological or neurological abnormalities
- (i) Current functional capacities of the child;
- (j) Family factors (e.g., parenting skills, living environment, and psycho-social problems)
- (k) Interaction with other agencies or providers
- (l) Appropriateness for Treatment Support

Overall, there is no single treatment approach indicated for the population of children eligible to receive HBTS. In relation to this, treatment effectiveness is multi-determined and unique to each child and family. It may be assumed that treatment intensity is equivalent to "maximizing progress" for a child. The issue of treatment effectiveness, however, is more complicated than the sheer number of hours that are provided. The most successful treatment models combine structured lessons, informal instruction, practice, and reinforcement with respect to skill development.

The responsibility for approving a treatment proposal rests with DHS or a CEDARR Family Center. The clinical review process takes into account all of the above information. When a treatment proposal is approved, the HBTS Treatment Plan provides the basis for defining treatment interventions, addressing treatment goals and objectives, and services to authorize payment to the provider-agency.

For additional information about treatment approaches refer to Appendix 8.

5.6 Duration and Continuation of Service

For a Medicaid eligible child under 21 years of age, there is no limit to the number of HBTS Treatment Plans that may be approved for an individual child.

Treatment is provided on a weekly basis for an approved number of hours for a period of six months. Treatment Plans may be modified based on formal action by the CEDARR Family Center or DHS following consultation with the provider-agency and approval from the family.

5.6.1 Categories of Treatment Requests

5.6.1.1 New HBTS Treatment Plans

A new Treatment Plan is defined as *either* a Treatment Plan written for any child when there is no history of HBTS, or it is the first HBTS Treatment Plan written for a child by a particular HBTS provider-agency.

In the instance of a child and family terminating services from one provider-agency in order to obtain care from another provider-agency, the receiving provider-agency must submit its own Treatment Plan. It is insufficient to copy and submit the previous Treatment Plan.

5.6.1.2 Reauthorization of Treatment Plan (Renewals)

A reauthorization request is for HBTS to continue for an additional six-month period of time with the current provider-agency. This proposal includes the results of treatment gains on particular goals and objectives during the past six-months, and new or revised treatment recommendations.

5.7 Family Involvement and Responsibility

HBTS is a home and community based service. These standards identify a series of requirements for certified provider-agencies with regard to family-centered care, communication and coordination with the family. Parents have the right to refuse a home-based worker from treating their child at any time during the course of treatment.

Provider-agencies should reasonably expect that families will recognize and respect the roles and responsibilities of providers. In order for HBTS to be effectively and safely provided, the family must ensure that the care setting is safe, and that family members will work positively with the provider-agency in maintaining a collaborative care relationship. The HBTS provider-agency must guide and assist their staff in the delivery of comprehensive, coordinated, family- centered care. It has the responsibility for creating a climate that is responsive to the child and family's needs as well as being supportive to HBTS personnel. Provider-agencies must recognize that HBTS staff is in the home to provide services in accordance with an authorized Treatment Plan. Under no circumstances is it permissible for home-based staff to engage in unauthorized tasks for families receiving HBTS (e.g. cleaning, cooking, running errands, or providing child care for siblings).

A commonly raised issue in home-based services is whether the parent or an adult family member must be present in the home when services are being provided. First, each provider-agency has the responsibility to determine with the family, the amount of time a parent is required to be physically present while services are being provided in the home. This is an issue of liability for both the provider-agency and family. DHS expects the level of parental presence to support the accomplishment of treatment objectives. In most cases this may require parents to be present in the home during HBTS a majority of the time (Appendix 9). A parent or "caretaker" must be in the home if siblings requiring supervision are present.

A related issue of a parent designee representing the family during HBTS activities may be allowable but is up to the discretion of the provider-agency. This will be determined on a case-by-case basis.

Second, regarding parental participation, an important objective of HBTS is for parents to be able to safely and satisfactorily address their child's behaviors and special needs. . Expectations regarding family presence and parental participation must be included in the HBTS Treatment Plan developed for any treatment period. Goals and objectives are required to be developed and included in a Treatment Plan that accomplishes this expectation.

Upon initiation of services, families must be provided with a copy of any policies regarding suspension or discontinuation of HBTS. The provider-agency has the responsibility to review and discuss these policies from outset of its involvement with a family.

If services are provided in the home and the family does not provide an appropriate environment for care, HBTS may be suspended until a review of the Treatment Plan can be scheduled with the notification to the CEDARR Family Center or DHS personnel. Examples of such

circumstances include presence in the home of dangerous weapons, illegal drugs, excessive use of alcohol, domestic violence, verbal abuse of HBTS workers, or significant failure of the family to participate with the providers in the Treatment Plan.

Issues that affect the safety of child as well as the home-based worker are of paramount importance. Interruption of HBTS to a child and family is a very serious concern to DHS. At times, circumstances warrant that such a step be taken. At any time, a provider-agency may seek consultation from DHS or a CEDARR Family Center if it is concerned about a developing situation. Appendix 10 provides further direction and requirements of provider-agency responsibilities for handling such circumstances.

5.8 Transportation and Travel Time Reimbursement

DHS has continued its policy to allow for transportation of a child for HBTS as long as this activity is essential to Treatment Goals. DHS has also made a change in policy to include travel time reimbursement for staff. Additional information is included in Appendix 22.

5.8.1 Transportation and 2:1 Coverage

The home-based worker may provide transportation during the course of HBTS. However, the State will not assume any liability or responsibility for these activities. Any transportation provided to an outside program, facility or activity must be related to a Treatment Goal, which has been approved by a clinical reviewer (i.e., CEDARR or DHS) and approved by the child's family.

DHS will not approve 2:1 coverage during transportation except under extremely unusual and time-limited situations subject to prior approval from DHS or CEDARR.

Specific requirements for the Transportation policy are outlined in Section 7.7.1. Provider-agencies are required to inform families of this policy, and obtain the necessary documentation including parent/guardian signatures prior to providing any transportation.

5.8.2 Travel Time Reimbursement

Effective May 1 of 2005, DHS reimbursed travel time for the purposes of rendering Clinical Supervision, Treatment Consultation, HBTS Specialized Treatment and Treatment Support to a child and family based on zoned mileage. The intent of this policy change is to facilitate the access and utilization of HBTS for children and families in outlying areas of the Rhode Island. DHS expects provider-agencies to expand their service areas. DHS will be monitoring the implementation of travel time reimbursement in order to measure the impact on service delivery.

Travel time reimbursement for HBTS Clinicians and HBTS Workers (Specialized Treatment and Treatment Support Staff) is for travel from the provider-agency to a child's home when the family resides in outlying areas. Travel Reimbursement Policy is contained in Appendix 22.

5.9 Management of Current HBTS Referral Lists

When DHS issued HBTS Certification Standards in February of 2003, it required every HBTS provider agency to submit a copy of its up to date, accurate referral list, on or before October 15, 2002. Families on referral lists submitted to DHS by October 15, 2002 have been eligible to seek new treatment plans directly from their HBTS provider-agency.

The provider-agency had the responsibility to ensure that families on a referral list were informed of their status and the availability of services on a quarterly basis. This included updating a family's commitment to remain on the referral list and reviewing the urgency of need for HBTS.

After October 15, 2002, HBTS provider-agencies were directed to no longer place families on a referral list directly without a CEDARR Family Center referral. Instead, families seeking HBTS have been directed to CEDARR Family Centers for support and assistance.

In summary, CEDARR Family Centers have had the responsibility to provide Prior Approval for all new Treatment Plans commencing January 1, 2003. DHS clinical reviewers are still available for consultation.

Individuals designated for referral lists include the following situations:

- 1. Waiting for HBTS
- 2. Receiving for HBTS while waiting for treatment from an ABA provider-agency
- 3. Receiving ABA intervention but will age out of a program but continue to need HBTS
- 4. A family with a grievance with an existing provider-agency such that a CEDARR Family Center has become involved and facilitated a re-referral to another provider-agency

5.10 Compliance with Other DHS Procedural Directives¹¹ for HBTS

- 1. It is expected that parents are involved in the development of all Treatment Plans with the provider-agency well in advance of the date the Treatment Plan is to be submitted. Provider-agencies shall allow parents sufficient time to review the Treatment Plan (minimum of five (5) calendar days) prior to submitting it to DHS or to a CEDARR Family Center. During this time, parents should have the opportunity to meet with the treatment team to discuss any questions or concerns they may have regarding their child's treatment
- 2. For Treatment Plans requesting reauthorization of services the following information should be included:
 - a) HBTS history

¹¹ Prior to the development of these standards, DHS provided written memoranda to existing HBTS provider-agencies clarifying its policies. All applicants are required to be aware and informed about these policies as summarized in this section.

- I. Dates of service for all previous HBTS plans
- II. Number of direct service hours approved for entire authorized treatment period. (Direct service hours include specialized treatment and treatment support services)
- b) For the most recent plan include the percentage of direct service hours delivered compared to the total number approved. For example, if 10 hours a week were authorized for 6 months, the total would be 260 hours (10 hours times 26 weeks). Sum all direct service of treatment hours delivered for the 6-month period and divide that by 260. To get a percentage multiply this number by 100
- 3. For each plan, include a schedule of the days and times services will be delivered
- 4. For billing issues DHS policies require the following:
 - a) If no direct service treatment hours were provided to a child during an authorized month, providers may not bill for Treatment Consultation or Clinical Supervision without written approval from DHS or a CEDARR Family Center
 - b) It is recognized that the delivery of treatment hours can vary in a given month due to extenuating circumstances. However, it is the responsibility of the provider-agency to ensure that Clinical Supervision, Treatment Consultation and Treatment Coordination continue based upon the level of treatment hours delivered that month
 - I. If fewer than 75% of direct treatment hours were provided in a given month, no more than an equal percentage of Clinical Supervision and/or Treatment Consultation hours can be billed for reimbursement
 - II. If 75% or more of the direct treatment hours were provided for a given month, all approved numbers of clinical supervision and/or treatment consultation may be billed
- 5. DHS has received requests for 2:1 staffing for some of the time HBTS workers work with a child. DHS does not fund these requests except under extremely limited conditions and for only a short-term period (i.e. up to one month). If a child requires a 2:1 staff ratio to safely participate in community/out-of-home activities, then either a family member or another adult appointed by the parents is responsible to provide this assistance. If this is not available, in order to assure safety, the child will need to refrain from community-based activities until his/her behavior will allow safe participation with only 1:1 staffing. The Clinical Supervisor, Consultant or Director of the HBTS program should make this determination for the Agency
- 6. HBTS is frequently requested for children for whom professional mental health services are clearly needed but are not being provided for a variety of reasons. For these children,

HBTS can serve as a useful and beneficial adjunct to therapy, but is not to be used as a substitute for professional counseling. The HBTS provider-agency should assist families, as needed, to identify a suitable therapist, and assist with scheduling these services and therapy appointments. This assistance and support should be documented in the Treatment Plan and daily progress notes kept by staff. DHS or CEDARR clinical reviewers may decline to approve plans when therapy is clearly needed and not being provided

- 7. Parents are required to participate in the development of and to approve all initial and renewal Treatment Plans. It is recommended that provider-agencies create a separate page that includes a statement indicating that the parent has reviewed the Treatment Plan, the hours requested and that they agree with the content of the plan. Provider-agencies are to have the parent sign and date this page and include it in the Treatment Plan
- 8. If a provider-agency wishes to change the services within the approved treatment period, an addendum to DHS and/or the CEDARR Family Center indicating the reasons for the change, new goals and objectives, a revised budget sheet and a signed parent signature form must be submitted. Requests will only be processed after all documentation has been received
- 9. Both initial and renewal Treatment Plans are to be submitted a minimum of 30 days prior to the date- anticipated services are to begin
- 10. Two sets of each Treatment Plan one original and one copy of the full proposal must be included. Documents in the following order are to be submitted:
 - I. New Data Sheet
 - II. Prescription from medical doctor (MD) for non-CEDARR referrals for new Treatment Plans
 - III. Parents' signature approving plan
 - IV. Prior Authorization and Non-PA Form
 - V. Treatment Plan including goals and objectives
 - VI. Accompanying information (e.g., IEP, assessments, letters)
- 11. Any additional information requested by DHS or the CEDARR Family Center clinical reviewers must be responded to in writing within <u>9 calendar days</u> upon receiving the request to avoid delaying authorization of services. Responses to reviewer's comments are to be provided using a separate sheet of paper for each response. Two copies should be included. Late responses may result in loss of funding for the period affected by the delay
- 12. General Medicaid policy states that claims must be filed with EDS within 12 months of the date of services

5.11 HBTS Service Performance Standards

5.11.1 Timeliness of Service Provision

In order to adequately meet the needs of children and families, HBTS must be provided in reliable and timely manner given the following requirements:

- 1. A CEDARR referral or family initiated contact must result in the provider-agency establishing an intake appointment. This face-to face meeting must occur within three weeks of the initial contact
- 2. The purpose of the intake appointment(s) is to review the CEDARR Family Care Plan with the family, introduce the family to the provider-agency's treatment orientation and approach, gather information regarding targeted behaviors, and arrange for observations at home, school and/or community settings. This is an initial meeting to establish a therapeutic working relationship with the child's family. As such, it is an opportunity for parents to ask questions regarding the provider-agency's HBTS program, parental involvement, and treatment expectations. Parents are to be provided with written information regarding HBTS and related policies such as client rights, transportation and grievance procedures
- 3. The provider-agency must develop and submit the Treatment Plan for review and authorization by the CEDARR Family Center within four (4) weeks of the initial intake appointment. This Treatment Plan is focused on goals and objectives, as recommended by the CEDARR Initial Family Assessment or Family Care Plan
- 4. When developing a Treatment Plan for a CEDARR Family Center or DHS, the provider-agency must fully staff a proposed level of treatment intensity within four weeks of the authorization date to start HBTS. The Treatment Plan must reflect the ability of the provider-agency to staff a child's plan

5.11.2 Parent Satisfaction

Routine and consistent parent feedback is an important aspect of quality assurance. HBTS provider-agencies shall design survey instruments to generate information for activities related to parent satisfaction with provider-agency services, accessibility, availability, and overall level of satisfaction. Section 6.5.5 provides further elaboration and direction. Parent/guardian information is strictly confidential.

5.11.3 Provision of Authorized Services

Certified HBTS provider agencies are expected to provide the service hours included in an approved treatment plan. However, the State recognizes that, for various reasons, including those related to staff capacity and availability of the child and family to engage in services, HBTS provider-agencies may not be able to successfully provide services for all authorized hours during a period or care.

Fully certified providers will be in compliance with the Certification Standards and meet performance standards. The performance standard for this Certification Standards is that an HBTS provider agency must provide at least 75% of authorized direct service hours to at least 90% of children in their caseload. Providing at least 75% of authorized direct service hours to less than 90% of children in their caseload in any ninety (90) day period may result in the provider agency receiving a Provisional Certification status.

6.0 CERTIFICATION STANDARDS

6.1 Requirements for Organization of Delivery of Service

An applicant for certification must demonstrate that it brings to the clinical program a sound combination of clinically proven treatment approaches, clinical management, skills and experience, and the capability to reliably provide HBTS.

Section 6 and 7 identify the requirements that must be addressed in a certification application. Applicants are to describe their approach to meeting these requirements. Further guidance as to how to complete the application is included in Attachment A, Application Guide.

6.2 Agreement to Accept Appropriate Referrals

Based on their clinical expertise and experience, certified HBTS provider-agencies will be expected to accept all appropriate referrals of Medicaid enrolled children who are determined to be eligible for HBTS, and to provide services on a timely basis as defined in Section 5.11.1 of these Certification Standards. Provider-agencies are allowed to decline to submit a Treatment Plan when they determine that their agency cannot meet the child's needs or HBTS is not an appropriate program for the child. Reasons might include lack of staff availability, lack of experience with a particular treatment condition, or geographic limitations. Documentation of the reason for declining to provide a treatment plan shall be maintained.

All provider-agencies seeking re-certification are expected to include Treatment Support, Pre and Post-Treatment Consultation and Group Intervention as part of their array of services.

6.2.1 Provision of Authorized Services

Certified HBTS provider-agencies are expected to consistently provide the service hours included in an approved Treatment Plan. However, the State recognizes that for various reasons, including those related to staff capacity and availability of the child and family to engage in services, HBTS provider-agencies may not be able to successfully provide services for all authorized hours during a period. The performance standard for a certified provider-agency is that an HBTS provider-agency must provide at least 75% of authorized direct service hours to at least 90% of children in their caseload for a period of three (3) consecutive calendar months.

6.3 Family Centeredness, Client Rights and Ethical Standards of Practice

6.3.1 Family Centeredness

HBTS provider-agencies must incorporate key components of family-centered care into their philosophy, service program and operations. Applicants must demonstrate the manner in which important principles of family-centered care are part of their approach to services. Areas of program policy shall include, but are not limited to, the following:

- 1) Established arrangements for ongoing communication with, and participation of the family in all aspects of the treatment program
- 2) Policy setting forth the degree and character of family involvement in care planning
- 3) Standards for working with family and the caregivers to help them safely maintain the child at home
- 4) Policies setting forth emphasis on family-centered service outcomes
- 5) Description of service arrangements flexible enough to meet special and individual needs
- 6) Demonstration of approaches to assuring families are encouraged to voice concerns and provide input

6.3.2 Client Rights and Family Service

The provider-agency shall have an established approach to ensure that client rights are clearly stated and communicated. Practices shall include maintaining written policies and procedures, as well as materials provided to families at the onset of care and periodically. Written materials shall also be provided to families identifying the circumstances under which a Treatment Plan will be discontinued.

Families must be informed of client rights, the expectation for their participation in treatment plan development, treatment modifications, and problem-resolution processes prior to the establishment and implementation of a Treatment Plan. The HBTS provider-agency shall have an established approach to ensure that this communication is maintained throughout the course of care. In this regard, the provider-agency shall have established policies, procedures and related records to ensure focus on customer service, solicitation of family input, documentation of and response to complaints, and prompt complaint resolution. This means being able to address complaints from parents or recipients of HBTS, as well as staff working for the agency.

The provider-agency must have written policies to facilitate an orderly transition of care, and/or follow-up or referral for services.

Provider-agencies shall also have written protocols as to how changes in service hours occasioned by changes in staffing will be communicated to families, including what accommodations will be offered in the event that service hours will be reduced. In the event that a provider-agency has elected not to accept *new* referrals that involve Treatment Plans that include Treatment Support, the agency shall also have written protocols that address circumstances where families in their current caseload elect Treatment Support.

Written materials shall be provided to families identifying the circumstances under which a Treatment Plan will be discontinued by the provider-agency. The provider agency must communicate verbally and in writing to the family the reasons for termination and provide this

information in writing to the CEDARR Family Center or DHS (See: Section 4.2.3). Appendix 10 provides additional information and requirements pertaining to provider-agency responsibilities for suspension or discontinuation of HBTS.

6.3.2.1 Termination of Care – No Safety Concerns

Provider-agencies must demonstrate compliance with the following DHS requirements when termination of services for non-safety concerns takes place:

- 1. The provider-agency must set forth its policies and procedures in writing regarding termination of services for non-safety concerns
- 2. Written notification shall be sent to the child's family or guardian, as well as DHS or CEDARR Family Center 30 days prior to discontinuing HBTS.
- 3. Reasons for discontinuing treatment must be stated.
- 4. Alternative resources and /or referrals, if appropriate, must be given

6.3.2.2 Termination of Care – Safety Concerns

The development of serious safety concerns can mean the immediate suspension or termination of HBTS care. The provider-agency shall have in place procedures for dealing with risks and safety to the well-being of the home-based staff including:

- 1. The provider-agency must conform to all aspects of mandated reporting of suspected child abuse
- 2. The provider-agency must seek emergency evaluation of the child when indicated
- 3. The provider-agency may need to seek intervention from the local police department if a situation warrants such action
- 4. The provider-agency must provide immediate notification to the family, DHS and the CEDARR Family Center when care becomes suspended or terminated (refer to Appendix 10 for additional information). A record of written documentation must be maintained that describe safety concerns and directives to staff and family that result in suspension or termination of care

6.3.2.3 Termination – Parent Initiated

A parent or guardian has the right to terminate HBTS at any time. It is expected, however, that the provider-agency will make every effort to satisfactorily acknowledge any reasons that may contribute to a parent or guardian's request to end care. It is also expected that the provider-

agency will assist the parent or guardian by referring to a CEDARR Family Center or other resource for assistance, if necessary.

6.3.3 Ethical Standards

Clearly articulated Principles of Ethical Care and Professional Conduct must be publicly posted. Protocols will identify standards of ethical practice for staff. The latter shall include, but will not be not limited to, the following issues:

- 1) Crisis intervention and management of emergency situations (i.e., family or staff)
- 2) Client and professional boundaries
- 3) Grievance policies and procedures
- 4) Use of aversive behavior modification techniques (including use of restraints)
- 5) Written description of services provided

6.3.3.1 Conflicts of Interest

Of major concern to DHS is the issue of individual(s) within a provider-agency having a multiple relationship such that a conflict of interest could occur thereby impairing objectivity, competence, or effectiveness while rendering HBTS to a child and family. This may occur when an individual is engaged in providing administrative or agency guidance (i.e., Member of Board of Directors or Executive Director) while simultaneously providing clinical services to HBTS (i.e., is the clinical supervisor or treatment consultant). It is the responsibility of the provider-agency to refrain from engaging in dual relationships.

Related to conflict of issue concerns, it is expected that the HBTS Director will use professional discretion when hiring clinical staff or HBTS workers if an applicant is a personal friend or relative. There is the potential for conflicts of interest to arise that could interfere with the exercise of professional discretion and impartial judgment. It is therefore incumbent to demonstrate that professional roles and expectations will be maintained and that recourse is available to address complaints in a fair and ethical manner. The family is to be informed when there is the potential for conflict.

For overall guidance regarding ethical principles, one is directed to the **National Association for Social Workers** (Code of Ethics) or the **American Psychological Association** (Ethical Principles of Psychologists and Code of Conduct):

- National Association for Social Workers http://www.naswdc.org/
- American Association of Psychologists http://www.apa.org/

6.3.4 Mandated Reporting of Child Abuse and Neglect

Be advised of DCYF policy, namely, "ALL persons in Rhode Island are required by law (RIGL 40-11-3) to report known or suspected cases of child abuse and/or neglect to the Department of Children, Youth, and Families within 24 hours of becoming aware of such abuse/neglect - 54e."

Call the DCYF Hotline at 1-800-RI-CHILD (1-800-742-4453) to report child abuse and/or neglect.

6.3.5 Use of Physical Restraint Interventions

The words "physical restraint" are sometimes used to address three different types of restraint. One is the use of mechanical devices (mechanical restraint) to restrict a person's movement. The other is chemical restraint with medication to control a person's behavior. A third type, and one that has received considerable attention, involves one or more individuals holding or physically manipulating another person in order to restrain that person's movement (physical restraint).

The use of physical restraint interventions has its origins in psychiatric hospitals and residential treatment facilities providing services to individuals under 21. These procedures have also been used in juvenile correctional facilities, acute medical care centers (emergency rooms) and, now, in public school settings. Professionals that utilize physical restraints claim that they are necessary to safely manage dangerous behavior in order to prevent imminent injury or harm to individual or others (including staff). Child advocates have argued that these procedures are too severe and cause too many children to suffer injury and even death. With the exception of needing to provide emergency medical care and related procedures to distressed children and youth in life threatening situations, there is minimal research supporting the use of physical restraint in other settings. There does not appear to be significant empirical support for its use with children and youth who are mentally retarded, developmentally disabled, or mental health concerns. There also does not appear to be literature reviewing the use of physical restraint for individuals receiving services like HBTS.

The use of mechanical restraints (i.e., any material including arm splints or equipment attached to a person's body that he or she can not easily remove that restricts freedom of movement) is a behavioral restraint used to prevent injury from severe self-injurious behavior. The use of a protective device (e.g., helmet) when ordered by a physician is <u>not</u> considered a mechanical restraint.

The use of chemical restraint, that is, the administration of medication for the purpose of restraint is not appropriate for home-based situations. A medication taken voluntarily in a non-emergency situation, as needed, ordered by a physician or other medical professional licensed to prescribe medication is <u>not</u> a chemical restraint.

As a result of moral and ethical concerns as well as legal interventions, the use of physical restraint with children and youth has resulted in the issuance of Federal and State regulations to prevent their maltreatment. Relevant policies are:

- a) The Child Health Act, Public Law No. 106-310 released by the Health Care Financing Administration (HFCA now called CMS) stipulates that it is unacceptable to apply restraint or seclusion in any form as a means of "coercion, discipline, convenience, or retaliation"
- b) State of Rhode Island Department of Children, Youth and Families Child Care Regulations: Addendum A Regulations Regarding the Use of Crisis Intervention, Restraint, and Seclusion Within Covered Residential Facilities; October 1, 2001
- c) The Rhode Island Board of Regents for Elementary and Secondary Education issued physical restraint regulations in September 2002 pursuant to R.I.G.L. 16-60-4. It stipulates that "school staff may use physical restraint only (1) when non-physical interventions would be ineffective and the student's behavior poses a threat of imminent, serious harm to self and/or others; or (2) pursuant to a student's IEP or other written plan developed in accordance with State and Federal law and approved by the school and parent or guardian." Staff training and incident reporting requirements are also required and explained in this document

In most medical, psychiatric and law enforcement settings today, strict guidelines also govern the use of physical restraint and conform to accreditation requirements such as the Joint Commission on Accreditation of Healthcare Organizations, the National Association of Psychiatric Treatment Centers for Children, and the American Academy of Pediatrics – Committee on Pediatric Emergency Medicine.

The position of DHS is that physical and mechanical restraint is the last resort. Restraint represents a "specialized treatment procedure" when an emergency situation constitutes an immediate and serious safety (i.e., dangerous) threat to the child and/or staff person. Inappropriate use of physical or mechanical restraint, however, can have grave consequences with a strong possibility of injury to the child or youth or staff member conducting the restraint, and possibly others in the environment. Therefore, the use of physical or mechanical restraint should require extreme caution and follow strict guidelines:

- 1. Staff shall be thoroughly trained in physical or mechanical restraint techniques (e.g., basket hold, floor, and prone restraints or application of mechanical devices)
- 2. Staff shall be thoroughly knowledgeable about physical and mechanical restraint guidelines and emergency situations that may dictate the use of this intervention.
- 3. Staff shall be thoroughly trained in conflict de-escalation and problem solving interventions (e.g., Nonviolent Crisis Intervention; Therapeutic Options; Professional Assault Response Training) such as:

- a) Nonviolent Crisis Intervention: <u>www.crisisprevention.com</u>
- b) Therapeutic Options: <u>www.therops.com</u>
- c) Professional Assault Response Training: (916) 723-3802
- 4. Staff shall also receive certification in First Aid and cardio pulmonary resuscitation in the event of an emergency related to restraint
- 5. Physical or mechanical restraint should never be performed as a means of punishment or to force compliance
- 6. There are no apparent guidelines regarding the duration of physical restraint with respect to outpatient settings or home-based situations. CMS guidelines for the duration of physical restraint within a residential or inpatient facility do not appear to be appropriate for home-based treatment (e.g., 4 hours for youth ages 18-21; 2 hours for 9 17 year-olds; and 1 hour for children under age 9). The following guidance is given:
 - a) Physical restraint shall be limited in duration and discontinued as soon as possible, and not exceed the duration of the emergency safety situation
 - b) In the event that a physical restraint lasts longer than 20 minutes and/or is utilized more than once per day within a three day period suggests that there is a high risk of frequent and dangerous behavior. Such occurrences necessitate: (1) immediate referral for emergency psychiatric evaluation (including notice to current providers of care), and (2) re-evaluation of the appropriateness of ongoing HBTS
 - c) The licensed clinical professional providing clinical oversight or engaged as the Clinical Supervisor shall have full responsibility for emergency safety management. The necessity of physical restraint for the child or youth must be continuously monitored. Each episode of physical restraint must be documented in writing and reviewed for appropriateness and use of alternative interventions
 - d) Mechanical restraint shall require the clear instruction regarding the type, use, situation and duration of this type of intervention based on written orders of the physician or prescribing health care professional
- 7. No restraint should be administered in such a manner that prevents the child or youth from breathing or speaking (respiration and skin color should be continuously monitored)
- 8. The application of physical restraint must involve 2 adults (i.e., HBTS treatment worker and parent)

- 9. The use of emergency safety management procedures requires parent notification and consent as part of the HBTS Treatment plan
- 10. Provider agencies may elect to decline referrals when there is a history of prior physical restraint and a likelihood of ongoing physical restraint

6.4 Coordination and Communication with CEDARR Family Centers

CEDARR Family Centers provide information and support services to families of children with special health care needs (CSHCN). Linking families to appropriate resources (e.g., clinical specialists or services) and providing care coordination are central aspects of the CEDARR system of care.

From the outset, the CEDARR Family Center works with the child and family to assess current circumstances, continuing needs, and reasonable next steps. Upon completion of an Initial Family Assessment (IFA) and clinical specialty evaluations, if indicated, a CEDARR Family Care Plan (FCP) is developed. Continuing Family Care Coordination services may be included in the plan, as appropriate. The plan may include CEDARR Direct Services and Supports. HBTS may be one of these Direct Services, if the family concurs. In this case, the CEDARR Family Center will provide the family with information about certified HBTS provider-agencies. The family and/or the CEDARR Family Center will schedule an intake appointment with the provider-agency it chooses. The CEDARR Family Center will help coordinate arrangements for all Direct Services and Supports. It is anticipated that in the majority of cases, the CEDARR Family Center will continue to work with the family to support efforts to gain access to needed services, track receipt of services, and monitor progress in meeting stated goals and outcomes. CEDARR Family Centers are responsible with the family to determine the types of service, intervention needed, the intensity of services, the required outcome of services, the appropriate approach to service intervention, and to recommend to DHS the authorization of service type, intensity, duration, and obtain prior authorization.

The HBTS provider-agency has the obligation to maintain communication with families and CEDARR Family Centers. Based on these communications, information is provided to the family about any projected waiting period that could delay the start of services. With this information, the family is better able to exercise informed choice regarding its preferred provider. The HBTS provider-agency must comply and adhere to communication and coordination requirements with CEDARR Family Centers.

A provider-agency seeking to offer HBTS must describe its processes to ensure coordination and communication with all CEDARR Family Centers. An applicant must demonstrate that it can work with CEDARR Family Centers and that it understands the role of the CEDARR Family Center in working with families. The HBTS provider-agency must be cognizant of processes and be responsive to a CEDARR Family Center in the following areas:

1. Accept referrals and relevant information to facilitate communication for the delivery of services

- 2. The basis of writing goals and objectives comes initially from the CEDARR Family Care Plan
- 3. Provide Treatment Plans to CFC
- 4. Provide feedback regarding progress toward goals
- 5. Inform CFC in writing of changes in the child's needs or ability of HBTS provider-agency to meet direct services hours, goals and objectives identified in the CFC Family Care Plan and the HBTS Treatment Plan

The provider-agency must demonstrate in its application that it has a process of communication with the CFC to ensure that this coordination will be in place. Refer to Appendix 2 for details regarding the CEDARR Authorization Process for HBTS.

6.4.1 Initial Referral to a CEDARR Family Center

Initial referral to a CEDARR Family Center of potential candidates for HBTS may occur in one of several ways:

- 1. Family referral
- 2. Referral from a medical provider (e.g., primary care provider, other medical specialist, or mental health clinician, etc.)
- 3. Referral from a Medicaid Managed Care Health Plan
- 4. Community referral (e.g., community mental health center, school, EI, CASSP, DCYF)
- 5. Hospital or residential level of care referral
- 6. Referral from an HBTS provider

All referrals for HBTS will be evaluated by a CEDARR Family Center. With the consent of the family, the CEDARR Family Center will engage the family in evaluating the needs of a child, which could include Home-Based Therapeutic Services. The HBTS provider-agency must work directly with all CEDARR Family Centers regarding prior authorization of new services, renewal of services, communication of treatment progress, and fully meet DHS expectations regarding timeliness. Applicants must describe their understanding of these arrangements, describe how they will interact with CEDARR Family Centers, and report on their initial contacts with CEDARR Family Centers (Appendix 16).

6.4.2 CEDARR Family Center Initial Family Assessment (IFA) and Basic Services

The goal of the IFA is to develop a working profile of the family that forms the foundation for the assessment. The assessment includes: an assessment of urgency, a developmental and diagnostic history (including physical health, behavioral health and cognitive development); an analysis of current interactions with the care system (e.g., RIte Care, pediatrician, specialist, hospital, or other provider); involvement with other programs (e.g., Early Intervention,

Medicaid, RIte Care, or school programs); family strengths, needs and supports; knowledge about or linkage with advocacy groups or professional associations; current insurance status and needs; and potential eligibility for various public programs or community supports.

6.4.3 CEDARR Family Center Care Plan

On the basis of the IFA, the Family Care Plan (FCP) will be developed in concert with the family. The Family Care Plan may result in any combination of referrals and/or services, which could include:

- Home-Based Therapeutic Services
- Pediatric Home Care
- Early Intervention Services
- Special Education Services
- Therapeutic Services in Child and Youth Care
- Personnel Assistance Services and Supports (when available)

Where Home-Based Therapeutic Services are selected and the family has chosen a preferred provider-agency, the CEDARR Family Center will make a written referral to a certified HBTS provider-agency. The referral will include:

- 1. Initial determination of need, and scope of treatment (i.e., Pre-treatment Consultation, Specialized Treatment, Treatment Support and Group Intervention)
- 2. Projected number of hours per week of HBTS
- 3. Expected duration for the service (six months is the maximum period for a Family Care Plan before a plan review, revision and reauthorization is required)

6.4.4 CEDARR Family Center Certified HBTS Treatment Plan

The referral from the CEDARR Family Center provides the initial information noted above. The CEDARR Family Center will provide materials from its own assessment to avoid duplication of effort by the HBTS provider-agency and recommendations for intensity of treatment hours. Next, the HBTS provider-agency will perform a more focused assessment directed toward determining the specifics of the Treatment Plan and the appropriate arrangements for actual provision of services. This Treatment Plan will detail treatment goals and objectives, schedule of service, approach to working with the family and other relevant considerations. In turn, the Treatment Plan is submitted to the CEDARR Family Center for clinical review.

The CEDARR Family Center has the authority to approve, reject or request modification to the provider-agency's proposed Treatment Plan. Once approved by the CFC, its recommendation for authorization is then transmitted to DHS and to EDS, and it is included in the finalized Family Care Plan.

6.4.5 CEDARR – HBTS Provider-Agency Dispute Resolution Process

The CEDARR Family Center and the provider-agency shall have established procedures to identify and resolve differences, and demonstrate how families will be informed with respect to the following occurrences:

6.4.5.1 HBTS Provider- Agency and CEDARR Family Center Disagreement Process

In the event of disagreement regarding elements of the Treatment Plan (e.g., treatment interventions and methods, focus of treatment, involvement of parents or collaterals, or treatment intensity), it is anticipated that the parties can reach resolution in most cases through joint review and discussion. Where resolution cannot be achieved, a request can be made to the Department of Human Services for an independent clinical review. After this has been done, the prior authorization necessary for claims to pay will be entered by the CEDARR Family Center or DHS.

6.4.5.2 DHS Fair Hearing Process

If a child's parents or guardian objects to the decision of the CEDARR Family Center or DHS they can request a hearing. An Administrative Fair Hearing allows for testimony to be presented from all concerned parties. In turn, the Hearing Officer renders a written decision. Upon completion of this process, the prior authorization necessary for claims to pay may be adjusted based on the hearing decision.

It is expected that the provider-agency develop a Treatment Plan consistent with CEDARR objectives or if absent a CEDARR will provide a Treatment Plan that is medically necessary and appropriate in terms of clinical intensity. Both the CEDARR and the provider-agency have the responsibility to develop a collaborative set of recommendations. In either case, the decision on how many hours per week of Specialized Treatment and Treatment Support to provide must involve the child's parents as part of developing a plan of care for a child receiving HBTS. When this is done, the process of clinical review is more readily facilitated and, in turn, may offset the need for a Fair Hearing.

Rules and procedures for requesting a Fair Hearing are as follows with related information provided in Appendix 11:

- 1. The recipient's parents or guardian will receive written notification of the approved Treatment Plan following the second independent clinical review
- 2. If the parents or guardian disagree with the results of the clinical review, they have ten (10) days from the date of authorization to file a request for a Hearing
- 3. If a request for a Fair Hearing is received by DHS within ten days, there will be no modification to the number of requested direct service hours until the conclusion of the hearing. During this time, the provider-agency may provide services and submit

claims for payment of services, as the proposed Treatment Plan in dispute remains in effect

- 4. If a request is received after ten days, the approved number of direct service hours will stand until the conclusion of the Fair Hearing. Claims will be paid in accordance with prior authorization
- 5. Fair Hearing decisions may be appealed with the Superior Court within 30 days of the date of the hearing decision pursuant to Rhode Island General Laws 42-35-1 et seq

6.5 Strength Of Program Approach: Process of Care and Management of Clinical Services

The applicant must demonstrate that it brings a sound combination of experience and skills to be certified as an HBTS provider-agency. The provider will use written standards of care to describe the process by which treatment is planned, delivered, monitored and evaluated. There will be evidence of initial and on-going involvement of family members. The applicant will ensure that staff has appropriate competencies, educational preparation, and clinical experience to engage in the delivery of HBTS.

In describing its program, the applicant will address the areas identified below:

6.5.1 Process of Care

The provider will demonstrate that the care process is systematically organized and grounded in sound clinical principles. In section 6.5.1, the applicant will describe its clinical model and policies and procedures for the provision of HBTS. In doing so, the applicant should address each of subsections 6.5.1.1 through 6.5.1.5.

6.5.1.1 Treatment Approach and Clinical Guidelines

Clinical guidelines will permit diversity and flexibility while promoting the best possible outcome for each child. The clinical guidelines must address screening and intake, assessment and treatment planning, treatment plan implementation, and treatment plan monitoring and modification. Written standards of care, policies and procedures will be in place for all levels and aspects of HBTS.

The applicant must describe the treatment approach used and the range of conditions for which the treatment approach is considered to be efficacious using clinically proven interventions. DHS reserves the right to have evidence-based information presented to support proposed clinical interventions. The applicant should clearly identify the guiding principles that govern the treatment program and their basis in empirical literature and/or nationally accepted practice standards. Sound clinical and program management are required.

The applicant must provide evidence of satisfactory written and professionally recognized clinical practice guidelines along with identification of how adherence to such guidelines is systematically monitored.

6.5.1.2 Screening and Intake for HBTS

Applicants must have an organized process for handling referrals; for screening and intake; and for determining the appropriateness of the services of this agency for a child and family. Screening and intake must be based on written policies and procedures that clearly define admission criteria and program services. These policies must ensure that contact with a family respects the family's privacy, and is conducted in a culturally sensitive and family-centered manner. A documented written record of the intake is to be maintained.

Applications for certification must include written policies and procedures for addressing the following:

- 1. Managing referrals
- 2. Screening and intake
- 3. Eligibility and admission criteria
- 4. Management of direct services
- 5. Management of current referral list and communication with families
- 6. Assisting families not eligible for HBTS by providing alternative recommendations

6.5.1.3 Assessment and Treatment Planning

There shall be a thorough identification of the specific problem(s) to be addressed. Components of the assessment shall be identified (e.g., communication with the CEDARR Family Center, review of IFA and FCP, parent interview, child observation, conversations with school representatives, collaboration with other health care providers, and review of past evaluations).

For non-CEDARR Family Center referrals, the IEP, 504 Plan or Individual Family Service Plan (IFSP) and any other evaluations or treatment materials should be attached and the assessment should identify the services being received through the schools or Early Intervention programs. Problem behaviors should be identified with the order of importance or priority indicated.

The involvement of the family and the preparedness of the family (including extended family and/or other potential caretakers) to participate in treatment shall be indicated. Unless exceptional circumstances are identified, this will include both parents even if they are not living together.

All requested documentation shall be obtained solely by parental written consent and all records shall be maintained to ensure their security and confidentiality.

As part of the assessment, licensed clinical staff should complete the Global Assessment of Functioning Scale (GAF) taken from the Diagnostic and Statistical Manual IV (DSM-IV). The Global Assessment Scale for Children (Appendix 12) may be used in its place.

6.5.1.3.1 Diagnosis and Treatment History

The child's diagnosis (DSM IV or ICD-9) must be clearly identified. Documentation shall identify who made the diagnosis, the basis for the diagnosis, when the diagnosis was made, and its current status. Treatment information is to be updated for any period of authorized care. There is no need to repeat treatment history, family history or other treatments if this information is included in the child's CEDARR Initial Family Assessment and/or Family Care Plan.

The approved Treatment Plan shall include information on the services that have been provided previously. Information should be present regarding any other providers that have been involved with the child and family (e.g. child psychiatrist), other treatments that have been tried or considered, and the sequence of events leading to the submission of the HBTS request. This should also include information regarding other Medicaid funded services such as Pediatric Home Care, other CEDARR Direct Services, and all CEDARR Family Center Services, when appropriate.

6.5.1.3.2 Treatment Plan Development

The applicant will describe the provider agency's specific protocols for the development of the Treatment Plan. Protocols will identify the provider-agency's overall approach and address each of the following:

- 1. The identification and prioritization of treatment goals and objectives shall be clearly based on the CEDARR IFA and FCP in addition to the provider-agency's assessment. Goals and objectives for each identified problem must be realistic, specific and measurable. There must be clear written criteria that define the anticipated level of achievement for each goal and objective.
- 2. A brief description of treatment interventions for each identified problem shall be included
- 3. The provider-agency will adhere to appropriateness criteria for admission to HBTS. It is critical to identify in detail if and why a child is at risk for inpatient hospitalization or out-of-home placement, as well as how the proposed HBTS services will address each identified problem and risk factor
- 4. The parental level of participation in the Treatment Plan must be identified. The plan may include a parent consultation component (i.e., behavior modification techniques, conflict resolution, information on child development) and a support component (e.g., referral for additional support or treatment for the child and family)

- 5. The Treatment Plan shall meet the criteria for Family-Centered Care (See: Section 1.3.6)
- 6. The parent/guardian/legal custodian must sign the proposed Treatment Plan
- 7. The number of weekly treatment hours shall be clearly explained and justified in detail
- 8. The Treatment Plan shall indicate the anticipated length of treatment and the method for measuring progress towards obtaining the stated goals including discharge criteria

A Treatment Plan for HBTS must present goals that are specific to the child's diagnosis and presenting problems. The following must be done:

- 1. All treatment interventions, including those addressing OT, PT and SLP goals and objectives, must be identified and described in detail. This means defining treatment methodologies (e.g., applied behavioral analysis, contingent time-out, response cost, schedules of reinforcement, social stories, modeling, etc.) linked to clearly defined goals
- 2. Goals must be written in behavioral terms with well-defined objectives. Goals are broad, generalized statements about what is to be learned within the 6 month period of treatment
- 3. Objectives are specific, measurable, short-term, observable behaviors. Objectives are the foundation upon which lessons and assessments are built to meet the overall treatment goals. The objectives should include:

• Audience: Who is this aimed at?

Behavior: What is expected that the child will be able to do?
Condition: How? Under what circumstances will learning occur?
Degree: How much? Must set the specific criteria to be met.

6.5.1.4 Treatment Plan Implementation

The applicant shall describe how it will provide effective, efficient, high quality treatment on a timely basis. The applicant must ensure that a child's assessment and Treatment Plan is completed in a timely manner consistent with certification expectations, namely:

- 1. After receiving a written referral or a request for HBTS, the provider-agency must make telephone contact with families within seven calendar days to schedule an intake appointment
- 2. If appropriate for HBTS, an intake appointment must take place within <u>three weeks</u> of the telephone contact
- 3. Treatment Plans must be developed and provided for review and authorization either to a CEDARR Family Center or DHS within 4 weeks of a completed intake appointment

4. When developing a Treatment Plan for a CEDARR Family Center or DHS, the provider-agency must be able to fully staff a proposed level of treatment intensity within four weeks of commencing treatment. The Treatment Plan must reflect the ability of the provider-agency to staff a child's plan

6.5.1.5 Treatment Plan Modification and Renewals

The applicant must describe its procedures for Treatment Plan monitoring and modification of treatment throughout a course of care. Resources (i.e., staff and staff responsibilities) and processes (e.g., clinical supervision, treatment consultation and treatment coordination) must be identified to ensure that data is collected, analyzed, and used to inform further treatment during an approved course of HBTS. It is necessary to demonstrate how data is used during clinical supervisory sessions and parent consultations to inform the delivery of care. It must also be evident that data is appropriately maintained and reviewed for determining future HBTS needs. It is recognized that achieving treatment objectives will vary for many reasons. However, when treatment progress falls significantly below expectations for the provider or family, or there is evidence of regression during a course of HBTS, each factor associated with an unsatisfactory outcome must be specifically addressed. Changes and modifications to treatment that result from this must also be described in detail. The provider must demonstrate that this takes place throughout a course of care.

When seeking re-authorization of HBTS, it is insufficient to simply list incomplete treatment outcomes or regression as justification for HBTS. The HBTS provider must demonstrate how HBTS services can maximize the achievement of goals and improved functioning for the child and, when appropriate, the family. This means reviewing methods of intervention and ensuring that best practices are followed. The latter must also address the use of other professional services (e.g., individual or family therapy, consultation with a child psychiatrist) as well as a reexamination of treatment objectives and treatment intensity when formulating requests for services.

The HBTS provider-agency must agree to provide a summary of the child's response to HBTS to the child's primary care provider and other interested parties related to the child's Treatment Plan upon written request from the parent or guardian.

6.5.2 Management of Clinical Services

The applicant must demonstrate a sound organizational approach to ensuring the provision of effective, timely and high quality services. Certified providers will demonstrate that the care process is systemically organized and grounded in sound clinical principles. Refer to section 7.4 for additional requirements. In this section, the applicant will describe its clinical and administrative organization and management model for Home-Based Therapeutic Services.

It is incumbent upon an applicant choosing to offer Home-Based Therapeutic Services to provide agency credentials, as represented by DOH, DCYF and/or MHRH licensure, and identify any

agency accreditation provided by national governing bodies (e.g., CARF, CASFC, COA, JCAHO, ORS, etc.). Therefore, the applicant must attach copies of appropriate licensure(s) and agency accreditation(s). Appendix 13 provides a list of common accreditation organizations. Management of clinical services must address the following areas:

6.5.2.1 Clinical Roles and Scope of Practice

The work of the HBTS team must be systematically organized with clear delineation of the staff roles, reporting relationships and supervision. Detailed job descriptions must be provided for each member of the team. Protocols will include clear delineation of the role and scope of practice of each position within the HBTS treatment team in such areas as:

- 1. Supervision and scope of practice
- 2. The ways in which clinical supervision is effected, ratio of supervisor time to treatment team staff time
- 3. Staff evaluation
- 4. Treatment Plan design, monitoring, evaluation and Treatment Plan modification.
- 5. Coordination with family
- 6. Coordination with CEDARR Family Center and communication with other service providers as appropriate

The organizational chart must identify the specific individuals who fill identified positions and list credentials. Personnel must meet all applicable State licensure and certification requirements. Position job descriptions must be provided and address such areas as:

- 1. Reporting relationship
- 2. Functional tasks and performance expectations
- 3. Required skills, training, and experience
- 4. Licensure qualifications

6.5.2.1.1 Agency Orientation and Training

All staff shall be provided with a general orientation to the provider-agency with respect to its mission, policies and procedures, administrative structure, training, and other relevant information. The provider-agency must have policies and programs for orientation, training of new staff, and for continuing education and professional development that fully meet these Certification Standards. Staff is required to participate in these activities, as specified by the individual's position and job description.

Personnel files shall contain documentation of the training programs staff has completed. Provider-agencies must have a written program for in-service training and orientation of all new employees. Provider-agencies shall annually determine staff training needs and develop a written plan and schedule of staff training.

6.5.2.1.2 Preparation of Home-Based Staff

The provider-agency shall delineate the requirements used to ensure that all clinical staff and home-based workers are fully qualified to implement all aspects of a Treatment Plan before engaging in the delivery of care to a family. As a condition of employment and on a case-by-case basis, home-based staff shall have basic knowledge and skills. The provider-agency shall demonstrate its required basic training for all home based workers. It is recommended that basic training for all home-based workers shall include, but not be limited to the following:

Clinical Training

Child Development
Behavior Modification
Family Dynamics
Crisis Intervention
Substance Abuse
Psychiatric/Medical Disorders
Developmental Disabilities

Medical Training

Emergency Measures
Medications
Appropriate Restraint Techniques
Universal Precautions
DOH Regulations regarding nonlicensed administration of
medications

Provider-agencies shall have an ongoing training that all staff must attend. Some training topics under HBTS Specifics (see below) shall occur prior to a home-based Specialized Treatment worker and home-based Treatment Support worker delivering services to a child.

HBTS Specifics

Client rights
Ethics and confidentiality
Reporting procedures and documentation requirements
Overview of the clinical record and HBTS Treatment Plan

6.5.2.2 Clinical Supervision

Clinical Supervision of home-based staff must occur throughout a period of authorized treatment. Policies and procedures must be in place to ensure the reliability and availability of supervision by qualified personnel. This means:

- 1. HBTS staff must have appropriate credentials and meet qualifying standards to provide Clinical Supervision
- 2. Written policies and procedures that demonstrate a clear supervisory structure that guides the delivery and implementation of treatment, supervision of home-based workers (i.e., specialized treatment and treatment support), and management of clinical services, including assessment of progress and modifications to Treatment Plan
- 3. Defining the ratio of supervisor's time to treatment team staff, showing the ways in which clinical supervision is provided

4. Protocols identifying team meetings, team participants, and the process for periodic assessment and treatment plan revisions as appropriate

6.5.2.3 Applied Behavior Analytic (ABA) Services

The use of ABA methods for individuals with Autism and their families has received considerable multidisciplinary study and is regarded as an evidence-based practice. The treatment of children with Autism and related disorders with ABA discrete trial interventions is, however, highly involved and requires specific competencies. While this type of intervention is not the only treatment approach for children with Autism, it is being used by some provideragencies, and is often requested by referring agencies or other health care providers.

For the purposes of certification, DHS regards ABA discrete trial interventions as being highly specialized and distinct from the application of basic behavior therapy principles. Provideragencies vary according to the degree to which ABA is employed, as well as the level of sophistication and training of staff providing clinical supervision to home-based specialized treatment staff.

DHS recognizes that ABA programs for HBTS may request additional hours to support planning, directing and supervising effective ABA interventions (e.g., lead therapy) when Treatment Plan requests exceed DHS maximum limits. These supports are approved for only ABA recognized providers.

For provider-agencies who would like to become recognized as having an ABA program, DHS must obtain a list of staff and their qualifications to develop and administer a program of intervention that emphasizes discrete trial learning. The clinical supervisor or "Behavior Analyst" must meet the following criteria:

- 1. Have a masters or doctorate in behavior analysis, or in psychology, special education, or another human service discipline with an emphasis in behavior analysis
- 2. Have extensive coursework in principles of learning, principles of behavior, or basic behavior analysis; experimental analysis of behavior, behavior assessment or methods of direct observation of behavior; applied behavior analysis; single-subject research designs; and legal and ethical issues
- 3. Have received supervised practicum, internship, fellowship, or employment experiences in ABA
- 4. Be a Board Certified Behavior Analyst (BCBS) or Board Certified Associate Behavior Analyst (BSABA)

To further avoid any confusion to families and provider-agencies, DHS recognizes the term "Behavior Analyst" as applying only to those individuals who can demonstrate satisfactory compliance with the above-mentioned requirements. Provider-agencies are directed to

Appendix 14 for additional information regarding the formal credentialing of professional behavior analysts.

6.5.2.3.1 Lead Therapy

When HBTS Certification Standards were issued in February 2003, three provider-agencies met criteria to provide intensive behavior analytic treatment (i.e., ABA discrete trial intervention). CCFH subsequently accepted Lead Therapy as an Administrative Service with specific functions offered by the Lead Therapist, which these provider-agencies then referenced in their HBTS Treatment Plans. Other provider-agencies may seek to include Lead Therapy if approved by DHS. The Lead Therapist reports to the Clinical Supervisor. The Lead Therapist functions and responsibilities are to:

- 1. Maintain a professional working relationship with families including how to apply instructional strategies
- 2. Attend all clinical treatment meetings for an assigned child and assist in the development of an HBTS plan. Maintain formal supervision sessions with the clinical supervisor
- 3. Develop instructional strategies including new materials and prepare new drills under the direction of the clinical supervisor; oversee the use of teaching materials, and update materials
- 4. Directly observe the assigned child and treatment workers once a month. Provide guidance and summarize observations
- 5. Manage treatment data: develop, distribute, collect and summarize data sheets, and provide this information to the clinical supervisor on a daily basis
- 6. Consistently update treatment staff about interventions and coordinate meetings
- 7. Provide emergency coverage as an HBTS treatment worker
- 8. Demonstrate the ability to set consistent and supportive therapeutic limits for HBTS

6.5.2.4 Staffing and Staff Qualifications

It is the responsibility of a provider-agency to conform to DHS requirements regarding staff credentials, training, personnel management, and practice guidelines. Staff requirements differ with respect to the role and functions associated with providing HBTS. The provider-agency shall demonstrate that it meets the specific staffing requirements for the home-based worker, Clinical Supervisor, Treatment Consultant, and Treatment Coordinator. The applicant must therefore give written assurances that these standards will be provided and maintained as a requirement for receiving and maintaining certification.

With respect to continuing education, the provider-agency shall have policies and procedures in place for all employees consistent with DHS certification and licensure requirements. This requires that:

- 1. Licensed health care professionals providing Clinical Supervision and Treatment Consultation conform to DOH continuing education requirements according to respective disciplines. Therefore, the applicant is directed to consult with Department of Health guidelines for licensure of individual health care professionals
- 2. For non-licensed clinical staff, and only for those considered under the variance from licensure condition, the following requirements apply:
 - a) 10 hours per year of continuing education are required from accredited programs with national or regional certifying authority
 - b) Agency training must be documented
 - c) Individual staff certification to provide Clinical Supervision and Treatment Consultation must be renewed on an annual basis. Therefore, the provider-agency must demonstrate that it can monitor and enforce this standard for employees who are subject to a *Variance from Professional Licensure*

6.5.2.4.1 Background Screening for New Staff

A Background Criminal Investigation (BCI) by local or state police and Child Abuse Neglect Tracking System (CANTS) by DCYF are required of all potential employees.

The provider-agency must demonstrate that it has policies in place to ensure that these screenings take place. In addition, the following requirements must be met unless otherwise approved by DHS for HBTS employees.

The provider-agency must also demonstrate that it has procedures in place to address issues of professional misconduct.

Agencies shall have policies and protocols that offer families clear information as to the qualifications of all staff assigned to work with their child, as well as their right to request substitutes.

6.5.2.4.1.1 <u>A Home-Based Specialized Treatment Worker Must</u>:

- 1. Be at least 19 years of age; have a high-school degree or equivalent, and two years of supervised experience working with children with special health care needs, or
- 2. Have an Associate's degree in human services (i.e., psychology, counseling, child development, education, nursing, etc), or

- 3. Be currently enrolled in not less than six (6) semester hours of relevant undergraduate coursework at an accredited college or university, or
- 4. Demonstrate competency to work with children with special health care needs (as outlined in 6.5.2.1.2) as evidenced by active participation in an agency-specific formal training program, approved by DHS, and successful completion of objective testing within twelve (12) months of hire, or
- 5. DHS will allow provider-agencies to employ as home-based workers for Specialized Treatment and Treatment Support individuals with at least 3 years of experience with adults with developmental disabilities to work as home-based workers. The provider-agency must provide this individual with training in child specific orientation, and
- 6. Have successfully passed a BCI and CANTs

6.5.2.4.1.2 A Home-Based Treatment Support Worker Must:

- 1. Be at least 19 years of age; have a high-school degree or equivalent, and one year of supervised experience working with children, or
- 2. DHS will allow provider-agencies to employ as home-based workers for Specialized Treatment and Treatment Support individuals with at least 3 years of experience with adults with developmental disabilities to work as home-based workers. The provider-agency must provide this individual with training in child specific orientation, or
- 3. Have an Associates degree in human services (i.e., psychology, counseling, child development, education, nursing, etc), or
- 4. Demonstrate competency to work with children with special health care needs as evidenced by provider-agency's skills validation requirement, and
- **5.** Have successfully passed a BCI and CANTs

6.5.2.4.1.3 A Clinical Supervisor Must:

Be a Rhode Island licensed health care professional eligible to provide Clinical Supervision. Licensure must be in one of the following: licensed clinical social worker, licensed independent clinical social worker, marriage and family therapist, mental health counselor, registered nurse with a Masters degree, or psychologist.

- 1. Have successfully passed a BCI and CANTs
- 2. Unlicensed individuals currently providing services must have DHS approval to render clinical supervision (See: Appendix 4)

6.5.2.4.1.4 A Treatment Consultant Must:

Be a Rhode Island licensed health care professional eligible to provide Treatment Consultation. This may include one of the following categories: licensed independent clinical social worker, marriage and family therapist, mental health counselor, registered nurse with a Master's degree, psychologist, Occupational Therapist, Physical Therapist or Speech and Language Pathologist.

- 1. Have successfully passed a BCI and CANTs
- 2. Unlicensed individuals currently providing services must have DHS approval to render Treatment Consultation (See: Appendix 4)
- 3. The hiring of OT, PT and SLP therapists employed by a Local Education Authority (LEA) may be subject to DOE and/or LEA regulations that could limit or prohibit their participation as Treatment Consultants for HBTS. Independently licensed professionals may not be prohibited from providing treatment consultation. It is the responsibility of the provider-agency to confirm if a conflict of interest exists

6.5.2.4.1.5 A Treatment Coordinator Must:

- 1. Possess a minimum of a Bachelor's degree
- 2. Have successfully passed BCI and CANTs

6.5.2.4.1.6 A Lead Therapist Must:

- 1. Be at least 19 years of age; have a high-school degree or equivalent, and two years of supervised experience working with children with special health care needs, or
- 2. Have an Associate's degree in human services (i.e., psychology, counseling, child development, education or nursing, etc.), or
- 3. Be currently enrolled in not less than six (6) semester hours of relevant undergraduate coursework at an accredited college or university, or
- 4. Demonstrate competency to work with children with special health care needs as evidenced by active participation in agency specific formal training with completion of objective testing within 12 months of hire, and
- 5. Have been employed as a Home Based Specialized Treatment Worker for 6 consecutive months in an ABA program; and
- 6. Has received 6 to 12 months of clinical supervision; and

7. Is deemed to be at a level where they are prepared to take on additional responsibilities by the ABA program director

6.6 Timeliness of Service, Other Access Standards

Certified providers will be in compliance with the certification standards and meet performance standards for the timeliness of services provided. Performing at a rate less than the timeliness certification standards outlined below may result in provisional certification status.

6.6.1 Timeliness Standards for NEW Referrals

The HBTS Provider Agency must meet the following timeliness performance standards.

6.6.1.1 Intake Appointment

The performance standard is an HBTS provider agency must conduct an intake appointment for at least 80% of those requesting an appointment within 3 weeks of treatment referral.

6.6.1.2 Treatment Plan Submission

The performance standard is that an HBTS provider agency must submit at least 80% of new treatment plans for those of whom home based therapeutic services are appropriate to CEDARR Family Centers for approval within 4 weeks of intake appointment.

6.6.1.3 Initiation of Direct Services

The performance standard is an HBTS provider agency must initiate direct services for at least 80% of new clients within 4 weeks of treatment plan authorization.

6.6.2 Timeliness Standards for RENEWING Cases

The performance standard is an HBTS provider-agency must submit for reauthorization all (100%) treatment plans requiring renewal at least 30 days prior to expiration.

6.6.3 Timeliness Standards for Treatment Plan Clinical Review Process

Provider agency cooperation is required throughout the CEDARR or DHS clinical review of the proposed treatment plan. Should a clinical reviewer require clarification or additional information, the HBTS provider-agency is required to respond in writing to the reviewer within 9 calendar days.

The performance standard is that additional information requested by DHS or CEDARR Family Centers in the process of treatment plan review and reauthorization will be provided within 9

calendar days for all (100%) treatment plans for which questions arise throughout the reauthorization process.

6.6.4 Hours of Service

HBTS must be available to families on a continual basis throughout a period of authorized treatment. The applicant shall define its administrative hours of operation, and direct service hours, which can include day, evening, and weekend coverage. Families must be informed of hours of operation for their child's Treatment Plan. Applications must include hours of operation, and procedure for accessing administrative or clinical staff.

6.6.4.1 Continuity of Care

It is the responsibility of the HBTS provider-agency to address in the application continuity of care to minimize disruptions in treatment (i.e., holidays, staff vacations, sick time, etc.). The provider-agency must demonstrate its process and procedures for maintaining continuity of care and inform parents of this responsibility. With respect to multiple home-based workers providing treatment, whether Specialized Treatment or Treatment Support, the provider-agency shall limit the number of workers assigned to a given case to ensure continuity and consistency in treatment.

6.6.5 Measures of Parent Satisfaction

Annual parent satisfaction surveys must be conducted. When multiple children within a family are receiving services, one survey is needed for each child receiving services within each family.

The format and content of the measurement tool is the responsibility of the provider agency. Areas of interest to DHS include, but are not limited to:

- 1. Sensitivity to family centeredness and cultural competencies
- 2. Availability of Clinical Supervisor, Treatment Consultant and/or Treatment Coordinator
- 3. Progress made during treatment (e.g., participation in community and quality of life outcomes)
- 4. Communication with family and others (e.g., CEDARR, medical professionals, school personnel)
- 5. Staff availability, promptness and actual delivery of authorized hours
- 6. Professionalism of staff and services (i.e., treatment coordination, clinical supervision, and treatment consultation, accounting of complaints, compliments, and grievances)

It is recommended that surveys include both quantitative and qualitative feedback from parents. Survey results will be analyzed, trended and reported to DHS. Further clarification regarding

content and/or reporting will be provided by DHS once the certification process has been completed.

6.7 Service Monitoring and Reporting

The HBTS Provider Agency must comply with the following service monitoring and reporting requirements. See Appendix 17 for additional information regarding reporting requirements.

6.7.1 Quarterly Reports

Provider agencies will be expected to report required data for January 1 – June 30th due July 1 each calendar quarter on the last business day of the month following the end of each calendar quarter The first Quarterly Report is due following the completion of the second calendar quarter following DHS certification of the provider agency. Provider Agencies are required to submit the following reports on a quarterly basis:

a. Report 1 – Provision of authorized direct service hours

The purpose of this report is to monitor the percentage of authorized direct service hours that are provided to families

b. Report 2 – Timeliness of direct service initiation

The purpose of this report is to monitor the percentage of families for which direct services are initiated within 3 weeks of treatment plan authorization

6.7.2 Annual Reports

Provider Agencies are also required to provide reports on an annual basis. Annual reports are to be submitted by July 31st annually.

Provider-agencies are required to submit the following reports and documents on an annual basis:

- 1. Licensure status report for unlicensed consulting/supervising staff
- 2. Documentation of trainings conducted and attended by direct service workers and clinical supervisors. Include continuing education units
- 3. New written documents provided to families regarding their rights and responsibilities and documents demonstrating family-centeredness since submission of last annual report
- 4. Summary of family satisfaction survey methods and results

- 5. Summary report on all grievances/complaints received and logs of timeliness of complaint resolution
- 6. Summary of all incident reports received during provision of HBTS (i.e., health and safety, allegations of abuse, accidents)

6.7.3 Additional Service Monitoring and Reporting

DHS may also request additional reports, documentation, and site visits, as necessary to monitor compliance with these Certification Standards and services provided by the provider-agency.

6.8 Record Keeping Requirements

The provider-agency must describe its policies and procedures for record keeping. Systematic recording of HBTS hours provided on a weekly basis with family verification is required. For the home-based worker, time sheets documenting the specific hours of service provided per shift must be co-signed daily by the family receiving services. Services billed shall correspond to the approved hours requested in a Treatment Plan and be supported by written documentation for each service provided.

Appendix 15 provides further detail for compliance with Medicaid regulations. The provider-agency must provide long-term storage of clinical records in accordance with Medicaid regulations. Additional record keeping requirements are described in Section 7.

6.9 Emergency Coverage

Whenever a home-based worker is working with a child during shifts, there shall be back-up staff (e.g., Clinical Supervisor or Treatment Consultant) immediately available to provide consultation and/or direction to staff and/or families should a crisis situation develop. This requires a response to a telephone call or page within 15 minutes and on-site assistance, as necessary, within 60 minutes of the initial call. An emergency or crisis is characterized by sudden onset, rapid deterioration of cognition, judgment, or behavior, is time limited in intensity and duration, and poses serious risk of harm to the individual or others.

HBTS provider-agencies shall describe its processes for developing an emergency plan. An emergency plan must be included with the HBTS Treatment Plan. This must be periodically reviewed/revised with a child's parents or guardian. During non-shift times, parents must be aware of specific steps to take in the event of a crisis (e.g., calling the police or seeking emergency evaluation for medical treatment). This plan must be developed in coordination with the CEDARR Family Center.

7.0 QUALIFIED ENTITY

A certified provider must be able to demonstrate that it complies with core State requirements as to organizational structure and process. These requirements pertain to areas such as incorporation, management of administrative and financial systems, human resource management, information management, quality assurance/performance improvement and others. State requirements in these areas are consistent with the types of expectations or standards, which would be set forth and surveyed by health care accrediting bodies, and which are generally held to be critical to effective, consistent, high quality organizational performance and care provision.

Applicants for certification are not required to systematically address in detail each of these areas in their certification applications. Rather, these are set forth as fundamental requirements for certified entities. In many areas applicants will be asked to provide assurances that their agency systematically addresses each of the standards identified. In certain areas, more specific description regarding the manner in which the agency meets the standard is required. The Application Guide provides guidance as to how the application should be structured and the areas, which need to be addressed.

In not requiring applicants to explicitly address the elements in Section 7, the State is seeking to simplify the effort needed to develop an application; these certification requirements remain in place. The State reserves the right to review certified entities for compliance with these certification requirements.

7.1 Incorporation and Accountable Entity

The applicant for certification as a Home Based Therapeutic Services provider-agency must be legally incorporated. The certified entity shall serve as the accountable entity responsible for meeting all of the terms and conditions for providing HBTS. Applicants must clearly present the overall structure by which services, requirements and programmatic goals will be met. The corporate structure of the entity must be clearly delineated.

7.1.1 Partnership or Collaboration

Satisfactory performance as a certified HBTS provider-agency calls for significant organizational capability. In some cases this capability may be present within a single organization and application for certification will be made based on the strengths of that single organization. In other cases the application may represent the joint effort of several parties, which have the combined capabilities to meet the certification requirements. This could come, for example, through a joint venture, a formal partnership or an integrated series of executed contractual arrangements. Regardless of form, a single legal entity will be certified with overall responsibility for performance. The certified HBTS provider-agency is to be the single billing agent for all HBTS.

7.2 Governance and Mission

The governance of the entity must be clearly delineated. Composition of the Board of Directors and any conditions for membership must be clear. The overall performance of an organization flows from the philosophy and oversight of the leadership. Leadership and stakeholders "build" the mission, vision and goals; this in turn shapes the business behavior and is reflected in the tone that leadership sets for the operation of the organization. The leadership strives to recruit members who reflect the cultures and ethnic backgrounds of clients, and to provide a mix of competencies that address organizational needs. Specific standards regarding governance and mission are as follows:

- 1. The agency has a clearly stated mission and publicly stated values and goals
- 2. The agency is operated/overseen by some type of legally or officially established governing body, with a set of governing documents or by laws. This governing body has full authority and responsibility for the operation of the organization
- 3. The governing body is self-perpetuating and has a recruitment and periodic replacement process for members to assure continuity and accountability
- 4. The governing body hires, supervises, and collaborates with a chief executive officer or director. Together the executive and governing bodies provide organizational leadership
- 5. The governing body has final accountability for all programs. Through a collaborative relationship with the executive and the management team, the governing body is responsible for developing the program goals and mission and ensuring compliance with legal and regulatory requirements

7.3 Well Integrated and Organized Management and Operating Structure

The HBTS provider-agency will be able to function in an efficient and effective manner, assuring consistency and quality in performance and responsiveness to the needs of families. The applicant shall provide clear identification of who is accountable for the performance of HBTS. This includes administration, clinical program quality, and management of service delivery and overall financial management.

7.3.1 Administration

Specific standards regarding administration are as follows:

1. The Executive, under supervision of the governing body, is responsible for financial management, achieving program outcomes, meeting client needs, and implementing the governing body's strategic goals

- 2. A current chart of organization, which clearly defines lines of authority within the organization, must be maintained and provided as part of the certification application.
- 3. The management of the organization is involved in the planning process for performance improvement and is involved in planning for priorities and setting goals and objectives for the written Quality Assurance/Performance Improvement plan
- 4. There is a written corporate compliance plan in place that is adopted by the governing body

7.3.2 Financial Systems

The organization must have strong fiscal management that makes it possible to provide the highest level of service to clients. Fiscal management is conducted in a way that supports the organization's mission, values, and goals and objectives in accordance with responsible business practices and regulatory requirements. Financial management requires a set of sophisticated financial planning and management capabilities if the organization is to remain viable. The organization must be able to obtain relevant data, process and report on it in meaningful ways, and analyze and draw meaningful conclusions from it. Managers must use financial data to design budgets that match the constraints of the organization's resources, and provide ongoing information to aid the governing body in managing and improving services. Therefore, the financial managers must have the ability to integrate data from all of the client and financial accounting systems (e.g., general ledger, billing and appointment scheduling). Data must also be utilized to make projections for planning and budgeting purposes.

Specific standards regarding financial systems are as follows:

- 1. Financial Management is provided by a Chief Financial Officer, Fiscal Director, or Manager with demonstrated experience and expertise in managing the finances of a human services organization with third party reimbursement. In larger organizations (e.g. with revenues in excess of \$1 million) this might be an MBA with demonstrated finance experience or a CPA; in smaller organizations a comptroller with a degree in accounting might be sufficient. This individual must possess expertise in financial and client/patient accounting, financial planning and management
- 2. The organization's financial practices are consistent with the most up to date accounting methods and comply with all regulatory requirements
- 3. The organization's financial planning process includes annual budgeting, revenue projections, regular utilization and revenue/expense reports, billing audits, annual financial audits by an independent CPA, and planning to ensure financial solvency
- 4. The organization has written policies and procedures that guide the financial management activities (including written policies for and procedures for expenditures, billing, cash control; general ledger, billing system; registration/intake system; payroll system; accounts payable; charge and encounter reporting system and accounting administration)

- 5. The organization has evidence of internal fiscal control activities, including, but not limited to cash-flow analysis, review of billing and coding activities
- 6. The system must track utilization of service units separately for each individual client and aggregate this information by payor, performing provider and diagnosis/problem
- 7. The organization has a billing office/function that bills for services rendered and collects fees for service and reimbursement
- 8. The organization assesses potential and actual risks, identifies exposures, and responds to these with preventive measures
- 9. The organization carries appropriate general liability insurance, and ensures that appropriate professional liability policies are maintained for program personnel
- 10. Where the organization contracts with outside entities and/or providers, policies and procedures mandate contract language to detail the entity's or provider's accountability to the Governing Body and its' By-laws
- 11. The organization has systems that facilitate timely and accurate billing of fee-for-service, capitated, and case-rated insurance plans, clients and other funding sources. Once bills are forwarded to payors, the system properly manages payments, follow-up billing, collection efforts and write-offs
- 12. The organization has a written credit and collections manual with policies and procedures that describes the rules governing client and third-party billing. Specifically, the organization has in place and adheres to policies and procedures ensuring compliance with Medicaid regulations pertaining to coordination of benefits and third party liability. Medicaid by statue and regulation is secondary payer to all other insurance coverage
- 13. Clinical, billing and reception/intake staff receives ongoing training and updates regarding new and changed billing and collection rules and regulations

7.4 Human Resources, Staffing

Human Resource activities within the organization are conducted to ensure that proper staffing for optimum service delivery to clients occurs through hiring, training, and oversight of staff activities. The activities are organized to serve the governing principles of the organization and compliance with these Certification Standards. The organization provides clear information to employees about job requirements and performance expectations, and supports continuing education, both internal and external, that is relevant to the job requirements of the individual. In addition, all staff receive training about major new organizational initiatives and about key issues that may affect the organization overall.

Specific standards regarding Human Resources and Staffing are as follows:

- 1. The organization's personnel practices contribute to the effective performance of staff by hiring sufficient and qualified individuals who are culturally and linguistically competent to perform clearly defined jobs
- 2. Employee personnel records are kept that contain a checklist tickler system to track appropriate training, credentialing and other activities. A copy of each employee's active license will be kept on file
- 3. The provider-agency must perform annual written performance appraisals of staff based on input from families and supervisors. These must be available in the personnel files for review by DHS upon request.
- 4. Policies and procedures contain staff requirements for cultural competency that are reflected in the job descriptions
- 5. Staff is hired that match the requirements set forth in both the appropriate job description and in the policies and procedures
- 6. Each employee's record contains a job title and description reflecting approved education, experience and other requirements, caseload expectations, supervisory and reporting relationships, and annual continuing education and training requirements. Supervisory job descriptions establish expectations for both contributing to the organization's goal attainment and for communicating the goals and values of the organization. All job descriptions include standards of expected performance
- 7. The organization provides a clear supervisory structure that includes plainly delineated areas of control and caseloads as appropriate. The roles of team members are defined with a clear scope of practice for each. Supervisors receive specialized training and coaching to develop their capacities to function as managers and experts in their clinical and/or technical fields. The organization holds supervisors accountable for communicating organizational goals, as well as for clinical and technical supervision. This includes:
 - a) Protocols for communication and coordination with all interested parties (e.g., special education, primary care physician, or other specialists)
 - b) Clear procedures for addressing unmet licensure requirements will be stated. Credentialing records will be maintained annually to document compliance
- 8. Credentials of staff established by the management team and approved by the Governing Body are contained in the job descriptions. An individual hired into a position has his or her credentials verified through primary source verification, as appropriate, and records maintained in the employee's record

- 9. A record of primary source verification is maintained in the individual employee record. This includes, at a minimum, verification of licensure, review of insurance coverage, liability claims history, verification of board certification for physicians, verification of education and training required by law, and professional references and performance evaluations about applicant's ability to perform requested duties. The individual employee record for behavioral health practitioners should also contain a signed statement from the practitioner that addresses if any Medicare or Medicaid sanctions have been imposed in the most recent three-year period
- 10. Staff has appropriate credentials and meets qualifying standards of the organization. These are updated and checked regularly
- 11. The organization provides training and training opportunities for all levels of staff
- 12. Staff is required to participate in training activities on an ongoing basis, as specified by the organization and position and job descriptions

7.5 Quality Assurance and Performance Improvement

The organization is required to have policies and procedures and demonstrable activities for quality review and improvement (e.g. formal Quality Assurance or Performance Improvement plan). The organization ensures that information is collected and used to improve the **overall** quality of service and performance of the program. The Quality Assurance/Performance Improvement (QA/PI) program that the organization develops strives to: improve the systems related to the delivery of service to the clients; include the preferences of clients in the provision of services; and measure the process and outcomes of the program services. The QA/PI program is an ongoing process of planning, monitoring, evaluating, and improving the system in order to improve the outcomes of service provided to clients.

Standards regarding Quality Assurance/Performance Improvement are as follows:

- 1. The organization has a Quality Assurance/Performance Improvement program that includes a written performance improvement plan with annual review of goals and objectives, data analysis, outcomes management, records review and operational/systems improvement. Written records are maintained for PI program activities
- 2. The PI program contains specific timetables for activities and measurable goals and objectives, which consider client concerns and input
- 3. Effective data analysis is conducted that includes an assessment of client or organizational needs, identification of service gaps, and integration of that data into organizational decision-making processes

7.6 Information Management and Record Keeping

The organization must use data to affect the performance, stability, and quality of the services it provides to clients in its governance, systems, and processes. Standards regarding information management and record keeping are as follows:

- 1. The organization obtains, manages, and uses information to enhance and improve its performance. Information it maintains is timely, accurate, and easily accessible, whether maintained in electronic or other format. Evidence exists that information gathered and maintained is used in decision-making for the organization
- 2. The organization maintains a written plan for information management which includes: client record-keeping policies and procedures; confidentiality policies and procedures; and record security policies and procedures. The plan provides for the timely and accurate collection of data and sets forth a reporting schedule
- 3. The organization shall ensure that its information management systems are protected from unauthorized outside access and shall meet all applicable HIPAA regulatory requirements when such standards are promulgated and effective
- 4. The information management plan specifies standard forms and types of data collected for client intake, admission, assessment, referral, services, and discharge
- 5. The information management plan has an incident reporting and client grievance-reporting component
- 6. Information management processes are planned and designed to meet the organization's internal and external reporting and tracking needs, and are appropriate to its size and complexity. Mechanisms exist to share and disseminate information both internally and externally
 - a) The organization maintains signed releases for sharing of clinical information
 - b) Where necessary, signed affiliation agreements exist
 - c) Reports are available on an appropriate schedule (weekly, bi-weekly, monthly, quarterly, etc.) for use by service providers, case managers, supervisors, managers, CEO, and the Governing Body for assessing client and organizational progress
 - d) Reports to authorities (state, federal, and other funding and regulatory entities) for review are submitted accurately, in the required formats and on a timely basis

- 7. The organization has written policies and procedures regarding confidentiality, security, and integrity of information, and has mechanisms to safeguard records and information against loss, destruction and unauthorized access or disclosure
 - a) The organization has policies and procedures in place to safeguard administrative records, clinical records, and electronic records
 - b) Electronic records are backed up, transmitted data is encrypted and secure, and access is password protected
- 8. Client information is accessible and is maintained in a consistent and timely manner, with enough information to support the consumer's needs or diagnosis, to justify services delivered, and to document a course of treatment and service outcomes
 - a) Every client will have a record that contains: an initial assessment, the detailed assessment of client assets and needs, client goals care/Treatment Plan, documentation of care/services provided, documentation of change in client's status, and where necessary, discharge summary
 - b) All records must include evidence of informed consent, where required
- 9. The client record documents treatments/interventions provided and results from the treatments/interventions. All entries into the client records are dated and authenticated, and follow established policies and procedures
 - a) Changes in client's condition or lack of change following service provision are recorded in the client record at the time of service provision and signed by the service provider
 - b) Achievement of a client objective or milestone toward an objective is noted in the client record. Achievement of an objective or milestone results in a revised assessment
 - c) Lack of progress in achieving a client objective or milestone toward objective results in a reassessment of the client
- 10. The client record will be the basis for billing. All service billings must be substantiated in the client record. Additional clarification regarding Medicaid and DHS requirements is included in Appendix 15

7.7 Health and Safety, Risk Management

The organization supports an environment that promotes optimal safety and reduces unnecessary risk for clients, family members and staff. The home-based nature of HBTS calls for

specific policies and procedures to assure that services are provided in a safe and effective manner for both the child and the staff.

Standards regarding Health, Safety, and Risk Management are as follows:

- 1. The organization's policies and procedures designate managers who monitor implementation of Health and Safety policies and report to the Quality Assurance Performance Improvement program committee and the Governing Body
- 2. The organization will have protocols for identification and monitoring of safety risks, family crises, medical emergencies and difficult situations
- 3. Health and safety policies and procedures are clearly communicated to agency staff, visitors, and clients
- 4. Programs will have an effective incident review process

7.8 Transportation of HBTS Clients

In the course of provision of services the provider-agency may want to provide transportation if clinically relevant. The State is approving only the service provision and accepts no liability or responsibility for transportation. Inclusion of transportation as part a Treatment Plan is only appropriate if it clearly relates to facilitating the accomplishment of defined and previously approved treatment objectives. Transportation can only relate to the child receiving HBTS and is not to be included in a treatment plan for solely convenience.

The provider-agency must demonstrate that it has procedures in place to protect the safety of child being transported. This means addressing certain minimum criteria for all staff and vehicles engaged in transportation:

- 1) Current and appropriate vehicle insurance that allows for transporting children
- 2) Current vehicle registration and valid State inspection
- 3) The driver's history must be free of accidents for the past year, with no history of DWI
- 4) Parents have signed a waiver for each driver releasing DHS of any liability and responsibility for anything that occurs as a result of transportation activities
- 5) DHS will not approve 2:1 coverage during transpiration except under extremely unusual situations subject to prior approval from DHS or CEDARR
- 6) Seat belts and./or child restraints must be utilized as required by State law

APPENDIX 1: DEFINITION OF MEDICAL NECESSITY

As defined and applied to all State Medicaid programs (See: RI DHS Medical Assistance Program, 300-40-3, September 1997), <u>Medical Necessity</u> refers to medical, surgical, or other services required for the prevention, diagnosis, cure or treatment of a health related condition. It includes services necessary to prevent a decremental change in either medical or mental health status. Services must be provided in the most cost effective, efficient and appropriate manner. Services are not to be provided solely for the convenience of the beneficiary or service provider.

The prescription or recommendation of a physician or other service provider of medical services is required for a determination of medical necessity to be made, but such prescription or recommendation does not mean that the Medical Assistance Program will determine the provider's recommendation to be medically necessary. The Medical Assistance Program is the final arbiter of determination of medical necessity (See RI DHS Medical Assistance Program, 300-40-4, September, 1997).

APPENDIX 2: CEDARR AUTHORIZATION PROCESS FOR HBTS

Families may seek HBTS through referral and authorization by a CEDARR Family Center. Services provided by a CEDARR Family Center are intended to benefit families by helping them obtain information and the appropriate services for their child. The CEDARR Family Center helps by arranging specialty clinical evaluation, if required, developing a Family Care Plan, identifying resources, and providing coordination of care.

The CEDARR Family Center makes recommendations, and parents are given information about professionals, resources, and different treatments. The following process takes place:

- 1. If HBTS is recommended, families are given several HBTS Direct Service Providers to choose from and referrals are sent out. At this time parents are informed about the availability of HBTS prior to referrals being made.
- 2. The HBTS provider-agency will review the referral and determine if the child is appropriate for services. The HBTS referral sheet includes a section for the provider-agency to indicate their acceptance or denial of the referral and the position on the provider-agency's Referral List. The referral sheet must be faxed back to the CFC with this information within 48 hours.
- 3. The HBTS Direct Service Provider then submits a Treatment Plan to the CEDARR Family Center for a clinical review and authorization.
- 4. The specifics are authorization are as follows:
 - a) HBTS provider-agencies must develop and submit written Treatment Plans to the CEDARR Family Center no later than four weeks prior to beginning treatment services. Retroactive requests and/or unauthorized periods of providing services are not allowed.
 - b) A Treatment Plan must include clearly defined treatment objectives with measurable outcomes. Treatment must conform to professionally recognized and established clinical practice guidelines. A CEDARR Family Center has the right to request published research regarding clinical efficacy. Hours of treatment and a weekly schedule must be stated. Individuals providing clinical supervision, treatment consultation, and treatment coordination must be identified along with their hours.
 - c) Each Treatment Plan receives a thorough clinical review. Treatment Plans are reviewed within thirty (30) days of their receipt at the CEDARR Family Center.
 - d) The clinical review process frequently involves written and verbal communication between provider-agencies and reviewers in order to facilitate a thorough understanding regarding a request for services. As a result, recommendations to

- modify an initial Treatment Plan may be made in terms of goals, treatment intensity, or indirect services. Changes in treatment intensity require parental consent.
- e) Based on clinical review, actions are taken by the CEDARR Family Center for authorization or denial of service. Written notification is sent directly to provideragencies by the CEDARR Family Center and to the parents.
- f) Provider-agency cooperation is sought to resolve treatment questions prior to issuing a denial of authorization. The provider-agency must then review proposed treatment changes with a child's parents/guardians and obtains their consent. Once a revised plan has been obtained, services may then be authorized with written notification to the provider-agency and family.
- g) If a Treatment Plan is unsatisfactory, the provider-agency has nine (9) days to respond to clinical reviewer's questions or concerns. Written responses are required of provider-agencies. Untimely or unsatisfactory responses may result in changes to a Treatment Plan including reductions in treatment intensity, duration of treatment, or denial of a Treatment Plan.
- h) If the concerns raised during a clinical review of a Treatment Plan are not successfully resolved, clinical reviewers at DHS will conduct a second independent review of the Treatment Plan. The reviewer's decision is sent to the CEDARR Family Center who then forwards this decision to the provider-agency.
- i) For families receiving HBTS, there is a right to appeal any denial of HBTS or modification of treatment intensity.
- j) If there are substantial changes in a child's level of functioning (e.g., inpatient hospitalization or regression) that could require service changes during an approved period of care, it is the responsibility of the provider-agency to inform the CEDARR Family Center and receive approval to amend the Treatment Plan. These requests are then subject to the same review process as for initial requests.
- k) Provider-agencies have the responsibility to submit requests for reauthorizations of care thirty (30) days -month prior to the expiration of an existing Treatment Plan for review.
- 5. The CEDARR Family Center is responsible to provide the following functions: clinical review of an HBTS Treatment Plan, authorization, oversight, and collaboration with the HBTS provider-agency.
- 6. The HBTS Direct Service Provider is responsible to the CEDARR Family Center for all utilization and authorization of care. In order to render HBTS as a direct service, the provider-agency must be certified as a CEDARR Direct Service Provider

APPENDIX 3: DHS AUTHORIZATION PROCESS FOR HBTS

For existing provider-agencies of HBTS, DHS has required the following steps in order to secure authorization and reimbursement for services. They are:

- 1. HBTS provider-agencies must develop and submit written Treatment Plans to DHS no later than thirty (30) days prior to beginning treatment services. Retroactive requests and/or unauthorized periods of providing services are not allowed.
- 2. A Treatment Plan must include clearly defined treatment objectives with measurable outcomes. Treatment must conform to professionally recognized psychotherapeutic approaches and established clinical practice guidelines. DHS has the right to request published research regarding clinical efficacy. Hours of treatment and a weekly schedule must be stated. Individuals providing clinical supervision, treatment consultation, and treatment coordination must be identified along with their hours.
- 3. Each Treatment Plan receives a thorough clinical review. Treatment Plans are reviewed within thirty (30) days of their receipt at DHS. Parents are notified of a plan's receipt.
- 4. The DHS clinical review process frequently involves written and verbal communication between provider-agencies and reviewers in order to facilitate a thorough understanding regarding a request for services. As a result, recommendations to modify an initial Treatment Plan may be made in terms of goals, treatment intensity, or indirect services. Changes in treatment intensity require parental consent.
- 5. Based on clinical review, actions are taken by DHS for authorization or denial of service. Written notification by DHS is sent directly to provider-agencies and parents.
- 6. Provider-agency cooperation is sought to resolve treatment questions prior to issuing a denial of authorization. The provider-agency must then review proposed treatment changes with a child's parents/guardians and obtains their consent. Once a revised plan has been obtained, services may then be authorized with written notification to the provider-agency and family.
- 7. If a Treatment Plan is unsatisfactory, the provider-agency has nine (9) days to respond to clinical reviewer's questions or concerns. Written responses are required of provider-agencies. Untimely or unsatisfactory responses may result in changes to a Treatment Plan including reductions in treatment intensity, duration of treatment, or denial of a Treatment Plan.
- 8. For families receiving HBTS, there is a right to appeal any denial of HBTS or modification of treatment intensity regarding reauthorization.

- 9. If there are substantial changes in a child's level of functioning (e.g., inpatient hospitalization or regression) that could require service changes during an approved period of care, it is the responsibility of the provider-agency to inform DHS and receive approval to amend the Treatment Plan. These requests are then subject to the same review process as for initial requests.
- 10. Provider-agencies have the responsibility to submit requests for reauthorizations of care thirty days prior to the expiration of an existing Treatment Plan for review.

APPENDIX 4: LICENSURE AND PRACTICE STANDARD

1.0 Core Requirements for HBTS Certification

1.1 Licensure

These Certification Standards require that individuals engaged in providing Clinical Supervision or Treatment Consultation for HBTS hold a currently valid license from the Rhode Island Department of Health (DOH). DOH and the Department of Mental Health, Retardation and Hospitals Regulations (MHRH) require that professionals be licensed for their respective specialties (i.e., mental health counselor, marriage and family therapist, nurse, psychiatrist, psychologist, or social worker).

- 1. DOH stipulates that licensure is required for health care professionals if:
 - a) You represent yourself in name, title, or abbreviation to the public as a psychologist, clinical social worker, marriage and family therapist, or mental health counselor; or if
 - b) You engage in providing diagnosis, assessment, treatment planning, and treatment to the public

2. Relevant DOH policies are:

- a) Clinical Social Worker: R5-39.1 CSW/ICSW
- b) Mental Health Counselor and Marriage and Family Therapist: R5-63.2 MHC/MFT
- c) Psychologist: RS-44-PSY
- d) Occupational Therapist: R5-40.1-OCC
- e) Physical Therapist: R5-40-PT/PTA
- f) Speech Pathologists and Audiologists: R5-48-SPA
- g) Physician: R5-37-MD/DO
- h) Nurse: R5-34-NUR/ED

3. Relevant MHRH policies are:

a) MHRH Section 800 Program Administration, Professional Qualifications and Supervision

1.2 Competency Requirements for Licensed Health Care Professionals

Licensure relates to broad areas of clinical practice and by itself does not ensure that providers have the specific and current competencies to work effectively with the special needs being addressed in HBTS. In addition to licensure, DHS requires that individuals engaged in providing Clinical Supervision or Treatment Consultation for HBTS demonstrate competency to work with

specific target populations. Specifically, evidence of the following is required unless otherwise noted:

1.2.1. Training: 2 years of supervision post degree while working with related

target population(s); and

1.2.2. Education: Ongoing continued professional education. Evidence of such

education could include, for example, certification as a "Behavior Analyst" granted by the Autism Special Interest

Group (SIG) of the Association for Behavior Analysis.

1.2.3 Continuing Education: Licensed clinicians must conform to the requirements of their

respective Boards for maintaining continuing education credits. Provider-agencies are responsible for oversight and management

of this requirement.

2.0 Special Provisions for Individuals Seeking Exception from Professional Licensure

In the 2003 HBTS Certification Standards, DHS recognized a set of unlicensed individuals providing Clinical Supervision and Treatment Consideration prior to the issuance of certification standards. DHS recognized that, in certain cases, unlicensed individuals might have relevant work and training to be considered under criteria for Variance from Professional Licensure under *Exception Status*:

2.0.1 Transition Clinician Group

At the time of initial certification, DHS granted *Exception Status* on a case-be-case basis to individual personnel engaged in rendering Clinical Supervision or Treatment Consultation if hired by provider-agencies prior to August 15, 2002. *Exception Status* was therefore limited to individuals making up the <u>Transition Clinician Group</u>. DHS has not changed the definition for Transition Clinician Group since issuing Certification Standards in 2003.

Individuals granted *Exception Status* must be supervised by a licensed clinician who shall approve and bear professional responsibility for all aspects of the Treatment Plan.

2.0.1.1 Demonstrated Competency Required for Exception Status

Determinations were established on a case-by-case basis related to the individual's level of education, acquired training, supervised experience, and work history. The individual must meet basic competency requirements, exception criteria, and continuing education requirements. The criteria are listed below:

1. Education: Minimum of a bachelor's degree in a related field (e.g., special education, child development, psychology, or counseling)

2. Training: Demonstration of in house training and professional development conferences attended over the last 3 years of current employment specific to target populations

3. Supervised practice: 2 years at a minimum of 1 hour per week provided by a licensed professional with experience serving target populations

4. Recommendations: 3 letters including one from a current supervisor and program director

2.0.2 Temporary Exception Status from Professional Licensure

In May of 2004, DHS addressed modifications to Professional Licensure for individuals eligible to obtain licensure from Rhode Island Department of Health. A policy change was made by DHS to include *Temporary Exception from Professional Licensure*. The following categories apply for *Temporary Exception from Professional Licensure* (See Table 9):

2.0.2.1 Temporary Exception Status for License Eligible Healthcare Professionals

Temporary Exception from Professional Licensure pending successful completion of a professional licensure examination is a <u>time limited condition</u> that allows provider-agencies to address staffing for Clinical Supervision and Treatment Consultation positions by considering for employment *license eligible* individuals.

The *license eligible* individual must have an established a date to sit for the respective licensing examination by the Rhode Island Department of Health within 6 months of hire.

Rhode Island Department of Health licensure pertains to the following professions eligible to provide Clinical Supervision and Treatment Consultation for HBTS:

- 1. Licensed Clinical Social Worker (LCSW)
- 2. Licensed Independent Clinical Social Worker (LICSW)
- 3. Mental Health Clinician (LMHC)
- 4. Marriage and Family Therapist (LMFT)
- 5. Psychologist

Individuals scheduled to take a professional licensure examination and receive licensure from Rhode Island Department of Health must submit to DHS the following documentation to qualify for Temporary Exception Status as licensed eligible:

- 1. Proof of the *license eligible* individual's professional examination date
- 2. Proof that the *license eligible* individual has received at least 50 hours of postgraduate clinical supervision from a licensed health care professional with appropriate clinical expertise prior to seeking this accommodation

- 3. Proof that the *license eligible* individual must have acquired at least 1 year of postgraduate supervised work experience with children having special health care needs
- 4. Proof that the *license eligible* individual will receive clinical oversight from the licensed health care professional with responsibility for the plan
- 5. The provider-agency must inform DHS of the candidate's examination results as soon as possible. Should the candidate fail the licensure examination, DHS will maintain the candidate's Temporary Status until the next regularly scheduled examination date, or for 12 months following the first attempt at licensure. The provider-agency has full responsibility for enforcing these requirements and must keep DHS informed about a candidate's status

In no circumstance will *Temporary Exception Status* last longer than 12 months to allow for a retaking of the examination

2.0.2.1.1 Licensed Clinical Social Worker (LCSW)

An individual with the credential of *Licensed Clinical Social Worker* may be provided with *Temporary Exception*. This status is conditional and offered on a case-by case basis given complete adherence to the following requirements:

- 1. Proof that the individual has received the credential of *Licensed Clinical Social Worker* from Rhode Island Department of Health at the time of provider-agency hire
- 2. The individual must obtain full licensure as an Independent *Clinical Social Worker* from the Rhode Island Department of Health as soon as possible. It is generally a two-year process; refer to DOH "Qualifications for Licensure"
- 3. Proof that the individual received at least 50 hours of postgraduate clinical supervision from a licensed health care professional with appropriate clinical expertise prior to seeking this accommodation
- 4. Proof that the individual acquired at least 1 year of postgraduate supervised work experience with children having special health care needs
- 5. The provider-agency must indicate when the individual will take and complete examination for the credential of *Licensed Independent Clinical Social Worker (LICSW)*. The provider-agency is responsible to require the candidate to comply with this stipulation
- 6. Proof that the individual will receive clinical oversight from the licensed health care professional with responsibility for the plan

2.0.2.2 Temporary Exception for Other Credentialed Professionals

The categories of Temporary Exemption Status for Other Credentialed Professionals includes individuals with professional certifications or are eligible to receive a certification from a professional body or state agency.

On a case-by-case basis, DHS will review a request from provider-agencies to apply the designation of *Temporary Exception for Other Credentialed Professionals* for the purposes of rendering Clinical Supervision and Treatment Consultation services. In all cases, clinical oversight is required and must be provided by a licensed healthcare professional. Other credentialed professionals eligible for Temporary Exception Status are:

2.0.2.2.1 Professional Teacher Certification as a School Social Worker

On a case-by-case basis, an individual in possession of a valid *Professional Teacher Certification as a School Social Worker* with a Master's Degree in Social Work shall be considered. The provider-agency must provide DHS with the following documentation demonstrating that the individual has obtained:

- 1. One year of postgraduate clinical experience with children with special health care needs,
- 2. One year of postgraduate clinical supervision (i.e., 50 hours) provided by an LICSW, and
- 3. Licensure at LCSW level from the Rhode Island Department of Health

Each provider-agency shall verify if the individual is able to engage in services outside of a particular school system by conferring with the individual's respective LEA. It is the LEA and not DHS that governs whether an individual can engage in outside employment.

2.0.2.2.2 School Psychologist

Individuals with certification from Rhode Island Department of Education as a <u>School Psychologist</u> may be considered. The provider-agency must submit written documentation fully addressing certification or its equivalency, which may be met by one of the following conditions:

- 1. An individual with a valid Rhode Island Department of Education Provisional Certification (i.e., Advanced Degree in School Psychology) or Professional Certification as a School Psychologist in grades PK –12; or
- 2. An individual certified as a School Psychologist in another state may be considered for exemption providing reciprocity from the Rhode Island Department of Education is obtained within 12 months from date of hire;

- 3. Each provider-agency shall verify if a candidate is able to engage in services outside of a particular school system by conferring with the individual's respective Local Educational Authority (LEA); or
- 4. An individual lacking certification from the state of Rhode Island or another state as a School Psychologist shall have valid certification from the National Association of School Psychologists (NASP) and maintain this certification to seek Exemption Status from DHS

In all cases relating to a School Psychologist, the following is required:

- 1. Proof of valid certification must be supplied to support a candidate's request
- 2. Certification must be maintained throughout a course of employment and available to DHS as part of a provider-agency's annual report

2.0.2.2.3 Teacher with a Masters Degree in Special Education

An individual in possession of a valid Professional Teacher Certification from the Rhode Island Department of Education and possession of a Masters degree in Special Education shall be considered given the following areas of expertise:

- 1. Early Childhood Special Educator (Birth through Kindergarten)
- 2. Special Educator for Elementary and Middle School Level
- 3. Special Educator for Middle and Secondary Level
- 4. Special Educator for Deaf/Hard-of-Hearing
- 5. Special Educator for Blind/Partially Sighted
- 6. Special Educator for Severe/Profound Disabilities

2.0.2.2.4 Rhode Island Department of Mental Health, Retardation and Hospitals (MHRH)

MHRH certifications for *Counselor, Mental Health Counselor, or Principal Counselor* may be appropriate shall be considered. DHS does not administer any aspect of certification offered by MHRH.

Please refer to MHRH policies (as of August 2002) for specific requirements governing certifications as *Counselor*, *Mental Health Counselor*, or *Principal Counselor*. The MHRH phone number is 462-2338

2.1 Reciprocity and Temporary Exception Status

On a case-by case basis, provider-agencies may seek consideration under the category of reciprocity. The following categories apply:

2.1.1 Individuals with Professional Licenses from Other States

The provider-agency must submit all of the following information to DHS:

- 1. The individual must have an active license from an issuing state
- 2. The individual has applied for reciprocity including the expected date of licensure from the Department of Health in Rhode Island
- 3. The provider-agency must confirm that licensure has been provided by the Department of Health in Rhode Island

In the rare and unlikely instance where reciprocity may be denied, the provider-agency may seek to obtain an additional 12-month grace period in which to rectify matters with the Department of Health. The provider-agency must submit the following information to DHS:

- 1. The individual must have an active license from an issuing state
- 2. The provider-agency must then submit a written plan of action including a copy of requirements from the Department of Health that remain to be satisfied

2.1.2 Certification from National Association of Social Workers (NASW)

In the case of an individual for whom <u>licensure has expired</u> (regardless of the issuing State), DHS will consider for Temporary Exception Status <u>if</u> the individual presents with a valid certification as a Social Worker and/or related specialty certifications from NASW. The provider-agency must submit on behalf of the candidate the following information:

- 1. Proof of current NASW membership
- 2. One year of postgraduate clinical experience with children with special health care needs
- 3. One year of postgraduate clinical supervised experienced (i.e., 50 hours) from an LICSW having expertise and competence in the area of children with special health care needs

Individuals with NASW certification are required to apply for licensure by Rhode Island Department of Health. Certification from NASW should permit a candidate to become licensed as a Social Worker at the LICSW level.

NASW certification alone is not a sufficient prerequisite. The individual must obtain professional licensure. DHS may grant a 12-month grade period from date of hire in order to achieve licensure. There are no exceptions to this stipulation.

2.1.3. Continuing Education Requirement:

All non-licensed staff approved to provide Clinical Supervision and Treatment Consultation must receive continuing education throughout the course of employment from accredited programs with national or regional certifying authority. Ten (10) hours per year are required.

2.1.4 Clinical Oversight of Non-Licensed Individuals

Clinical Oversight is required for all non-licensed staff engaged in the delivery of clinical care. Clinical Oversight can only be provided by a licensed healthcare professional. The licensed healthcare professional must fully satisfy all of the following requirements:

- 1. The licensed health care clinician must meet expertise and competency requirements to able to render professional oversight
- 2. The licensed clinician must practice with the limits of his or her professional licensure

It is insufficient and unacceptable to use individually licensed clinicians that provide Specialty Consultations (i.e., OT, PT, and SLP) for the purposes of rendering clinical oversight for the entire HBTS Treatment Plan.

2.1.5 Exemption from Professional Licensure

On a case-by-case basis, DHS will consider a qualified candidate for *Exemption from Professional Licensure* with one of the following credentials (See Table 9):

2.1.5..1 Board Certified Behavior Analyst (BCBA) or Associate Certified Behavior Analyst (BSABA)

For information regarding standards and applications for examination, the address for the Behavior Analyst Certification Board (BACB) is:

BACB Metro Building – Suite 102 1705 Metropolitan Boulevard Tallahassee, Florida 32308-3796

Web: info@BACB.com

Individual's qualified to receive Exemption from Professional Licensure do not require professional oversight. Such individuals can also provide Clinical Supervision and Treatment Consultation.

Insert Table 9

Table 9: Summary of Licensure and Practice Standards

TEMPORARY EXCEPTION FROM PROFESSIONAL LICENSURE

Licensed Eligible	Exam Date -	Duration of	1 Year Post Graduate Work	50 Hours Clinical	Credentials Required:	Clinical Oversight
Categories	6 months	Exception	- Children with	Supervision	Certification	Required
	of Hire	•	Special Health	•	or	-
			Care Needs		Licensure	
LCSW	NA	12 months	YES	YES	Must have LCSW	YES
LICSW	YES	24 months	YES	YES	License Eligible	YES
LMHC	YES	12 months	YES	YES	License Eligible	YES
LMFT	YES	12 months	YES	YES	License Eligible	YES
NASW	YES	12 months	YES	YES	License Eligible	YES
Psychologist	YES	12 months	YES	YES	License Eligible	YES
School Social	NA	NA	YES	YES	RIDE Certification	YES
Worker						
					RIDE Certification	YES
School					or	
Psychologist					NASP	
					Certification	
Counselor	YES	NA	YES	YES	MHRH	YES
Mental Health	YES	NA	YES	YES	MHRH	YES
Counselor						
Principal	YES	NA	YES	YES	MHRH	YES
Counselor						
Special	NA	NA	YES	Required	RIDE Certification	YES
Educator with				Teaching		
Masters Degree		ψ 7 Γ1 1	, :1 T	Experience		

^{*}The above cannot provide Treatment Consultation

EXEMPTION FROM PROFESSIONAL LICENSURE

Eligible Categories	Exam Date - 6 months of Hire	Duration of Exemption	1 Year Post Graduate Work - Children with Special Health	50 Hours Clinical Supervision	Credentials Required: Certification or	Clinical Oversight Required
	of thic		Care Needs		Licensure	
BCBA*	NA	NA	YES	YES	BCBA	No
BSABA*	NA	NA	YES	YES	BSABA	No

^{*}Meets qualifications to be engaged as a Clinical Supervisor or Treatment Consultant

APPENDIX 5: DESCRIPTIONS OF CONDITIONS ASSOCIATED WITH TARGET POPULATION

Diagnostic conditions for which HBTS may be appropriate can include, but are not limited to, those noted below. This is provided as a point of reference only. For many children with the conditions or diagnoses noted here, HBTS will not be the optimum service. HBTS may be effective for children with diagnoses other than those noted here. Appropriateness determination should be based on multiple factors and not diagnosis alone.

- Autistic Spectrum Disorders and Pervasive Developmental Disorders refers to a wide continuum of associated cognitive and neuro-behavioral disorders characterized by, but not limited to, three core defining features: impairments in reciprocal social interactions, impairments in verbal and nonverbal communication, and restricted and repetitive patterns of behaviors or interests. There is marked variability in the severity and complexity of symptomatology across individuals as well as intellectual functioning that can range from profound mental retardation to the superior level of cognitive ability.
 - Autistic Disorder
 - Pervasive Developmental Disorder Not otherwise Specified
 - Asperger's Disorder
 - Rett's Syndrome (Rett's Disorder)
 - Childhood Disintegrative Disorder
- Mental Retardation refers to significantly sub-average general intellectual functioning accompanied by significant limitations in adaptive functioning (e.g., communication, self-care, home living, social or interpersonal skills, use of community resources, functional academic skills, work, leisure, health and safety). The severity of mental retardation and level of adaptive functioning varies given the degree of a child's impairment:¹⁰

Mild Mental Retardation
 Moderate Mental Retardation
 Severe Mental Retardation
 Profound Mental Retardation
 IQ level 35 – 50
 IQ level 20 – 35
 IQ level below 20

- Psychiatric and Behavioral Disorders refer to children and adolescents with a range of conditions, which result in impaired or compromised levels of functioning across various domains.
 - Attention Deficit Hyperactivity Disorder

⁰ The term developmental disability means a severe, chronic disability, other than mental illness, which: a) is attributable to a cognitive or physical impairment or combination of cognitive and physical impairments, b) is

- Conduct Disorder
- Intermittent Explosive Disorder
- Opposition Defiant Disorder
- Tourette's Disorder
- Mood Disorders (e.g., Depression and Bipolar Disorders)
- Anxiety Disorders (e.g., Panic Disorder, Post Traumatic Stress Disorder, Generalized Anxiety Disorder, Obsessive – Compulsive Disorder, and Social Phobia)
- Psychotic Disorders (e.g., Schizophrenic Conditions, Delusional Disorder, or Brief Psychotic Disorder)
- 1. General Medical and Physical Conditions refers to a wide range of conditions with complex genetic, metabolic and/or neurological factors that significantly effect a child's functioning. Some of these conditions are:
 - Angelman's Syndrome
 - Cerebral Palsy
 - Duchenne's Muscular Dystrophy
 - Klinefelters Syndrome
 - Landau-Kleffner Syndrome
 - Prader Willi Syndrome
 - Tuberous Sclerosis
 - Down's Syndrome

APPENDIX 6: PROVIDER-AGENCY RESPONSIBILITY FOR MONITORING MEDICAID ELIGIBILITY

A recipient's eligibility to receive Medicaid can change at any time. It is the responsibility of the HBTS provider to verify eligibility. This can be accomplished by contacting the Recipient Eligibility Verification System (REVS) 784-8100.

Loss of Medicaid coverage results in nonpayment of claims. Providers may request retroactive reimbursement from EDS once Medicaid coverage has resumed if the child is retroactively reinstated providing:

- There has been no lapse in coverage
- The child has had an approved HBTS Treatment Plan

APPENDIX 7: TREATMENT SUPPORT DOMAINS

With the development of new services for children with special health care needs, such as Personal Assistance Services and Supports (PASS), the degree to which Treatment Support is utilized may change. The following gives guidance to help understand the purpose of Treatment Support as a service that provides structure, guidance, redirection and supervision to a child. The following components help to distinguish this service from Respite. As part of an overall Treatment Plan, Treatment Support is a complement to Specialized Treatment.

Acquiring and Using Information: Is the application or use of information a child has learned. It involves being able to perceive relationships, reason, and make logical choices. Individuals think in different ways. Some children think in pictures, that is, they may solve a problem by watching and imitating what another person does. For others, thinking involves using language to understand others as well as to express oneself. Related tasks could involve:

- Learning to read, write, do arithmetic and understand new information
- Follow directions and instructions
- Ask for information
- Explain something
- Communicate basic information name, address, telephone number, yes/no, take/give messages, make requests, and express functional needs (e.g., toileting, drink, food, etc.)
- Learning to take a bus
- Shopping

Attending and Completing Tasks: Involves directing and sustaining attention while engaged in an activity or task. This means focusing long enough to begin and complete an activity and being able to return to it if distracted. Related tasks could involve:

- Attends to directions and instructions
- Listens and attends to what others are saying
- Remains at a designated task or activity for a specified time

Interacting and Relating to Others: Involves participating with one's family and others for practical and social purposes. Interactions and relating require a child to respond to a variety of emotional and behavioral cues. Speaking intelligently and fluently, turn taking, responding to authority, and understanding another person's feelings form the foundation of social interactions. Related tasks could involve:

- Playing games and turn taking
- Developing and using manners while in the community or at home
- Using appropriate language
- Joining community activities
- Helping others

Caring for Your Self: Is important to a child's sense of mastery and development of competence. It involves engaging in self-care activities as independently as possible for physical, developmental and emotional needs. Related tasks could involve:

- Hygiene activities
- Grooming
- Arranging, preparing meals, and eating meals
- Doing laundry
- Selecting clothes and dressing

Maintaining Health and Physical Well-being: Involves developing an understanding of daily habits that are necessary to good health. For a child, this means learning to recognize healthy practices as well as having time to engage in meaningful recreation. Related tasks could involve:

- Outdoor activities
- A schedule of physical exercise
- Participating in after school sports
- Movement exercises
- Media activities computer, television, and video games

APPENDIX 8: FACTORS RELATED TO TREATMENT INTENSITY FOR HBTS SPECIALIZED TREATMENT

Children with behavioral and/or developmental conditions learn in complex ways with unique learning styles. They demonstrate highly individualized differences in cognitive and affective processes, language, motor, visual, auditory, spatial, and relationship capacities.

Some professionally recognized interventions include:

- Applied Behavior Analysis (ABA)
- ABA Discrete Trial Teaching
- Developmental Social Interventions including communication therapies and the Developmental, Individual Difference, Relationship-based approach (DIR)
- Educational models exemplified by TEACCH (Treatment and Education of Autistic and Related Communication Handicapped Children)

RELEVANT TERMINOLOGY

Applied Behavior Analysis (ABA):

Refers to a program of systematic analysis and tracking of behavior including its cues and consequences (i.e., reinforcement).

ABA Discrete Trial Teaching:

This is a rigorous form of behavioral intervention that is time intensive, highly structured, uses repetitive sequences, contingent reinforcement, and requires extensive collection and analysis of data.

Developmental, Individual Difference, Relationship-Based Approach (DIR):

A comprehensive model proposed by the Interdisciplinary Council of Developmental and Learning Disorders that focus on the functional developmental capacities, individual differences and relationship interactions when assessing a child.

Developmental-Social Pragmatic Instruction:

Developmental Social Pragmatic Instruction is a model of intervention that includes aspects of behavior theory, language development and social-communicative functions. Strategies typically involve imitation, solitary play, group play, social skills training, use of social stories,

incidental or experiential learning opportunities, and various communication modalities (e.g., picture exchange communication system, language board, or Bliss symbols).

TEACCH:

TEACCH uses educational and visual communication strategies to establish routines and to reduce frustration. This approach seems to be especially helpful when language is very delayed.

APPENDIX 9: PARENT PARTICIPATION IN HBTS

Parent participation in the development of a child's HBTS Treatment Plan is required. Parent or guardian approval of all Treatment Plans – initial, revised, and renewal requests for continuing HBTS – is required also. Treatment Plans must contain a parent or guardian's signature. It is recommended that provider-agencies create a separate page that includes a statement indicating that the parent has participated in the development of the plan, reviewed it, and supports the plan. The parent or guardian shall sign this page and it must be included in the proposal.

A child's parent or guardian is expected to be present when home-based therapy is being provided. At other times, it may be appropriate for the home-based worker to work with a child alone and/or participate in community-based activities necessary to accomplish defined therapeutic objectives.

For families involved with DCYF and having a case plan, provider-agencies must consult and coordinate all aspects of a child's HBTS plan to insure compliance with DCYF mandates or expectations for a family under its care.

APPENDIX 10: PROVIDER-AGENCY RESPONSIBILITIES FOR DISCONTINUATION OF HBTS

DHS recognizes that there may be critical situations whereby continued HBTS becomes compromised or clinically inappropriate, necessitating suspension or discontinuation of care. These can include, but are not limited to, risks to the safety and welfare of the child, family or home based worker, and treatment non-compliance. As such, the HBTS provider-agency has responsibility to exercise prudent judgment and prevent abandonment of care when responding to crisis situations. The following guidelines are to be followed:

- 1) Parental participation in the development and application of a child's HBTS Treatment Plan is an ongoing process whereby critical situations or circumstances are identified and addressed
- 2) Written documentation regarding critical treatment concerns is required and will identify risks and actions to be taken to reduce or eliminate further recurrence of problem situations. A copy is to be maintained as part of the child's Treatment Plan. Possible responses may involve modifications to the authorized Treatment Plan, including referrals for emergency psychiatric evaluation, hospitalization, individual or family therapies, DCYF services, or other actions (e.g., reassessing treatment intensity, treatment objectives, treatment methods, etc.,)
- 3) When multiple efforts to resolve difficulties (including lack of participation in treatment by parents or guardians) have failed and are documented, the provider-agency can initiate discontinuing HBTS, namely:
 - a) The child's family or guardian as well as DHS or CEDARR Family Center must receive written notification 30 days prior to discontinuing HBTS. Reasons for discontinuing treatment must be stated. Alternative resources, and/or referrals if appropriate, must be given
 - b) All other providers and professionals related to the HBTS plan must receive written notice
- 4) DHS recognizes that provider-agencies may need to remove a home-based worker immediately when confronted with sexual harassment, threats of violence, verbal abuse, assault, or health risks (e.g., individual who is substance impaired). As such, it may be inappropriate to resume HBTS, if at all, until circumstances have been fully resolved to the satisfaction of the HBTS provider-agency. DHS or CEDARR Family Center must receive immediate written notification when such situations develop, and be fully informed regarding problem resolution, which could include immediate suspension or termination of HBTS

- a) If immediate termination is indicated, the provider-agency has the responsibility to also notify DCYF when issues of child neglect or abuse are suspected
- b) Parents or guardians should also be informed about how to obtain emergency psychiatric services (e.g., contacting the police or local community mental health center) for immediate assistance
- 5) A parent or guardian has the right to terminate HBTS at any time during an authorized course of care. The provider-agency must have written policies to facilitate an orderly transition of care, and/or follow-up or referral services, and communicate with the CEDARR Family Center or DHS

APPENDIX 11:APPEAL RIGHTS RHODE ISLAND DEPARTMENT OF HUMAN SERVICES

APPEAL RIGHTS - READ CAREFULLY

You have a right to discuss this action further with my supervisor, or me or to request an adjustment conference with the appropriate DHS Supervisor. If you have questions regarding this notice, call the Agency representative at the telephone number listed on the first page of the notice.

You have the right to request and receive a hearing if you disagree with the decision made regarding the level or length of services, in the approved treatment plan. You must request a hearing in writing within thirty (30) days of this notice.

If you request a hearing regarding your medical services within ten (10) days of this notice, you will continue to received the current amount of Medical Assistance Services until a hearing decision is made.

The form to request a hearing is enclosed. If you request a hearing you may represent yourself or authorize another person, such as a relative or legal counsel to represent you. Free legal help may be available by calling Rhode Island Legal Services at 274-2652 (outside the Providence calling area, call toll free at 1-800-662-5034).

EXCEPTION: If this action implements a hearing decision, you may not have the right to another hearing on this action. See the hearing decision letter for your right for judicial review in accordance with Rhode Island law (42-35-1 et seq.).

TO REQUEST A HEARING

All requests must be in writing. To request a hearing, complete Section I., the 'Statement of Complaint' on the REQUEST FOR A HEARING form or else submit your complaint in writing. Briefly describe the Agency action you wish to appeal. You can fill out the form yourself, or with the help of the Agency representative if you need help in completing the form. The form is signed by the person to whom the notice is addressed or her/his representative.

Mail or bring the hearing request form to the Center for Child and Family Health, Department of Hunan Services Forand Building, 600 New London Avenue, Cranston, RI 02920. In order to receive a hearing, you must do so within the time periods specified on this page. You will be notified of the time and place of the hearing. At the same time, you will also receive a statement of the Agency's position, an explanation of the policy on which the decision was based, and additional information about the hearing process.

INFORMATION ABOUT HEARINGS FOR APPLICANTS AND RECIPIENTS OF FINANCIAL ASSISTANCE, FOOD STAMPS, MEDICAL ASSISTANCE AND SOCIAL SERVICES

The Department of Human Services (DHS) has a responsibility to provide financial assistance, food stamps, medical assistance, and social services to individuals and families for whom eligibility is determined under the provisions of the Social Security Act, the Rhode Island Public Assistance Act, the Food Stamp Act, the Rhode Island Medical Assistance Act and Title XCX Social Services.

The hearing process is intended to insure and protect your right to assistance and your right to have staff decisions reviewed when you are dissatisfied. You have asked for a hearing because of an agency decision with which you disagree. The following information is sent to help you prepare for your hearing and to inform you about what you may expect and what will be expected of you when it is held.

1. WHAT IS A HEARING?

A hearing is an opportunity provided by the Department of Human Services to applicants or recipients who are dissatisfied with a decision of the agency, or a delay in such a decision for a review before an impartial appeals officer to insure correct application of the law and agency administrative policies and standards.

2. WHO CONDUCTS A HEARING?

A hearing is conducted by an impartial appeals officer appointed by the Director of the Department of Human Services to review the issue(s) and give a binding decision in the name of the Department of Human Services.

3. WHO MAY ATTEND A HEARING?

A hearing is attended only by persons who are directly concerned with the issue(s) involved. You may be represented by legal counsel if you chose and another witness or a relative or friend who can speak on your behalf. The Agency is usually represented by the staff member involved in the decision and/or that worker's supervisor. Legal services are available to persons wishing to be represented by legal counsel through Rhode Island Legal Services (274-2652) or (1-800-662-0534).

If an individual chooses to have legal representation, e.g. be represented by an attorney, paralegal, or legal assistant, the representative must file a written Entry of Appearance with the Hearing Office at or before the hearing. The Entry of Appearance acts as a release of confidential information, allowing the legal representative access to the agency case record. It is also needed for the Hearing Office to confirm the representation for purposes of follow-up, review, request for continuances, etc.

4. WHERE IS THE HEARING HELD?

The hearing may be held at a regional or district office or in an individual's home when circumstances require.

5. HOW CAN YOU LEARN ABOUT THE DEPARTMENT'S RULES AND REGULATIONS?

Section III of DHS-121 form shows the policy manual references, which are at issue in your hearing. You may review the Department's regulations at any local welfare office during regular business hours.

You may also review the Department's hearing decisions rendered on or after April 1987. They are available only at the DHS Central Administration Building, 600 New London Avenue, Cranston Rhode Island, between the hours of 9:00 a.m. and II:00 a.m. and between the hours of 1:00 p.m. and 3:00 p.m. Monday through Friday.

6. WHAT ARE YOUR RIGHTS RELATIVE TO THE HEARING?

You have a right to examine all documents and records to be used at the hearing at a reasonable time before the date of the hearing, as well as during the hearing.

You may present your case in any way you wish without undue interference, by explaining the situation yourself or by having a friend, relative, or legal counsel speak for you, and you may bring witnesses and submit evidence as discussed above to support your case. You will have an opportunity to question or refute any testimony or evidence and to confront and cross-examine adverse witnesses.

7. HOW IS A HEARING CONDUCTED?

A hearing differs from a formal court procedure because you are not on trial and the appeals officer is not a judge in the courtroom sense. However, any person who testifies will be sworn in by the appeals officer.

After you have presented your case, the staff member will explain the provisions in law or agency policy under which s/he acted. When both sides have been heard, there will be open discussion under the leadership and guidance of the appeals officer. The entire hearing is recorded on tape.

8. HOW WILL THE HEARING DECISION BE MADE?

The tape recording of the testimony of the persons who participated in the hearing, together with all papers and documents introduced at the hearing, will be the basis for the decision.

The appeals process is generally completed within 30 days of the receipt of your request, but will never exceed sixty (60) days for food stamps and ninety (90) days for all other programs unless you request a delay, in writing, to prepare your case.

The appeals officer will inform you of her/his findings, in writing, following the hearing. If you are still dissatisfied, you have a right to judicial review of your case. The agency staff member wants to be as helpful as possible in assisting you to prepare for the hearing. If you have any questions about what you may expect, or what may be expected of you, be assured that you may call your eligibility technician or worker.

APPENDIX 12: EXAMPLES OF GLOBAL ASSESSMENT SCALES

GLOBAL ASSESSMENT OF FUNCTIONING (GAF) SCALE

DESCRIPTION

RANGE

100	0.1	Superior functioning in a wide range of activities, life's problems never seem to get out of hand, is sought

- Superior functioning in a wide range of activities, life's problems never seem to get out of hand, is sought out by others because of his or her many positive qualities. No symptoms.
- Absent or minimal symptoms (e.g., mild anxiety before an exam), good functioning in all areas, interested and involved in a wide range of activities, socially effective, generally satisfied with life, no more than every day problems or concerns (e.g., an occasional argument with family members).
- If symptoms are present, they are transient and expectable reactions to psychosocial stressors (e.g., difficulty concentrating after family argument); no more than slight impairment in social, occupational, or school functioning (e.g., temporarily falling behind in schoolwork).
- Some mild symptoms (e.g., depressed mood and mild insomnia) OR some difficulty in social, occupational, or school functioning (e.g., occasional truancy, or theft within the household), but generally functioning pretty well, has some meaningful interpersonal relationships.
- Moderate symptoms (e.g., flat affect and circumstantial speech, occasional panic attacks) OR moderate difficulty in social, occupational, or school functioning (e.g., few friends, conflicts with peers or coworkers).
- Serious symptoms (e.g., suicidal ideation, severe obsessive rituals, frequent shoplifting) OR any serious impairment in social, occupational, or school functioning (e.g., no friends, unable to keep a job).
- 40 31 Some impairment in reality testing or communication (e.g., speech is at times illogical, obscure or irrelevant) OR major impairment in several areas, such as work or school, family relations, judgment, thinking, or mood (e.g., depressed man avoids friends, neglects family, is unable to work; child frequently beats up younger children; is defiant at home, and is failing in school).
- 30-21 Behavior is considerably influenced by delusions or hallucinations OR serious impairment in communication or judgment (e.g., sometimes incoherent, acts grossly inappropriately, suicidal preoccupation) OR inability to function in almost all areas (e.g., stays in bed all day; no job, home or friends).
- 21 11 Some danger of hurting self or others (e.g., suicide attempts without clear expectation of death; frequently violent; manic excitement) OR occasionally fails to maintain minimal personal hygiene (e.g., smears feces) OR gross impairment in communication (e.g., largely incoherent or mute).
- 10-1 Persistent danger of severely hurting self or others (e.g., recurrent violence) OR persistent inability to maintain minimal personal hygiene OR serious suicidal act with clear expectation of death.

From: American Psychological Association DSM - IV

Adaptation - CHILDREN'S GLOBAL ASSESSMENT SCALE (4 Through 16 years)

Rate the subject's level of functioning in the last 3 months by selecting the level which describes his/her functioning on a hypothetical continuum of mental health illness. For example, a subject who exhibits "difficulties...previously differentiation of reality and fantasy" (level 30) should be noted at that level even though he/she has "major impairment in functioning..." (Level 40). Rate actual functioning independent of whether or not subject is receiving and may be helped by treatment or has previously shown better or worse functioning (i.e., history).

Range	Description
100 - 90	Superior functioning in many areas, good function in all areas.
89 – 80	Good functioning in all areas. Fundamentally secure in family, school and with peers so that situational responses are transitory, non-symptomatic (i.e., acknowledged and managed by the child), and do not interfere with functioning.
79 - 70	<u>Slight interference with functioning</u> in family, school or with peers. Health responses to situational crisis may produce symptoms, but symptoms are minimal, brief and only slightly interfere with functioning.
69 - 60	<u>Some difficulty in functioning</u> in family, school, or with peers due to normal responses to developmental crisis (e.g., age-appropriate phobia, separation anxiety), but these symptoms do not seriously impair functioning. The symptomatic behavior would ne be sufficiently intense to label the child as disturbed.
59 - 50	Moderate difficulty in functioning in family, school, or with peers due to mild symptoms (mild adjustment reactions, reactive disturbance, mild peer relationship problems, mild psychosomatic problems, mild habit or conduct disturbances, bed wetting, neurotic traits). Functioning may be constricted, but still appropriate.
49 – 40	<u>Clear interference in functioning</u> in family, school, or with peers due to serious impairments in personality development (e.g., personality disorders, oppositional child, impulsive-ridden child, poorly socialized child). Most clinicians would agree that these symptoms represent disturbance.
39 - 30	<u>Major impairment in functioning</u> in family, school, or with peers due to sever symptoms (e.g., psychoneurotic disorders, severe behavior disorders, withdrawal aggression, hyperactivity, severe or persistent psychosomatic complaints, recurrent destructive or self-destructive behavior, compulsiveness, obsessions, severe or frequent anxiety).
29 – 21	<u>Unable to function in some but not all areas</u> in family, school, or with peers due to gross disruption in behavior or difficulties in object relations or in differentiation of reality and fantasy.
20 - 19	<u>Needs almost constant supervision</u> due to severely destructive or self-destructive behavior or gross impairment in reality testing, communication, cognition, affect, object relations or personal hygiene.
10 - below	<u>Needs constant supervision</u> (24 hour care) due to severely destructive or self-destructive behavior or gross impairment in reality testing, communication, cognition, affect, object relations or personal hygiene.
Autistic, symbio to function.	tic, psychotic, or borderline children may appear anywhere from 1 to 30 on this scale, depending on ability

David Shaffer, Division of Child and Adolescent Psychiatry, Columbia University, New York Psychiatric

Institute, 722 West 168th Street, Unit 78, New York, NY 10032

Reference:

APPENDIX 13: DESCRIPTIONS OF ACCREDITATION ABBREVIATIONS

COA	Council On Accreditation	
CARF	Commission On Accreditation Of Rehabilitation Facilities	
CORF	Comprehensive Outpatient Rehabilitation Facility	
DCYF	Department Of Children Youth And Families	
DOE	Department Of Education	
DOH	Department Of Health	
DOL	Department Of Labor	
JCAHO	Joint Commission On Accreditation Of Health care Organizations	
MHRH-DD	Mental Health Retardation and Hospitals – Developmental Disabilities	
ORS	Office Of Rehabilitation Services	

APPENDIX 14: GUIDELINES FOR CONSUMERS OF APPLIED BEHAVIOR ANALYSIS SERVICES TO INDIVIDUALS WITH AUTISM

Autism Special Interest Group (SIG), Association for Behavior Analysis Adopted May 23, 1998

The Autism Special Interest Group (SIG) of the Association for Behavior Analysis believes that all children and adults with autism and related disorders have the right to effective education and treatment based on the best available scientific evidence. Research has clearly documented the effectiveness of Applied Behavior Analysis (ABA) methods in the education and treatment of people with autism. Planning, directing, and supervising effective ABA programs for people with autism requires specific competencies. Individuals with autism, their families, and other consumers have the right to know whether persons who claim to be qualified to direct ABA programs actually have the necessary competencies. Consumers also have the right to hold those individuals accountable for providing quality services, i.e., to ask them to show how they use objective data to plan, implement, and evaluate the effectiveness of the interventions they use.

Formal credentialing of professional behavior analysts (i.e., registration, certification, or licensure) can provide safeguards for consumers, including means of screening potential providers and some recourse if incompetent or unethical practices are encountered. At present, however, procedures for credentialing professional behavior analysts are in place in only a few states. Until they are implemented more widely, the Autism SIG recommends that consumers seek to determine if those who claim to be qualified to direct ABA programs for people with autism meet the following minimum standards:

I. The qualifications embodied in the standards for certification as a behavior analyst in the State of Florida, Department of Children and Families, which can be summarized as follows¹¹:

Formal training

- Master's or doctorate in behavior analysis, or in psychology, special education, or another human service discipline with an emphasis in behavior analysis
- Coursework in principles of learning, principles of behavior, or basic behavior analysis; experimental analysis of behavior; behavioral assessment or methods of direct observation of behavior; applied behavior analysis; single-subject research designs; legal and ethical Issues
- Supervised practicum, internship, or employment experiences in applied behavior analysis

For details, see "Identifying qualified professionals in behavior analysis" by G.L. Shook & J.E. Favell in *Behavioral Intervention for Young Children with Autism*, edited by C. Maurice, G. Green, & S.C. Luce; Austin, TX: PRO-ED, 1996; and "Essential content for training behavior analysis practitioners," by G. L. Shook, F. Hartsfield, & M. Hemingway, *The Behavior Analyst*, 1996, Vol. 18, pp. 83-91

Competencies

- Ethical considerations
- Definition and characteristics of applied behavior analysis
- Basic principles of behavior
- Descriptive analysis
- Demonstrating functional relations
- Measurement of behavior
- Data display and interpretation
- Selection of target behaviors and goals
- Behavior change procedures
- Generalization and maintenance of behavior change
- Managing emergencies
- Transfer of technology
- Support for behavior analysis services

II. Additional training and experience in directing and supervising ABA programs for individuals with autism:

- Formal training and/or self-study to develop knowledge of the best available scientific evidence about the characteristics of autism and related disorders, and implications of those characteristics for designing and implementing educational and treatment programs, including their impact on family and community life
- Formal training and/or self-study to develop knowledge of at least one curriculum consisting of:
 - Scope and sequence of skills based on normal developmental milestones, broken down into component skills based on research on teaching individuals with autism and related disorders
 - Prototype programs for teaching each skill in the curriculum, using behavioral methods
 - Data recording and tracking systems
 - Accompanying materials
 - At least one full calendar year (full time equivalent or 1000 clock hours [@ 25 hrs/wk for 40 weeks]) of hands-on training in providing ABA services directly to children and/or adults with autism under the supervision of a behavior analyst with a master's or doctorate and at least 5 years' experience in ABA programming for individuals with autism. The training experience should include at a minimum:
- a. Provision of AHA programming to at least 5 individuals with autism
- b. Designing and implementing individualized programs to build skills in each of the following areas: "learning to learn" (e.g., observing, listening, following instructions, imitating); communication (vocal and nonvocal); social interaction; self-care; academics; school readiness; self-preservation; motor; play and leisure; community living; work.
- c. Using both discrete-trial and incidental or "naturalistic" teaching methods to promote skill acquisition and generalization

- d. Incorporating the following into skill-building programs: prompting; error correction; discrimination training; reinforcement strategies; strategies for enhancing generalization
- e. Modifying instructional programs based on frequent, systematic evaluation of direct observational data
- f. Designing and implementing programs to reduce stereotypic, disruptive, and destructive behavior based on systematic analysis of the variables that cause and maintain the behavior
- g. Incorporating differential reinforcement of appropriate alternative responses into behavior reduction programs, based on the best available research evidence
- h. Modifying behavior reduction programs based on frequent, systematic evaluation of direct observational data
- i. Provision of training in ABA methods and other support services to the families of at least 5 individuals with autism
- j. Provision of training and supervision (at least I hour of supervision per 10 hours of client contact for at least one-half of the training period) to at least 5 professionals, paraprofessionals, or college students providing ABA services to individuals with autism

The Autism SIG urges consumers to ask prospective directors or supervisors of ABA services to provide documentation of their qualifications in the form of: membership in the Association for Behavior Analysis; degrees; letters of reference from employment supervisors and/or families for whom they have directed ABA programming for similar individuals with autism (with appropriate safeguards for privacy and confidentiality); any registration, certificate, or license in Applied Behavior Analysis per se (i.e., not psychology, special education, education, or another discipline with no emphasis in behavior analysis); results of any competency exams they may have taken in Applied Behavior Analysis; participation in professional meetings and conferences in behavior analysis; publications of behavior analytic research in professional journals. A few workshops, courses, or brief hands-on experiences do not qualify one to practice Applied Behavior Analysis effectively and ethically.

DISCLAIMER: This document suggests guidelines for consumers to use in determining who is qualified to direct Applied Behavior Analysis programs for individuals with autism, as recommended by the Autism Special Interest Group of the Association for Behavior Analysis.

APPENDIX 15: DOCUMENTATION GUIDELINES FOR HBTS

RHODE ISLAND DEPARTMENT OF HUMAN SERVICES Center for Child and Family Health October 2000

Early and Periodic Screening, Diagnosis and Treatment (EPSDT):

Home Based Therapeutic Service Providers

I. Documentation Requirements

A. Providers are required to keep all records necessary to fully disclose the nature and extent of the services provided to children receiving HBTS. Providers must furnish to DHS and/or the Medicaid Fraud Control Unit of the Attorney General's Office such records and any other information regarding payments for claimed or services rendered that may be requested. These guidelines are applicable to all children receiving home–based services authorized by DHS.

Documentation - The Basics

The following are the basic principles of documentation. They apply to all types of services in all settings (i.e., Specialized Treatment, Treatment Support, Clinical Supervision, Treatment Consultation and Treatment Coordination).

- 1. The service/client record should be complete and legible
- 2. The documentation of each client/consumer encounter should include or provide reference to:
 - a) The reason for the encounter, and as appropriate, relevant history
 - b) Current status
 - c) Written treatment or progress notes including care provided and the setting in which the services were rendered
 - d) Date and time and legible identity/credentials of care provider
 - e) The amount of time it took to deliver the services

- 3. The client's progress, response to and changes in treatment and any revision of the Treatment Plan should be documented
- B. A clear trail must be maintained. Each provider is responsible for devising a system that documents those services, which have been provided. This back-up information is usually contained in the client record, daily log, or both and must be sufficiently detailed to show that a client received a specific number of hours of treatment services and that a corresponding number of hours were billed to Medicaid.
- C. All Home-Based Therapeutic Services must be provided in accordance with an approved comprehensive treatment program that clearly documents the medical necessity of the services. Treatment Plans must conform with Guidelines for the Review and Approval of Early and Periodic, Screening, Diagnosis and Treatment (EPSDT) Prescription Treatment Plan Proposals issued, November 1999 by DHS.

D. Methods of Documentation:

- 1. Information may be coded on a log or worksheet, however, weekly and monthly summaries of the overall relationship of the services to the treatment regimen (goals and objectives) described in the Treatment Plan with and update describing the client's progress and containing the clinician's judgment of the effects of the treatment must be recorded
- 2. The client's progress and current status in meeting the goals and objectives of his or her Treatment Plan must be regularly recorded in the client record in the form of progress notes. Progress notes must include:
 - a) Documentation of the implementation of the Treatment Plan
 - b) Chronological documentation of the client's clinical course
 - c) Significant events and/or changes in the client's condition should be documented with a full narrative note whenever they occur
 - d) Periodic documentation of all treatment provided to the client
 - e) Descriptions of the response of the client to treatment as well as the outcome of treatment
- 3. A discharge summary must be entered into the client record within a reasonable period of time after discharge. The Discharge Summary must include:

- a) Significant findings including final primary and secondary diagnoses
- b) General observations about the client's condition initially, during treatment and at discharge
- c) Whether the discharge was planned or unplanned and, if unplanned, the circumstances
- d) Assessment of attainment of the Treatment Plan objectives
- e) Documentation of referral to other appropriate program or agency

II. Monitoring and Quality Assurance

Site visits will be conducted by DHS staff to monitor appropriate use of Medicaid services and compliance with the procedures outlined in this manual. During these visits, staff will review the following:

- 1. Client records and Treatment Plans
- 2. Staff orientation programs and attendance logs
- 3. Agency policy and procedures related to HBTS service provision
- 4. Claims information/documentation
- 5. Staff time sheets
- 6. Complaint log

Providers will be notified of DHS site visits in advance if possible. Unannounced site visits may also be conducted at the discretion of the Department. DHS staff may contact or visit families as part of the oversight and monitoring activities.

In the event of adverse findings of a minor nature, repayment to DHS will be required. In situations where, in the opinion of the Department, significant irregularities in billing or utilization are revealed, providers may be required to do a complete self-audit in addition to making repayments. In either case, technical assistance in developing and implementing a plan of corrective action, where appropriate and applicable, will be offered to the provider.

In addition to monitoring conducted by DHS, providers are subject to periodic fiscal and program audits by the Health Care Financing Administration.

III. Client Record Guidelines

All Home Based Therapeutic Services must be provided in accordance with a comprehensive Treatment Plan that documents the <u>medical necessity</u> of the services. Treatment Plans for clients for whom providers are billing Medicaid must conform to the following guidelines:

- 1. Each client shall have a current written, comprehensive, individualized Treatment Plan that is based on assessments of the client's medical/behavioral needs
- 2. Responsibility for the overall development and implementation of the Treatment Plan must be assigned to an appropriate member of the professional staff
- 3. The Treatment Plan must be reviewed at major decision points in each client's course of treatment including:
 - a) The time of admission and discharge
 - b) A major change in the client's condition
 - c) The point of the estimated length of treatment and thereafter based on the estimated length of treatment, e.g., re-reviews of the Treatment Plan
 - d) At least every six months of treatment
- 5. The Treatment Plan must contain specific goals that the client must achieve and/or maintain as well as maximum growth and adaptive capabilities. These goals must be based on periodic assessments of the client and as appropriate, the client's family

IV. Supplemental Guidelines:

- 1. Medicaid is, by definition, a <u>medical</u> program, which pays for <u>medical</u> services. A Treatment Plan is regarded as a prescription for services and must be signed by an appropriate professional, in this case, a Licensed Practitioner of the Healing Arts.
- 2. The diagnosis must clearly be evident in the Treatment Plan and the diagnosis must be considered as the overall plan is developed. There must be a clear connection between the diagnosis and the symptoms of the condition for which the client is being referred.
- 3. The reasons for, and the amount and duration of each specific intervention should be evident in the plan.
- 4. Progress notes should reflect a judgment being made be the provider regarding the results of the treatment rendered, i.e., an assessment of why the interventions prescribed are/are not working. The notes should also show that the writer is aware of why things were done rather than merely what was done.

APPENDIX 16: CEDARR FAMILY CENTERS

About Families CEDARR Family Center

203 Concord Street, Suite 335 Pawtucket, RI 02860

Phone: 365-6855 Fax: 365-6860 www.aboutfamilies.org

About Families – Satellite Office 1

1 Cumberland Street, 4th Floor Woonsocket, RI 029895

Phone: 671-6533 Fax: 671-6532

About Families - Satellite Office 2

1 Frank Coelho Drive Portsmouth, RI 02871 Phone: 683-3570

Empowered Families CEDARR

82 Pond Street Pawtucket, RI 02860

Phone: 365-6103 Toll Free: 888-881-6380 Fax: 365-6123

www.empoweredfamilies.org

Family Solutions CEDARR

134 Thurbers Avenue, Suite 102 Providence, RI 02905

Phone: 461-4351 Fax: 461-4953 www.solutionscedarr.org

Family Solutions CEDARR - Satellite Office

Lafayette Mills 610 Ten Rod Road – Unit 13 North Kingstown, RI 02852

Phone: 294-6138 Fax: 277-3373

Families First CEDARR Center

Hasbro Children's Hospital 593 Eddy Street, Room 120 Providence, RI 02903

Phone: 444-7703 Fax: 444-6115

Additional CEDARR Family Centers may be certified periodically by DHS

APPENDIX 17: SERVICE MONITORING AND REPORTING REQUIREMENTS

Quarterly Report Protocol

Report 1: Provision of authorized direct service hours

- Step 1. Identify all treatment plans with an expiration date in the reporting period (e.g., January 1 March 31, 2006)
- Step 2. Log the client's name, client MID, date the treatment plan started, and date of treatment plan expiration in columns 1 4 on the Report 1 Worksheet
- Step 3. For each expiring treatment plan, sum the number of direct service hours authorized for the entire treatment plan authorization period (e.g., for a plan expiring on January 31, 2006, the treatment plan authorization period could be August 1, 2005 January 31, 2006) and enter the number of authorized hours in column 5 on the Report 1 Worksheet
- Step 4. For each expiring treatment plan, sum the number of direct service hours provided during the entire treatment plan authorization period and log the number of hours provided in column 6 on the Report 1 Worksheet
- Step 5. For each client identified divide the sum of provided hours by the sum of the authorized hours (column 6/column 5) and enter that percentage into column 7
- Step 6. If the percentage of authorized hours provided (column 7) is at least 75%, enter "yes" on column 9, meets standard. If column 7 is less than 75%, enter "no" on column 9, meets standard
- Step 7. Count the number of treatment plans for which meet the standard of at least 75% of authorized hours are provided ("yes" s in column 9)
- Step 8. Divide the number of "yes"s in column 9 by the total number of treatment plans with an expiration date in the reporting period. (Number of plans meeting standard yes's from column 9/total number of treatment plans with an expiration date in the reporting period)
- Step 9. Average the percentage of direct service hours at the bottom of the column
- Step 10. Report the percent from step 8 on the Quarterly Report sheet, in the row labeled Report 1, Provision of authorized direct service hours

Report 2: Timeliness of direct service initiation for new clients

- Step 1. Identify all new clients who had direct services initiate during the reporting period (e.g., January 1, 2006 and March 31, 2006)
- Step 2. Log the client's name, client MID, the date of receipt of treatment plan authorization, and date of the initiation of direct services (first date for which direct services are billed) in columns 1 4 on the Report 4 Worksheet
- Step 3. Calculate the number of days between the date the treatment plan authorization was received from the CEDARR Family Center (column 3) and the date of direct service initiation (column 4), and enter that number into column 5, timeliness of initiation of services
- Step 4. If the number of days between receipt of authorization from the CEDARR Family Center and the initiation of direct service (column 5) is ≤ 21 days, then enter "yes" in column 7, timeliness of initiation of direct services. If the number of days is greater than 21, enter "no" in column 7
- Step 5. Count the number of "yes's" in column 7
- Step 6. Divide the total number of clients for whom direct services were initiated within 21 days of receipt of treatment plan authorization by the number of clients for whom services were initiated during the reporting period to determine if the provider agency meets the performance standard (number of plans meeting standard "yes's" from column 7/total number of treatment plans for which services were initiated in the reporting period)
- Step 7. Report the percentage from Step 6, on the Quarterly Report sheet, in the row labeled Report 4, Timeliness of direct service initiation for new clients

Please submit reports on a quarterly basis using the attached Quarterly Reporting Form to:

Anne M. Roach, RN, MEd,
Consultant Public Health Nurse
Center for Child and Family Health
Rhode Island Department of Human Services
600 New London Avenue
Cranston, Rhode Island 02920

APPENDIX 18: RECERTIFICATION REQUIREMENTS AND PROCESS

To Be Supplied

APPENDIX 19: DOCUMENTATION GUIDELINES AND REQUIREMENTS FOR SPECIALTY CONSULTATIONS – OT, PT AND SLP

Effective May 1 of 2004, DHS issued practice and documentation requirements to provider-agencies for the inclusion of Treatment Consultation by specialists (i.e., Occupation Therapy, Physical Therapy, and Speech and Language Therapy). The following requirements must be met for the inclusion of these Specialty Consultations as part of a child's HBTS Treatment Plan.

- 1. A current and valid IEP or IFSP must be in force by the local LEA at the time that a clinical review takes place. To be considered current and valid, an IEP or IFSP must show a range of effective dates, for example, from May 1, 2004 to May 1, 2005, to be considered.
- 2. A current and valid IEP or IFSP must also list the frequency of each ancillary therapy being provided with specific treatment objectives. For example, SPL might be .5 hours 3 times per week, etc.
- 3. Treatment Consultation provided by a specialist for an HBTS Treatment Plan is appropriate to consider when techniques and objectives reinforce objectives provided in an IEP or IFSP, and that these objectives clearly relate to HBTS goals and objectives.
- 4. It is insufficient justification to consider specialty Treatment Consultation in an HBTS Treatment Plan if the IEP or IFSP lists monitoring of goals and objectives by any ancillary service (e.g., SPL, OT, or PT) as the primary objective. There must be evidence of active treatment taking place during a child's school day. Frequency, duration and treatment objectives by specialty must be readily identified in the IEP or IFSP.

Documentation Requirements:

It is the responsibility of the HBTS Treatment Consultant Specialist to explain and demonstrate how the inclusion of, OT, PT, or SPL goals and objectives are appropriate for a child's HBTS Treatment Plan

Documentation of Treatment Consultation by the Specialist (OT, PT, or SPL) is required using the attached form and must accompany the HBTS Treatment Plan.

Refer to the Specialty Treatment Consultant Supervision and Recommendations Form

Specialty Treatment Consultant Supervision and Recommendations Form

Client Name:						_	
Treatment Consultant Type:	Occupational Language	Therapy	Physica	l Therapy	, and/or	r Speech	anc
Consultant's Name:						_	
Date of Observation(s):						_	
Length of Observation(s):						_	
Place & Summary of Observ	ration(s):						
Recommendations for Trea exercises) and Implementation		Goals (sp	ecify), P	rocedures	(e.g., i	ntervention	s &

APPENDIX 20: HBTS PROVIDER AGENCIES AND PROGRAMS

- 1. Adeline LaPlante Memorial Center
- 2. ARC of Northern Rhode Island
- 3. Bradley Hospital
 - ❖ Home-Based Treatment Program
 - ❖ Intensive Behavioral Treatment (IBT)
- 4. Cranston ARC
- 5. Family Service of Rhode Island
- 6. Frank Olean Center
- 7. Groden Center
 - Children's Intensive Treatment Program (CIT)
 - ❖ In-Home Support and Training Program (IST)
- 8. John Hope Settlement House
- 9. Kent County Chapter RIARC Arthur Trudeau Memorial Center
- 10. Looking Upwards, Inc.
- 11. Ocean State Community Resources, Inc.
- 12. Perspectives Youth and Family Services
 - **❖** ABA Programs
 - Deaf and Hard of Hearing Program
 - Behavioral Health Program
- 13. Spurwink RI
- 14. TIDES Family Services
- 15. United Cerebral Palsy of Rhode Island, Inc. (UCPRI)

APPENDIX 21: PRIOR APPROVAL AND NON-PA SERVICES

DHS, as a function of its management and oversight responsibilities, can make changes to the Prior Approval process when circumstances warrant. The determination to remove services from PA is intended to improve provider-agency interactions regarding the processing of claims. Removing a service from PA in no way terminates the provider-agencies responsibility to fully comply with certification standards. DHS will continuously monitor utilization. DHS has the discretion to make changes to the forms used for PA and Non-PA Services. In each instance,

DHS will supply provider-agencies with these forms including necessary instructions, as needed.

1.0 Prior Approval Required Services

The following services must be listed on the <u>Rhode Island Medical Assistance Prior Authorization Request Form – HBTS Certified Provider Agency</u> at the time of Treatment Plan submission:

- □ Home Based Specialized Treatment
- □ Home Based Treatment Support

2.0 Non-PA Services

The following services must be listed on the <u>HBTS Form for Non-PA Services</u> at the time of Treatment Plan submission:

- Child Specific Orientation
- Clinical Supervision
- □ Group Intervention
- □ Lead Therapy
- □ Specialty Consultations (i.e., Occupational Therapy Consultation, Physical Therapy Consultation, & Speech and Language Therapy Consultation)
- □ Travel Reimbursement
- □ Treatment Coordination
- ☐ Treatment Consultation (including Pre and Post Treatment Consultation)

APPENDIX 22: TRAVEL TIME REIMBURSEMENT POLICY

The intent of this policy is to improve HBTS access and support the delivery of clinical services for children and families residing in outlying areas. The rationale used by DHS to formulate the HBTS Travel Reimbursement Policy is based on the following history:

- 1. DHS has been very concerned about improving access to HBTS while also seeking to make HBTS a reliable set of services for children with special health care needs. In part, the issuance of Certification Standards in February of 2003 began this process by establishing requirements and procedures for provider-agencies to follow while also linking referrals to HBTS through CEDARR Family Centers. Both steps have resulted in a more uniformed approach to helping families and children with special health care needs access services.
- 2. In May of 2004, DHS made a number of operational adjustments to HBTS practice requirements in an effort to make HBTS more flexible. Enhanced Home-Based Therapeutic Services (e.g., Pre and Post Treatment Consultation or Specialty Consultations) and CEDARR Enhanced Services were then initiated. Both were designed to provide immediate clinical services to families while enabling CEDARR Family Centers to also provide clinical supports while helping to identify other possible services like HBTS or PASS.
- 3. In May of 2004, DHS also broadened Variance to Professional Licensure in an effort to improve provider-agency's ability to recruit clinical staff.
- 4. As a result of subsequent site visits to provider-agencies and follow up discussions regarding their operations of HBTS as well as ongoing feedback from families, DHS began to reconsider the issue recognizing travel time for clinical staff, especially when seeking to serve outlying areas. On April 2, 2005, John Young, Associate Director of DHS, notified provider-agencies and CEDARR Family Centers of this policy change and identified a mechanism for accomplishing based on "zoned mileage."
- 5. Travel Time Reimbursement is intended to improve the access and utilization of HBTS for children and families residing in outlying areas of Rhode Island.
- 6. DHS will monitor the implementation of travel time reimbursement in order to measure the impact on service delivery.
- 7. Provider-agencies are expected expand and increase their service to children and families in outlying areas in a prompt and reliable manner. It is unsatisfactory for access to HBTS to remain problematic.
- 8. DHS reserves the right to make adjustments to the HBTS Travel Reimbursement Policy including its definition, application, or termination.

Travel Reimbursement Policy:

Travel reimbursement is intended to improve access to HBTS and support the delivery of clinical services to children and families residing in outlying areas. Effective May 1, 2005, travel time for the purposes of HBTS clinicians rendering Clinical Supervision, Lead Therapy and Treatment Consultation as well as HBTS Workers (Specialized Treatment and Treatment Support staff) to a child and family will be reimbursed for providing services to children and families in outlying areas based on "zoned mileage."

Requirements and Limitations:

The following requirements and limitations apply to travel time reimbursement without exception:

- 1. HBTS Clinicians (i.e., Clinical Supervisors, Treatment Consultants, and Lead Therapists) and HBTS Workers (Specialized Treatment and Treatment Support Staff) are eligible for travel time reimbursement if the child's home is located in an outlying area.
 - Travel reimbursement for Lead Therapist activities is allowable for provider-agencies approved to offer ABA discrete trial interventions. At any time, provider-agencies without this specific designation can apply to DHS for consideration as a provider of Applied Behavior Analysis with expertise in discrete trial intervention.
- 2. Travel reimbursement is <u>one-way</u> from the agency's location to a child's home based on "zoned mileage."
- 3. The definition of what constitutes an "outlying area" is based on whether a child's home is located 11 miles or more from the provider-agency. Two zones apply, namely, 11-20 miles (Zone 1) and 21 miles or greater (Zone 2). Travel reimbursement for clinical staff and home-based workers can be given if a child's home is in either of these two zones.
- 4. Travel Reimbursement is case based. This means that provider-agencies can bill for travel time no matter where staff started out as long as the child's home that falls within a zoned area. Claims can be submitted for all cases daily for all eligible staff.

Limitations of Travel Reimbursement:

- 1. DHS will not reimburse for travel to attend meetings or conferences relating to a child's HBTS Plan.
- 2. Staff engaged in providing Treatment Coordination is not eligible for travel time reimbursement. As defined in HBTS Certification Standards, Treatment Coordination is not a direct clinical function.

Requirements for Travel Reimbursement:

- 1. The HBTS program director is directly responsible for establishing a written procedure to demonstrate that a child's home falls within Zone 1 or Zone 2 for the purposes of receiving Travel Reimbursement. DHS reserves the right to review that the procedures are reliable.
- 2. A written record of mileage from the agency to a child's home must be kept.
- Provider-agencies will expand the geographical locations that they serve as a result of seeking Travel Reimbursement. DHS reserves the right to review this information and obtain specific reports on a quarterly basis to monitor utilization and compliance with policy requirements and expectations.

Procedures for Travel Reimbursement:

1. Staff eligible for Travel Reimbursement will use Procedure Code X0188 and the following modifiers for determining reimbursement:

Summary of Zoned Mileage and Allowable Units

Location of Child's Home from Provider-Agency	Allowable Units (1 Unit = 30 minutes)			
11 to 20 miles	1 unit per service			
21 miles or more	2 units per service			

Procedure Code	Clinical Staff	Modifier	Rate	Zone 1	Zone 2
				1 Unit	2 Units
T2003	Specialized Treatment Worker	None	\$13.23	\$13.23	\$26.23
T2003	Treatment Support Worker	None	\$13.23	\$13.23	\$26.23
T2003	Clinical Supervisor – Bachelor Level	JE	\$16.50	\$16.50	\$33.00
T2003	Clinical Supervisor – Masters Level	НО	\$30.00	\$30.00	\$60.00
T2003	Clinical Supervisor – Doctoral Level	JF	\$35.00	\$35.00	\$70.00
T2003	Treatment Consultant – Masters Level	НО	\$30.00	\$30.00	\$60.00
T2003	Treatment Consultant – Doctoral Level	JF	\$35.00	\$35.00	\$70.00
T2003	Lead Therapist – Bachelor Level	JE	\$16.50	\$16.50	\$33.00

Please refer to DHS directives regarding the preparation of claims and submission process for Travel Time reimbursement.

APPENDIX 23: FREQUENTLY ASKED QUESTIONS

Licensure Related Questions:

1. Is it necessary to have a licensed health care professional oversee an HBTS Treatment Plan if the clinical staff has been granted an Exception from Professional Licensure?

Yes. DHS and Medicaid law requires that a licensed health care professional with expertise and competence supervise non-licensed individuals providing Clinical Supervision and Treatment Consultation. Refer to Appendix 4 for additional information.

2. Is there a lifelong exception or "grandfathering" to professional licensure?

No. Regardless of previous experience or background, there is no lifelong exception for someone who is not licensed and for whom Exception to Professional Licensure applies. Individuals must comply with DHS rules and requirements regarding Variance from Professional Licensure.

3. Does DHS recognize any other State issued certification?

Yes. The provider-agency and the individual may want to consult with MHRH regarding its credentialing to qualify as a Counselor, Mental Health Counselor or Principal Counselor. DHS does not process applications for certification by MHRH. It is best to contact MHRH directly (462-2238). If an individual elects to obtain credentialing from MHRH, DHS must be provided with a copy of the individual's certification as soon as it is issued.

4. Can a non-licensed Bachelor's level clinician provide Treatment Consultation and Clinical Supervision?

No. First, a non-licensed individual with a Bachelor's degree at the time of certification in February of 2003 must have fully met the following criteria: a Bachelor's degree in special education, child development, psychologist, counseling, social work or nursing. If awarded Exception from Professional Licensure, the individual clinician could then provide and bill for Clinical Supervision. Exception status did not permit this individual to also provide Treatment Consultation. The latter service continues to be the responsibility of licensed health care professionals or those covered by Exception status at a Master's level.

5. Can an individual who is completing a Masters degree in social work but has not yet obtained licensure, as an LCSW, be considered as license eligible under Temporary Exception from Licensure?

No. The individual must have obtained licensure by Rhode Island Department of Health as an LCSW in order to be considered under the category of *Temporary Exception from Licensure*.

Billing and Reimbursement Questions:

1. How can an HBTS provider bill for CEDARR approved treatment plans when the Prior Approval has not been received at EDS?

First, the provider should confirm that EDS has not received the PA. Second, the provider should contact the involved CFC and speak to someone responsible for "batching." Please keep DHS informed of ongoing problems.

Treatment Plan Questions:

1. Is it possible to stagger the submission of Treatment Plans more evenly so that a provider-agency does not have an inordinate number of Treatment Plans due at one time?

Yes. A provider-agency can at its own discretion elect to shorten a period of authorization for Treatment Plan(s) to allow for better management.

2. Can a child receive HBTS services from more than one provider-agency at the same time?

Not necessarily. Medicaid law limits overlap such that one cannot bill for the same services on the same day. The following situations are permissible. DHS allows for a child receiving Specialized Treatment, Pre or Post-Treatment Consultation from one provider-agency to also receive Group Intervention from another at the same time.

3. Can Specialized Treatment take place if the child has been hospitalized?

No. While the home-based worker may elect to visit a child who has been hospitalized, DHS cannot authorize payment for time spent with the child. The Clinical Supervisor or Treatment Consultant assigned to the case can use administrative time to attend hospital planning or discharge meetings. The latter falls under duties assigned to the Clinical Supervisor or Treatment Consultant and is a billable activity.

4. Does the HBTS Treatment Plan need to include Specialty Consultations if a child has PT, OT or SPL provided through the IEP?

No. Specialty Consultation must meet medical necessity and have relevance to Specialized Treatment in order to be included in a HBTS Treatment Plan.

5. What role does the Specialty Consultant have?

The Specialty Consultant (i.e., SLP, OT or PT) is to observe the child, formulate treatment goals, and inform the Clinical Supervisor and home-based worker regarding methods of intervention and measurement of functioning. The Specialty Consultant has responsibility for identifying and approving treatment within the consultant's area of expertise. Documentation requirements must be met and the Specialty Consultation Form must be submitted with a child's Treatment Plan.

6. Can an HBTS Treatment Worker go with a child to day camp?

No. HBTS is intended to address treatment. A child's participation in a day camp program does not directly relate to the objectives in a Treatment Plan.

DHS does permit the use of Treatment Support as a way to facilitate a child's participation in day camp. For this to occur, the provider-agency must submit a modification to a Treatment Plan requesting Treatment Support hours.

7. Is it permissible for home-based staff to take a child to the worker's home or engage in personal business during an HBTS shift?

No. HBTS is a service that is provided in the child's home and, when clinically appropriate and approved, may involve community-based activities. It is unacceptable and against HBTS policy for home-based staff to engage in personal activities while delivering HBTS. Taking a child to a home-based worker's home is also against HBTS policy. Clinical Supervisors and Program Directors have the ongoing responsibility to educate home-based workers about professional responsibilities and boundaries.

Training Questions:

1. Can DHS pay for HBTS staff to attend conferences?

No. DHS does not provide compensation for training or professional development. At various times, however, DHS may be able to sponsor trainings and will notify provideragencies.

Prior Approval Questions:

1. Does removing services from the PA cause reimbursement problems?

No. EDS will automatically pay the number of DHS recognized units for these services. If a provider-agency requires more units, it must request this from a CEDARR or DHS.

2. Why are some HBTS services removed from PA?

The intent is to improve the process associated with provider reimbursement. DHS has the right to change HBTS services requiring a PA as part of its overall management responsibilities. Deciding on which services to remove from PA can take time in order to insure that DHS has alternative means to track utilization of authorized services.

3. What is the purpose of aligning CEDARR Family Care Plans with HBTS Treatment Plans?

CEDARR Family Centers have responsibility for identifying resources and/or professional services for children with special health care needs. For some children, it is likely that there will be more than one service indicated. The CEDARR staff review and revise the Family Care Plan at the end of the plan dates. During this time all CEDARR Direct Services are also reviewed in order to coordinate care for a child and family.

4. How will aligning of dates effect HBTS provider-agencies?

During the initial period of aligning treatment dates, there may be cases where EDS will issue a PA that is for less than 6-months. Once a child's treatment dates have been aligned, there should be no further changes in dates.

5. What happens with alignment of dates?

If the authorization by EDS is for less than 6-months, you do not need to resubmit a Treatment Plan at the end of the authorization period. A staff member from the CEDARR Family Center will call the contact person for the HBTS Treatment Plan to review treatment progress and to obtain additional information in order to recommend a new period of authorized care. If care is to continue, the CEDARR Family Center will issue another 6-month PA.

If authorization from EDS is not received by the end of the first month of the treatment period, the HBTS provider-agency should first contact EDS to check on the status of the PA. If needed, a representative from the HBTS provider-agency may call the CEDARR Family Center to address PA questions.

Travel Reimbursement Questions:

1. Why can't staff get paid for their travel while providing HBTS?

This is no longer true. The issue of travel time for clinical and home-based staff has received considerable thought and discussion within DHS. Travel time for staff is one factor that provider-agencies consider when developing an HBTS service area. Families in outlying areas have been disadvantaged as a result of provider-agencies only operating within specific service areas. In order to address this issue, DHS formulated a policy that was consistent with Federal IRS regulations. IRS regulations prohibit reimbursement for an individual's personal travel to work.

On April 29 of 2005, DHS issued its Travel Reimbursement Policy to HBTS provider-agencies. Effective May 1, travel reimbursement can be billed using a zoned mileage formula that is case based.

2. Why can't Treatment Coordinators receive Travel Reimbursement?

Travel Reimbursement is intended to support the delivery of clinical services. Treatment Coordination, as defined in HBTS Certification Standards issued in 2003, was not considered to be a clinical service. DHS may review this decision in the future but for now, travel reimbursement for Treatment Coordinators is not approved.

3. If a set of siblings is receiving HBTS is it permissible to bill for Travel Reimbursement for each child?

It is appropriate to bill for Travel Reimbursement for the HBTS Treatment Worker or Treatment Support Worker involved with each child.

It is not appropriate to bill for Travel Reimbursement when a Clinical Supervisor or Treatment Consultant observes both children at different times during one home visit. If there are separate visits then it is permissible to bill Travel Reimbursement per child for the clinicians providing service.

ATTACHMENT A: APPLICATION GUIDE

Available Upon Request